# Audit partners' perspective: Mandatory audit partner rotation in New Zealand

# A thesis

submitted to the University of Canterbury

in partial fulfilment of the requirements for the degree of

Master of Commerce

in Accounting and Information's systems

by Adam Fairburn

#### **Abstract**

New Zealand mandated audit partner rotation regulations for New Zealand Exchange (NZX) entities in May 2004, since then audit partner rotation policies have been implemented for Financial Markets Conduct entities and entities overseen by the Office of the Auditor General. Time-on periods for engagement partners range from five to seven-year followed by a five-year cooling-off period. Rotation policies also apply to engagement quality control reviewers and other key audit partners. The multi-dimensional structure of rotation regulations that have been implemented aligns New Zealand with countries that have significantly larger populations and GDP's. The purpose of audit partner rotation is to reduce the 'familiarity threat' and increase independence between the engagement partner and client. The incoming audit partner should also bring a 'fresh perspective' to the engagement to increase audit quality and efficiencies. Using semi-structured interviews to conduct a qualitative study, the purpose of which is to discover how audit partners perceive mandatory audit partner rotation and what the implications are from transposing audit partner rotation regulations from large-scale economies to a small-scale economy. Fifteen interviewees were selected consisting of Big Four and non-Big Four audit partners. The research design for this study was to replicate a study done by Braun and Clark (2006) using thematic analysis to analyse the qualitative data and to document the real-life experiences of audit partners following a change to their realities. The key findings and conclusions from this study are that regulators have chosen to align New Zealand audit partner rotation policies to large-scale economises rather than 'right size' them to New Zealand's demographic environment. Audit partners perceive that mandatory audit partner rotation disproportionately disadvantage small to mid-sized audit firms due to limited audit partner resources. The implications of which has been that a number of small audit firms have left the registered audit firms market increasing market concentration. Extended cooling-off periods are the biggest challenge to audit firms

that may no longer be necessary due to significant changes occurring within client entities that naturally reduces the likelihood of a 'familiarity threat'. An incoming audit partners 'fresh perspective' can be described as a critical reassessment of the material risks that threaten the entity and previous judgments made. Making audit partner rotation a governance issue for the entity and an audit quality increasing exercise. In response to imposed mandatory audit partner rotation, audit firms have implemented planning and coordination mechanisms to manage the implemented policies. The purpose of which is to retain institutional knowledge within the engagement, have audit team members present with knowledge of a client's specific matters, and replace the engagement partner with an audit partner that has industry experience.

# Acknowledgments

I would like to sincerely thank my research supervisors, Dr Julia Wu and Assoc. Prof. Richard Fisher for their continued support and guidance throughout this process. This research would not have been possible without their advice and tolerant approach, I will always be very grateful.

I would also like to thank all the respondents who were willing to the participate in this study and sparing me some of your valuable time. The knowledge and experiences that you imparted was invaluable as were the insights into the audit industry.

Thank you to my family for your support and encouragement, as you are all aware without your help this achievement would have never been possible.

Thank you all!

<u>Table of Contents</u>
Abstract
Acknowledgments4
Chapter 1: Introduction8
1.1 Research background8
1.2 Research Problem
1.3 Research Objective
1.4 Statement of research question
1.5 Structure of thesis
Chapter 2: Literature review
2.1 Introduction
2.2 Regulating audit partner rotation
2.3 Literature review – Audit partner rotation
2.3.1 Transposing regulations from large-scale economies to a small-scale economy 24
2.3.2 Audit partner tenure
2.3.3 Familiarity and familiarity threat
2.3.4 A fresh perspective
2.3.5 Internal controls
2.3.6 The loss of client specific knowledge
2.3.7 Cooling-off periods
2.3.8 Auditor negotiations
2.3.9 Auditor transitions
2.3.10 Large client entities
2.3.11 Managing audit partner rotation
2.3.12 Summary
2.7 Gaps in the research
Chapter 3: Methodology
3.1 Introduction
3.2 Interview design
3.3 Data collection and analysis
3.3.1 Populations and Sample
3.3.2 Data Analysis51
3.3.3 Ethical considerations53
Chapter 4: Findings55
4.1 Introduction
4.2 Implications of audit partner rotation on audit practice in New Zealand55
4.2.1 Solidifying audit partner rotation policies56

4.2.2 Audit partner rotation and market concentration	57
4.2.3 Audit partner rotation effects both Big Four and Non-Big Four audit firms	58
4.2.4 The practicality of audit partner rotation policies in New Zealand	59
4.2.5 Audit partner rotation puts pressure on the audit firm to increased their license audit partner resources	
4.2.6 There are significantly more resources involved in auditing large complex ent	
4.3 Planning and coordination mechanisms	
4.3.1 Audit partner rotation involves a complex matrix	65
4.3.2 Transitioning the engagement partners involves a number of complexities for audit firm	
4.3.3 Audit firms need to consider an engagement partners capacity and portfolio	70
4.3.4 Rotating the engagement quality control reviewers challenges an audit firms resources	71
4.3.5 Audit partners with client's specific knowledge increase audit quality	72
4.3.6 The audit team drives audit quality in an engagement	74
4.3.7 The more complex the business, the more time on an audit partner needs	75
4.4 Relationships between engagement partners and their clients	76
4.4.1 The first years of an audit partner's engagement	77
4.4.2 The engagement partners 'fresh perspective' can lead to increased audit qualit	y 79
4.4.3 Audit partner rotation enhances an engagement partners ability to challenge	80
4.4.4 Audit partner rotation reduces the familiarity threat	82
4.4.5 Clients perceive audit partner rotation as an inconvenience	84
4.4.6 Boards of directors' view audit partner rotation as good governance	85
4.4.7 Organisational employees are constantly changing	86
4.5 Chapter summary	87
Chapter 5: Discussions	88
5.1 Introduction	88
5.2 Implications of audit partner rotation on audit practice in New Zealand	88
5.2.1 Audit firms in New Zealand were already rotating their audit partners before a partner rotation became mandatory	
5.2.2 Mandatory audit partner rotation has resulted in increased market concentration.  New Zealand	
5.2.3 The practical implications of audit partner rotation regulations in the New Zea environment	
5.2.4 Audit partner rotation and non-Big Four audit firms	92
5.2.5 Large client entities and audit partner rotation	94
5.2.6 Audit partner rotation and Big Four audit firms	96
5.2.7 Conclusions	97

5.3 Planning and coordination mechanism	98
5.3.1 Audit firm size and the matrix of audit partner rotation	98
5.3.2 Transitioning engagement partners within the matrix of audit partner rotation	.100
5.3.3 Audit firms consider an audit partners entire portfolio before allocating them to engagement	
5.3.4 Rotating the EQCR in a small-scale economy creates problems for the audit fi	
5.3.5 Institutional knowledge is kept within the audit team	.104
5.3.6 Cooling-off periods are the biggest obstacle for New Zealand audit firms	.106
5.3.7 Conclusions	.107
5.4 Relationships between engagement partners and their clients	.108
5.4.1 Audit partners face a knowledge and experience imbalance in the first year of engagement	
5.4.2 The 'fresh perspective' is a re-assessment of the material threats that may three the entity	
5.4.3 Audit partner rotation leads to internal and external challenges	.114
5.4.4 Audit partner rotation ends the relationship between the engagement partner at the client	
5.4.5 Audit partner rotation disrupts the relationship between the engagement partner and management	
5.4.6 Boards of directors perceive audit partner rotation as a best practice	.120
5.4.7 Employees in client entities are constantly changing	.121
5.4.8 Conclusions	.122
Chapter 6: Conclusions, limitations and future research.	.125
6.1 Conclusions	.125
6.2 Contributions to practice	.127
6.2.1 Right sizing regulations to an economies GDP and population	.128
6.2.2 Implementing a matrix for audit partner rotation management	.129
6.3 Future research	.130
6.3.1 Mandatory audit firm rotation	.130
6.3.3 The implications of gender on audit engagement completions	.131
6.4 Limitations	.133
References	.135
Appendices	.159
Appendix 1: Initial contact email to participants – email template	
Appendix 2: Interview Protocols	.160

# **Chapter 1: Introduction**

## 1.1 Research background

This thesis is to investigate the imposition of mandatory audit partner rotation regulations in New Zealand. Rotation policies have been implemented in many countries based on the perception that longer engagement partner tenures result in a 'familiarity threat'. The purpose of bringing in a new engagement partner is so that they will bring a 'fresh perspective' to the engagement, which is presumed to increase independence and objectivity (Dodgson et al. 2020). Between 2004 and 2014 a multi-dimensional system of audit partner rotation has been implemented for different entity types including extended cooling-off periods. This complex structure of rotation regulations aligns New Zealand with much larger countries that have significantly larger populations and gross domestic products (GDP). Research on audit partners rotation has predominantly been done countries with large commercial entities and large audit firms. Little is known about how mandatory audit partner rotation could affect a small-scale economy such as New Zealand. This research will provide evidence on the two research questions: First, how do audit partners perceive the imposition of mandatory audit partner rotation regulations? Second, what are the implications of mandatory audit partner rotation in New Zealand?

New Zealand has adopted trans-national audit partner rotation regulations. Complex multi-level regulations can create inherent tensions between different organisational actors such as Big Four and non-Big Four audit firms (Quack and Schuler, 2015). New Zealand commercial environment is made up of mainly small and mid-sized entities with a few large commercial entities (MYOB). Auditing larger complex entities requires increased audit partner resources to cover all the available roles (Stewart and Kinney, 2013). Extended cooling-off periods have described by practitioners as onerous (EY, 2018) and should only be implemented if they can be proven to increase audit quality (Mattock et al. 2022). The

question this raises, is whether trans-national audit partner rotation regulations are practical in New Zealand and are there different implications for Big Four and non-Big Four audit firms from implementing audit partner rotation regulations and extended cooling-off periods.

Audit partner rotation regulations and cooling-off periods have been transposed from much larger economies. Of importance to this study is how these implemented regulations fit into the New Zealand demographic environment. Some researchers argue against audit partner rotation because of the loss of client specific knowledge which could affect audit quality in the first few years of the engagement (Myers et al. 2003). Research has found that auditors with specialist industry experience outperform non-specialists in detecting errors (Knechel et al. 2013). To maintain audit quality, audit firms need to replace the incumbent audit partner with an incoming audit partner with specialist industry experience. Which makes audit partner rotation an audit partner resource issue for audit firms.

Dodgson et al. (2020) researched audit partner transitions discovering that audit firms use 'relationship manager' and that senior audit partners are involved in the transition.

Pittman et al. (2022) suggests that audit firms train an incoming partner into the engagements partner's position. Sharma et al. (2017) found that audit firms incur increased costs around rotation. What can be interpreted from this research evidence, is that audit firms apply resources to the management of audit partner rotation. Research does not provide answers to how audit firms manage audit partner rotation. This research will investigate how audit firms manage the imposition of audit partner rotation in New Zealand.

The multi-dimensional structure of audit partner rotation regulations indicates that audit firms will have to manage their human resource capital. Human resource management provides literature on how organisation can implement matrixes and rosters for the efficient allocation of human resources (Rogstadius et al. 2011). Audit firms focus on replacing engagement partners with industry specific knowledge (Arthur et al. 2017). An organisation

internal policies should include managing the collective knowledge of their employees to fulfil the needs of their clients (Baird and Meshoulam, 1988). This research will suggest a practical framework for audit firms in small-scale economies to manage audit partner rotation.

Auditor-client relationships are complex, engagement partners need to facilitate the transference of information from the client to complete a quality audit. Yet, challenge the management of the entity on the key assumptions and judgements they have made in auditor negotiations (Beattie et al. 2000). The extent of this challenge is dependent on whether the engagement partner determines the judgement to be a material risk to the entity. Auditor negotiations can result in conflicts between the engagement partner and management of the entity around material risks to the entity. The primary relationship between an auditor and client is critical to securing a quality outcome (Beattie et al. 2004). Audit partner rotation periodically ends the auditor client relationship. Little is known about how audit partner rotation impacts auditor negotiations in the first few years of an engagement partners tenure, which motivates the researcher to investigate this issue.

Lennox et al. (2014) researched audit partner rotation, finding that the incoming audit partner conducts a peer reviews of the incumbents' assumptions and judgments. In 2015 the International Auditing and Assurance Standards Board (IAASB) required that the name of the engagement partner to be included in the auditor's report for listed entities. This regulation means investors can track an engagement partner history. The Audit regulations Act 2011 allows the Financial Markets Authority (FMA) in conjunction with CAANZ to commence civil or criminal prosecutions against audit partners in the event of an audit failure or corporate collapse. The implications of this for the individual audit partner is that they share litigation responsibilities with the audit firm. These two regulations increase the internal

challenges between audit partners. This research will look to discover how these regulations impacts auditors' internal relationships.

#### 1.2 Research Problem

Audit partner rotation became a prominent issue in research after the US House of Representatives passed the Sarbanes Oxley Act 2002 (SOX). The motivation for passing this act was a number of significant frauds that took place between the late 1990's and early 2000's (e.g., Enron, WorldCom, Tyco, Waste Management). The purpose of this act was to enhance the public confidence in governance and auditing of US public companies (Coates, 2007: Romano, 2004). Section 203 of the SOX required the engagement partner and concurring partner to rotate every five years followed by a cooling–off period of five years for Securities Exchange Commission (SEC) companies (US House of Representative, 2002).

Within two years of the SOX being passed the UK, Australia, Canada and New Zealand all followed suit introducing audit partner rotation requirements. Before the introduction of audit partner rotation New Zealand audit partners were able to carry out engagements on a continuous basis. What was stipulated by the New Zealand Institute of Chartered Accountants (NZICA) was that audit companies were required to consider a long association of personnel (including audit partners) with an audit client.

In adopting audit partner rotation regulations, New Zealand is essentially keeping in line with major capital markets overseas and ensuring that the quality of New Zealand's financial reporting is compatible to leading practices in the world. There are undoubtedly benefits to adopting international auditing standards, the problematic question is whether the benefits of implementing international regulations outweigh the costs. Audit firms in New Zealand can be described as comparatively small but highly developed (Johnson et al. 1995) which patently effects the amount of audit partners available for rotation with the relevant experience and skills to conduct specialist audits.

## 1.3 Research Objective

The objective of this research is to contribute to the body of knowledge in this area by interpreting the lived experiences of licensed audit partners in New Zealand. Its purpose is to discover how audit partners perceive these regulatory changes and what the implications are to a small-scale economy such as New Zealand. This is an exploratory study that is looking to gain the perceptions of audit partners on how regulations are affecting the industry. By providing direct evidence from audit partners, this study aims to provide useful information to regulators on the effectiveness of the regulation they have imposed and the audit regime they have constructed.

## 1.4 Statement of research question

New Zealand has implemented audit partner rotation regulations in line with world leading security markets. To understand how mandatory audit partner rotation in New Zealand affect audit partners work and life, we ask our first research question.

**RQ1:** How do audit partners perceive the imposition of mandatory audit partner rotation regulations?

The countries New Zealand's has aligned itself with have significant larger commercial entities and audit firms. To understand the affects this has on New Zealand audit partners and firms, we ask our second research question.

**RQ1:** What are the implications of mandatory audit partner rotation in New Zealand?

To investigate these issues, the researchers conducted Thirteen semi-structured interviews with Big Four and non-Big Four audit partners. By using a qualitative methods, the researcher was able to have in-depth conversations with participants to gain an understanding of how audit partners perceived the implemented regulations (Ogharanduk et al. 2016). This is an inductive explorative study which seeks to discover binding principles to

construct generalisations. Thematic analysis can be a constructionist method that examines the way in which events, realities, meanings, and experiences can reflect changing social realities. By using thematic analysis to analyse the data, the researcher can interpret the meanings of experiences to discover the implications that audit partner rotation has had on audit partners and firms (Braun and Clark, 2006).

The key findings from this study were that audit partner rotation disproportionately disadvantaged small to mid-sized audit firms due to limited partner resources forcing them to merge or leave the domestically registered audit market. Some consumers of audit services have also been disadvantaged due to the increased market concentration that audit partner rotation has caused. Extended cooling-off periods were perceived as posing the biggest challenge to New Zealand audit firms. Audit partners from Big Four audit firms provided qualitative research evidence that there are only a few audit partners in New Zealand capable of auditing a large complex entities, which implies reduced audit quality from audit partner rotation regulations. Audit firms have constructed matrixes to manage the multi-dimensional structure of audit partners rotation, to retain institutional knowledge, and provide an audit partner with industry experience. Rotation regulations have a negative effect on auditor-client relationships and negotiations in the first year of the engagement. The 'fresh perspective' has a positive affect with the incoming audit partner providing a fresh challenge to management in the second year of the engagement.

Audit partners perceive that audit partner rotation regulations should be 'right sized' to the New Zealand demographic environment and that regulators were implementing regulations into New Zealand that are designed for large international economise. This represents a disconnect between how audit partners perceive regulators should be implementing regulations to how regulations are actually being implemented. The biggest problem for New Zealand audit firms were cooling-off periods, which may no longer be

necessary due to employees in client entities constantly changing. Audit partners perceive that audit partner rotation makes them more reliant on the audit managers and team relationships as well as their knowledge of client specific matters. The practical reality is that audit partner rotation becomes a resource issue in a small-scale economy such as New Zealand, disproportionately affecting small and mid-sized audit firms. The implications of audit partner rotation are that audit firms need to increase their audit partner resources to provide services to their clients. Audit firms construct matrixes to provide audit partners with industry experience and maintain audit quality. For Big Four audit firms, rotation regulations mean not being able to provide the most suitable audit partner, while some non-Big Four audit firms struggle to provide audit partners services for certain engagements.

This research contributes to literature by providing qualitative research evidence on audit partner rotation regulations being transposed for large-scale economise to small-scale economies. A key finding was that regulators should consider a countries demographic environment before implementing regulations and 'right size' them to the resources available to the audit firm. This research contributes to knowledge by providing planning and coordination mechanisms for audit firms to manage audit partner rotation. Another contribution to knowledge was that audit partner rotation increases the challenge to management in the second year of the engagement by providing a 'fresh perspective' that can be considered a critical reassessment of the material risks that threaten the entity and previous judgments made, making audit partner rotation a governance issue for the entity.

#### 1.5 Structure of thesis

Following the introduction chapter, Chapter 2 contains the literature review which: Firstly, gives an overview of the regulations imposed. Secondly, matches prior research to the identified themes discovered in this research. Chapter 3 covers the methodologies and methods undertaken in this thesis and will including methods, paradigms, research design,

interview design, interview question, data collection and analysis. Chapter 4 presents the findings from this research, which is split into three central topics: Firstly, Implications of audit partner rotation on audit practice in New Zealand. Secondly, Planning and coordination mechanisms. Thirdly, relationships between engagement partners and their clients. Chapter 5, then discusses and interprets the findings from this study on the three central topics. Chapter 6 concludes this study by providing conclusions that answer the research question, contributions to practice, recommendations for future research, and limitations that have affected this research.

## **Chapter 2: Literature review**

#### 2.1 Introduction

Interviews were conducted with fifteen audit partners across New Zealand. In inductively analysing the qualitative data, a number of consistent themes were identified of being important to audit partners. The purpose of this literature review is to provide an overview of the existing literature and provide background information on the major themes expressed by audit partners. The first section of this chapter will present on overview on the regulations that have been imposed on audit partners. The second section will provide a summary of the relevant literature to audit partner rotation and the identified themes.

Through this literature review, topics will be assessed for their potential implications to audit quality. The methods and methodologies of specific pieces of literature that closely aligns to the research topics will be critically disseminated for their relevance in answering the research question. By doing this, this literature review will provide an information background to be expanded on in the findings and discussions chapter of this research.

## 2.2 Regulating audit partner rotation

Audit partner rotation has been implemented in many countries around the world. Following the introduction of the SOX in the US, New Zealand, Canada, UK, and Australia all followed suite implementing the same rules, interestingly in 2010 Canada changed its five—year rotation regulation back to seven—years for the engagement partner of listed countries (Daugherty et al. 2012). The European Union in response to the financial crisis brought in audit firm rotation regulations as well as audit partner rotation regulations, though a number of its member states already had audit partner rotation regulations in place. Under Directive 2014/56/EU and Regulation (EU) No. 537/2014 audit partner rotation would be mandated at seven-years followed by a cooling—off period of three—years. Which incidentally were the rules that the UK, US, and Canada had in place before the SOX was passed. The

European Union has a diverse range of countries in it with population rates ranging from 441,543 in Malta to 83,783,942 in Germany, the average populations per country is 11,817,542 with the medium population being 7,977,422. New Zealand more closely resembles a number of countries in European Union by population than it resembles the UK, Canada, Australia and US. A comparison of regulations is presented in the table below to show how regulations have been aligned between the countries.

Table 1: Cross Country audit partner rotation regulations

Country/ Union	Legislation	Reason	Regulation  The engagement partner and concurring partner to rotate every five years followed a cooling–off period of five years for listed companies	
US	Sarbanes-Oxley Act 2002	Was passed mainly in response to the corporate scandals of the Dot-com- era		
UK	Auditing Practices Board Ethical Standard 3 for rotation.	After the fall of Enron the UK set up the (GCAA) co-ordinated group of auditing and accounting	The engagement partner and EQCR to rotate every five years followed a cooling—off period of five years.	
Canada	Independence – Long Association (Rule 204.4 (20))	In recognition that a long association may create a familiarity threat	The engagement partner and EQCR to rotate every five years followed a cooling-off period of five years for reporting issuers and listed entities.	
European Union	Directive 2014/56/EU and Regulation (EU) No. 537/2014	In response to the 2007- 2008 Financial crisis the European union commission The Green Paper	The audit partner to be rotated every seven— years followed by a three—year cooling off period.	
Australia	Section 290 and 291 of APES 110 Code of Ethics for Professional Accountants.	That there is a diminution in the quality of audits after associated with long audit partner tenure.	The engagement partner and EQCR to rotate every five years followed a cooling—off period of five years.	
New Zealand	PES 1 for FMC entities. Listing rule 2.13.3 (f) for NZX companies and AG PES 1 Code of Ethics	New Zealand passed audit partner regulations to keep in line with its trading partners	The engagement partner and EQCR to rotate every five years followed a cooling-off period of five years for NZX companies.	

Audit partner rotation regulation were initially imposed on NZX entities in May 2004 under listing rule 2.13.3 (f). Since then, engagement partners, engagement quality control reviewer (EQCR) and other key audit partners have had numerous regulations placed upon them affecting their social reality and work-life balance. This section of the chapter will: Firstly, define the roles of auditors that are subject to audit partner requirement. Secondly, present the specific audit partner rotation time frames that audit partners must adhere to, and thirdly, provide contextual information useful to the readers of this thesis.

The *engagement partner* is the person that has been appointed by the audit firm to the client to perform the task of completing an audit. An *engagement partner* is the person responsible for the effective function of all phases in the audit and for exercising their professional judgment (Lawinsider), an interchangeable term for the *engagement partner* is *audit partner*. In large client audits the engagement partner can be in charge of several audit partners, audit mangers and audit team members (Stewart and Kinney, 2013). The *engagement partner* is responsible for providing an opinion on the consolidated financial statements that may include a number of subsidiaries. Along with the engagement quality control reviewer (EQCR) and other senior audit staff, the *engagement partner* is responsible for identifying the material threats to the entity and challenging the assumptions and judgements made by the entity. They are the person that has the appropriate authority from a professional, legal or regulatory body (IESBA, 2007: XRB, 2011).

The XRB (2011), describe engagement quality control review as a process designed to provide an objective evaluation, of the significant judgements made by the engagement team and the conclusions it has reached in formulating the auditor's report. The EQCR is the person that conducts the engagement quality control review. EQCR's review selected working papers relating to significant judgements made by the engagement team and evaluate conclusions made for their appropriateness assessing the audit high–risk transactions and

account balances. The review procedure should involve a discussions between the engagement partner and EQCR on any significant matters, that no matters arising breach generally accepted accounting principles and ensure there are no matters outstanding. The engagement partner is required to gain EQCR approval before the issuance of an audit opinion. An EQCR should not be considered part of the audit team but as a quality control mechanism to provide reasonable assurance.

The International Ethics Standards Board for Accountant (IESBA) define an 'other key audit partner' as a person that conducts an audit of a subsidiary or division within an entity. As opposed to the engagement partner who is responsible for the 'group' and consolidated financial statements, the 'other key audit partner' is responsible for the subsidiaries or divisions financial statements that will become part of consolidated financial accounts. This person may work independently of the engagement partner, essentially conducting an audit of stand-alone entity with a separate audit manager and audit team working under them. They do however report directly to the engagement partner. This person takes responsibility for the key decisions on significant judgements matters and is expected to have an in-depth knowledge of the industry they are auditing because a lack of knowledge presents a threat to audit quality (IESBA, 2007).

The key provisions of the Auditor Regulation ACT 2011 require the auditor of an FMC reporting entity to be licensed and the audit firms to be registered. The term 'auditor' refers to the engagement partner and EQCR who are both required to be licensed to complete an FMC audit (Section 8, 1). To be a registered audit firm, the firm must have at least one licensed auditor, it is the responsibility of the audit firm is to ensure that the engagement partner and EQCR performing an FMC audit are licensed (Section 9, 2). To become a licensed auditor, the auditor must first have their personal qualifications recognised by the company's office which includes having five years of audit experience and 3000 hours of

audit work immediately prior to applying for a license, 750 of these hours must be spent auditing an NZX or FMC entity (FMA, 2020).

The engagement partner and EQCR must also adhere to External Reporting Board (XRB) requirements, PES 1. This means that an engagement partner and EQCR must be licensed to carry out NZX audits, and similarly for OAG entities, under AG PES 1 Code of Ethics. Section 78 of the Auditor Regulation ACT 2011 gives the FMA powers to suspend or cancel an auditor's license. Under part 1 of the preliminary provisions, the FMA has the power to gather certain general information, issue fines and demand reimbursement for costs incurred in investigations. Criminal or civil proceeding are carried out in conjunction with the auditor's professional membership bodies. The passing of The Financial Markets Conduct Act 2013 introduced a system of de–escalating levels of liabilities and a new set of regulatory powers and infringement offences which increased the emphasis on civil liabilities for contraventions, reserving criminal proceeding for the most serious violations (O'Niel, 2015).

In 2015 the International Auditing and Assurance Standards Board (IAASB) through their Enhanced Reporting Requirements, requires engagement partners to sign the audit opinion a provision. This requirement was already in place in the US and has now been implemented in the UK, Canada, New Zealand and Australia for issuer audits. The defined purpose for this initiative is it allows investors to track an audit partners history relating to industry experience, restatements, involvement in disciplinary proceedings and litigation. This gives investors and stakeholders a better understandings of potential risks associated with the audit (Goelzer, 2019). The accounting profession responded by pointing out that this could deter audit partners from accepting riskier clients (Christensen et al. 2021). The regulations that have been imposed on audit partners in New Zealand are presented in the table below.

Table 2: Audit partner legislation in New Zealand

May 2004	NZX makes audit partner rotation mandatory requiring the key audit partners to rotate every five years. The definition of the key audit partner in the rules has the meaning given in the External Reporting Board Professional Code of Ethics PES (1) revised.
May 2011	Financial Markets Authority Act 2011, part 1 preliminary provisions (a) establish the Financial Markets Authority as an independent crown entity (c) provide the Financial Markets Authority to have certain general information—gathering and enforcement powers.
May 2011	Auditors Regulation Act 2011, (5) Functions under this act (e) to conduct quality reviews and investigations under subpart 6 and 7 of part 2: Section 8 – Auditors in respect of FMC audits must be licensed (1) Every natural person who acts as the auditor in respect of a FMC audit must hold a license. (e) to conduct quality reviews and investigations under subpart 6 and 7 of part 2:
January 2014	Professional and Ethical Standard 1 (Revised). Definition, key audit partner: The engagement partner, the individual responsible for the engagement quality control review, and other audit partners. 290.151 in respect to the audit and review of a public interest entity, an individual shall not be a key audit partner for more than seven years. After such time, the individual shall not be a member of the engagement team or be a key audit partner for the client for two years.
October 2015	Auditor Reporting Enhancements. Additional independence statement: The IAASB requires, for auditor reports of listed entities, that the auditors' responsibilities paragraph include a statement that the auditor is required to disclose to those charged with governance all relationships and other matters that may be reasonably be thought to bear on the auditor's independence. Engagement partners name (19): The IAASB requires that the name of the engagement partners be included in the auditor's report for listed entities.
January 2018	Professional and Ethical Standard 1 (Revised). 290.155 If the individual acted as the engagement partner for seven cumulative years, the cooling—off period shall be five consecutive years. 290.156 Where the individual has been responsible for the engagement quality control review and has acted in that capacity for seven cumulative years, the cooling off period shall be a period of three consecutive years.

The initial audit partner rotation regulations were for audit partners that audited NZX listed companies. NZX Listing rule 2.13.3 (f) states that the *key audit partner* is changed at least every five—years. Under the Auditor Regulation Act 2011 this can be interpreted to

mean the engagement partner and EQCR. Cooling-off periods to be applied are found in the XRB PES 1, which requires an engagement partner to have a five—year cooling-off period and the EQCR to adhere to a three—year cooling-off period. Other key audit partners for NZX companies are required to rotate every five—years followed by a two—year cooling-off period.

Audit partner rotation requirements for FMC reporting entities are set out in PES 1. The criteria for a Tier 1 entity that the PES 1 applies to, can be described as both listed debt and equity issuers and include banks, insurers, credit unions and buildings societies etc that are not listed on New Zealand's stock exchange. For these entities the engagement partner is required to rotate every seven—years followed by a cooling—off period of five—years. The EQCR is required to rotate every seven—years followed by three—year cooling-off period. Other key audit partners are required to rotate every seven—years followed by a two—year cooling-off period. In 2018 the cooling-off period for EQCR's was extended from two—years to three.

The office of the Auditor General (OAG) is responsible for the audits of public entities. Examples of public entities include Ports, schools, government departments and tertiary education institutions etc which totals over four thousand entities. Audit New Zealand is the OAG preferred supplier, in addition to Audit New Zealand the OAG outsources audit work to sixty—one external audit service providers to carry out audit services and engagements. For these outsourced audits the OAG conducts performance reviews on the audited services that have been provided. For OAG entities the engagement partner is required to rotate six—years followed by a cooling—off period of five—years, EQCR's are required to rotate every six—years followed by three-year cooling—off period. Other key audit partners are required to rotate every six—years followed by cooling-off period of two—years (OAG, 2004). Table 3 presents the myriad of different audit partner rotation regulations that have been implemented in New Zealand.

Table 3: Rotation Requirements for New Zealand Entities

<b>Entity Type</b>	Engagement partner		EQCR		Other Key audit partner roles	
	Time-on	Cooling- Off	Time-on	Colling- Off	Time-on	Cooling- off
NZX Listed Entities	5 Years	5 Years	5 Years	3 Years	5 Years	2 Years
FMC Reporting Entities	7 Years	5 Years	7 Years	3 Years	7 Years	2 Years
Office of the Auditor General	6 Years	5 Years	6 Years	3 Years	6 Years	2 Years

Note 1: Time on, refers to the maximum cumulative years an audit partner may serve in an engagement.

Note 2: Cooling—Off, refers to a minimum consecutive period where a key audit partner shall not be involved with the entity in any professional capacity.

Audit firms are managed by a group of senior auditors that are usually led by a managing partner. That person is responsible for allocating audit partners to clients and manging their portfolios (Lennox et al. 2023). (Dodgson et al. (2020) provides an extrospective view of other members of the audit firm that are typically involved in the client engagement. Sector leaders are often involved in managing rotations internally for a complex client as is a national risk management group. Dodgson et al. (2020) also states that senior partners may have contact with the client during the rotation process and 'relationship partners' are often assigned to clients on a non–decision making basis. The extend of their involved isn't known which presents an opportunity for future research, though extracting this information from audit firms may prove difficult.

## 2.3 Literature review – Audit partner rotation

The discussions with audit partners resulted in a number of themes being identified.

Following an inductive study approach, relevant literature has been determined after the data

collection. The purpose of this literature review is to overview quantitative and qualitative research evidence on audit partner rotation regulations. Research evidence and regulatory perspectives will be matched to the identified themes in this research.

This section of the chapter will be set out as follows: Firstly, transposing regulations from large-scale economies to a small-scale economy. Secondly, audit firm tenure. Thirdly, familiarity and familiarity threat. Fourth, a fresh perspective. Fifth, internal controls. Sixth, the loss of client specific knowledge. Seventh, cooling-off periods. Eighth, auditor negotiation. Ninth, auditor transitions. Tenth, managing audit partner rotation.

## 2.3.1 Transposing regulations from large-scale economies to a small-scale economy

Of importance to this research is that New Zealand regulators have transposed audit partner rotation regulations from large-scale economies into a small-scale economy. In 2011 the New Zealand parliament passed legislation to establish the Financial Markets Authority (FMA) creating a regulatory regime to oversee the New Zealand audit market. In 2012 the FMA joined the International Forum of Independent Audit Regulators (IFIAR) who's purpose is to support regulatory regimes in coordinating international audit regulations and to give investors and other stakeholders confidence in a country's capital markets (FMA, 2012: IFIAR, 2020). The establishment of the FMA and joining the IFIAR aligned New Zealand with a number of its significant trading partners. The IFIAR strategic goal is to improve global audit quality to reassure capital market investors and currently has fifty-four members. The IFIAR encourages its members to adopt international audit regulations and provide specialist knowledge to support their implementation but does not appear to provide a framework for applying audit regulations into different sized economies. What this suggests is that international audit regulations.

Rodrigo (2005) found that small countries when implementing regulations rely more on benchmarking, which implies that smaller countries apply regulations proportionate to the

size of their economy while larger countries rely more on empirical methods. Quack and Schuler, (2015) discuss how the implementation of regulations may have different implications for small and large firms. It should also be considered that larger organisations may seek regulations to their own advantage and lobby against regulations that could disadvantage them in the marketplace. Regulators supply information on how to implement regulations but little analysis has been done on effects of implementing regulations from large-scale economies to small-scale economies. Prompting more research into this subject area.

## 2.3.2 Audit partner tenure

Audit partner rotation restricts an engagement partner from carrying out an audit client engagements on a continuous basis. Carey and Simnet (2006) explain that the regulatory policy of audit partner rotation is based on the assumption that there will be a diminution in audit quality over the tenure of an engagement. The argument for limiting an audit partners tenure to a specific period of time is based on a perception that audit partners in longer tenures are less likely to challenge opportunistic management decisions, that a 'familiarity threat' will develop, which implies that audit partners may start to agree with management decisions, and that audit partners may sacrifice objectivity to retain a prestigious client (Carey & Simnet, 2006: Chi & Huang, 2005: Chi et al. 2009: Fitzgerald et al. 2018). What researcher and regulators suggest is that long tenures erode an audit partners capacity for critical appraisal. If that erosion leads to a reduction in audit quality or an audit failure then there will be an imposed cost to society and the profession (Carey & Simnett, 2006)

The above research and regulatory position could be considered to be at odds with capital market actor such as banks and insurance companies who associate long tenures with accuracy, credibility, and quality (Ball et al. 2008: Bharath et al. 2008: Kim et al. 2011). The argument is that auditors gain more experience from long tenure and are better able to

understand the financial choices made by management are proper (Chen et al. 2008). Myers et al. (2003) found that long tenure is associated to higher earnings quality and on average results in greater constraints on extreme management decisions. Why capital markets actors associate long tenures with accuracy and credibility is that the extent of earnings management is generally lower when auditor tenure is longer (Johnstone et al. 2002: Knechel & Vanstraelen, 2007). These researchers suggest that as an auditor gains more experience earnings quality increases because they are able to determine and constrain poor choices made by management.

Audit partner rotation has been made mandatory in many countries based on the assumption that longer tenures reduce independence and objectivity (Carey and Simnett, 2006: Chen et al. 2008). Proponents for mandatory audit partner rotation argue that "limiting auditor tenure reduces concerns about deteriorating independence and audit quality" (Myers et al. 2003, pg. 781). Consistent with the regulatory view, Mautz and Sharaf (1961) suggest that extended auditor client relationships may have a deteriorating effect on independence which can be attributed to a reduction in the auditor's objectivity of a client in longer engagements. The intent of mandatory audit partner rotation is that the incoming audit partner will provide a 'fresh perspective' of the engagement which is presumed to increase independence and objectivity (Dodgson et al. 2020). Prior research suggests that mandatory audit partner rotation improves independence which positively effects audit quality (Daugherty et al. 2012).

## 2.3.3 Familiarity and familiarity threat

Mautz and Sharaf (1961) state that auditors must be aware of both obvious and subtle pressures, which can act as a slow, gradual, erosion of an auditor's ability to remain sceptical in the audit engagement. The American Institute of Certified Public Accountants (AICP) code of professional conduct developed a construct that would be termed a "familiarity

threat" offering the definition that implies "that due to a long or close relationship with a person or an employing organisation, a member will become too sympathetic to their interest or to accepting of the persons work or employing organisation product or service" (AICPA, 2015, p.24). Researchers have found that over familiarity between an engagement partner and a client can decrease an auditor's willingness to challenge management on an aggressive financial position when auditing the financial statements (Chia–ah & Karlson, 2010). The acquiesce of an auditor to a clients–preferred position can be supported by a large body of knowledge in social psychology finding that social identification can significantly affect an individual's behaviour, resulting in a change of attitudes and behaviours (Ellemers et al. 2002: Hogg and Terry, 2000: Riketta, 2005). Regulators perceive that in long tenures engagement partners will start to socially identify with a client's opinions. Should that be an aggressive financial position, due to a long association the engagement partner will acquiesce to that position resulting in the financial statements being misstated.

The counter argument to long tenures leading to the financial statements being misstated is that an audit failure is more likely to occur early in the auditor—client relationship, when the auditor is less familiar with the client's processes and risk (Myers et al, 2005). Such failures may be attributable to the period when an incoming engagement partner is building up knowledge and experience (Daugherty et al. 2012). In this period the incoming engagement partner lacks the knowledge and experience to challenge management on potential misstatements found in the audit, which could reduce audit quality (Abbot et al. 2000). In mandating audit partner rotation, regulators may perceive over familiarity to be a greater threat to audit quality than the loss of client specific knowledge and experience.

Research contends that audit quality improves with long tenure as the auditor becomes more competent gaining client specific knowledge and specialization within the industry (Garcia—Blandon and Argiles, 2017: Carcello and Nagy, 2004). Audit firms have been organizing

themselves along industry lines reflecting a belief that industry specialization leads to higher quality audits. It is also motivated by a propensity of audit committees to select industry—specialist auditors (Abbott and Parker, 2000). Hammersley (2006) conducts research to determine if auditors with unique industry—specialist knowledge, allows them to detect misstatements. The method for his research is an experiment instrument that contains two cases from different industries to assess the effects of industry specialist knowledge on performance with sixty—five designated industry specialists taking part. Findings from this study suggest that industry specialisation improves audit quality. What can be interpreted from this research is that changing the audit partner could be threat to audit quality if they are not replaced by an audit partner with specialist knowledge.

Regulators perceive over familiarity between the engagement partner and client to be threat to independence and audit quality. This current study posits that familiarity of an entities systems and processes can reduce the audit risk (Carey and Simnett, 2006). Audit firms and engagement partners must control for sized and complexity in an engagement. Dodgson et al. (2020) qualitative research evidence confirms that audit firms apply more resources to complex audits such as assigning a 'relationship partner' and having senior partner involvement at the executive level. During the planning stage of an audit engagement, partners plan evidence collection based on incomplete information. In a complex audit, individual auditors collect information and run tests on only one division of the audit and don't have enough information on their own to identify if the financial statements are misstated (Hammersley, 2006). For a large entity the number of transactions can add an extra layer of complexity to this process (Werner and Gehrke, 2015). This makes the workload of the engagement partner more onerous having to judge whether the patterns of transactional data from multiple divisions constitute the financial statements being misstated. Complexity and the loss of client specific knowledge represent a threat to investors that the financial

statements might be misstated when the engagement partner is in first few years of an engagement. Which emphasizes the need for information's sharing between engagement partners so that the incoming audit partner is familiar with the transactional history of the entity.

#### 2.3.4 A fresh perspective

The expectation of regulators in implementing mandatory audit partner rotation is that the incoming engagement partner will bring a 'fresh perspective' to the audit engagement (AICPA, 2011: Carey & Simnet, 2006: Chi & Huang, 2005: Lennox et al. 2014: PCAOB, 2011: Winn, 2021) and reduce familiarity between the auditor and client (Carey & Simnet, 2006: Chi & Huang, 2005: Daugherty et al. 2012: Lennox et al. 2014). A benefit to audit partner rotation is the sharing of knowledge between audit partners which can result in increased efficacies as audit partners peer review each other's work (Dodgson et al. 2020: Gue'nin-Paracini et al. 2015: Lennox et al. 2014). What is disputed is whether shorter or longer relationships between audit partners and their clients increases or decreases an audit partners ability to challenge a client when they have taken an aggressive financial positive. Prior research states that in the first year of an audit engagement the audit partner is less likely to be able to challenge management on complex issues in the financial statements, regulators are therefore relying on audit quality increasing in subsequent engagements based on audit partners sharing their knowledge and experience as they transfer from one client to another collectively increasing their individual knowledge base and the firms.

From a client's perspective audit partner rotation ensures that the critical judgements made by auditors are the correct ones. Each rotation results in a re-examination of the treatments of accounting standards by the entity. This re–examination inevitably reassures the client that the treatment of the accounting standards has been appropriately interpreted, especially when management have multiple interpretive alternatives at their discretion giving

them the opportunity to manage earnings (Barth et al. 2008). The identifiable problem is whether an engagement partner in a longer tenure with client—specific knowledge is more likely to confront management on their interpretations of accounting standards. Social psychology asserts the view that longer tenures result in behavioural change. Regulators negatively interpret this sociological perspective when defining a 'familiarity threat' yet 'trust' in social psychology is a determining factor for increasing communications. Wilson et al. (2017) conducts research into audit firm tenure, familiarity and trust. The method was to construct an experiment using respondents as proxies for entry—level employees, finding that auditor familiarity increases trust, which in turn, increases an employee's or associate's propensity to risk communicating an opinion. This research perspective implies that as trust develops in long relationships the engagement partner is more likely to challenge management on an aggressive accounting decisions (Simpson, 2005).

Research identifies that there are direct costs associated to implementing audit partner rotation and gaining a 'fresh perspective'. These costs are associated to inefficiencies as the new partner builds up knowledge and experience (Sharma et al. 2017). Dodgson et al. (2020) found that audit firms apply more resources in the initial years of the audit engagement.

Research does not provide answers to the extent of additional costs and resources applied to rotating audit partners. What can be assumed is that, though the audit partner has been rotated the audit firm's methodology, procedures, and other engagement personnel do not necessarily change (Bamber and Bamber, 2009: Chi et al. 2009: Bedard and Johnston, 2010). Whether audit partner rotation increases audit quality is still being debated. Lennox et al. (2014) studied mandatory audit partner rotation in China, finding that audit adjustments occur more often in the incoming engagement partners first year, consistent with the ideology of a 'fresh perspective' and that the incoming engagement partner has conducted a peer review of the incumbent audit and corrected any financial reporting problems. This study indicates that

audit partners experience increased workloads in the initial year of an engagement, which can increase the costs of audit partner rotation to the audit firm.

#### 2.3.5 Internal controls

Before audit partner rotation was made mandatory audit firms were using their own internal controls systems to mitigate the risk of a long association of personal (XRB, 2018). The risks outlined in research is that in longer tenures audit partners are less likely to challenge management on opportunistic management decisions. The purpose of a 'fresh perspective' is so that the incoming audit partner reassesses the material threats to the entity and judgments made (Daugherty et al. 2012). Which indicates that an audit firms internal control procedures for managing a long association of personal could be to have another external auditors conduct a material risk assessment of the potential threats an entity faces.

International Auditing and Assurance Standards Board (IAASB) required that the name of the engagement partner to be included in the auditor's report for listed entities. The Audit regulations Act 2011 allows the Financial Markets Authority (FMA) in conjunction with CAANZ to commence civil or criminal prosecutions against audit partners in the event of an audit failure or corporate collapse. To interpret how these two regulations could affect the individual audit partner in the event of an audit failure or corporate collapse, the engagement partner is isolated by having to share litigation responsibilities with the audit firms and suffering significant individual reputational damage if proven to be legally liable. Which motivates the audit partner to ensure that the financial statements do not contain any material misstatements.

Lennox et al. (2014) describes how the incoming audit partner peer reviews the incumbents' assumptions and judgements. The legal framework that has been implemented on audit partners increases the internal challenges between incoming audit partners and incumbents. (He et al. 2015) researched the reputational damage audit partners can suffer

after corporate collapse. Finding that audit partners associated to large corporate collapses can struggle to gain employment with reputable audit firms. Which suggests that audit partners are less likely to succumb to internal pressures from the audit firm or senior audit partners to except judgments they consider material because of the individual consequences they face.

The engagement quality control review is an internal objective evaluation of the significant judgements made by the engagement team. The engagement quality control reviewer (EQCR) is the person delegated to conduct that review and should have sufficient and appropriate experience and authority to objectively evaluate the conclusions made by the audit team in formulating the audit report (ISA 220). NZX and FMC audits require the engagement partner and EQCR to be licensed audit partners. The engagement quality control review must be completed before the audit report is issued. The EQCR should discuss significant matters arising during the audit to improve audit risk judgements. The purpose of which is to induce engagement partners to conduct higher levels of testing (Bedard et al. 2008). Beattie et al. (2004) suggests that audit firms should be aware of age and experience differences between engagement partners and members of the management of the entity. It stands to reason that the EQCR should be of a similar age and experience levels as the engagement partner to competently discuss matters arising in the audit.

## 2.3.6 The loss of client specific knowledge

Regulators in mandating audit partner rotation assume an interchangeable supply of audit partners with the relevant industry experience. It is assessed by professional bodies around the globe that industry expertise is an important factor when selecting an auditor (Abbott and Parker, 2000). Audit partners with industry specialisation can be considered to outperform non–specialists in detecting errors when performing analysis (Knechel et al.

2013: Messier et al. 2008). Arthur et al. (2017) researches the effects of audit partner rotation on audit quality and whether industry specialisation moderates such a relationship. The method for this study was to use discretionary accruals and audit fees as proxies for audit quality. Findings from this study would indicate that audit quality deteriorates unless the audit partner was replaced by an audit partner with specialist experience. Hogan and Jeter, (1999) observed that audit firms have trended towards providing specialist industry knowledge with the aim of providing higher quality audits to specific industries since the 1990's. Arthur et al. (2017) study concluded that audit partner specialisation combined with audit firm specialisation has a moderating effect on audit quality, which is support evidence from (Daugherty et al. 2012: Litt et al. 2014) that audit firm specialisation mitigates the effects of audit partner rotation. Research suggests that audit firms can minimise the effects of audit partner rotation by becoming industry specialists.

From an audit partners perspective, audit partner rotation also assumes an interchangeable supply of audit clients. Audit partners that specialise in banks and insurance companies for example, could be faced with having to relocate or retrain from not having a similar client to rotate onto when their rotation falls due, retraining in this instance may be within the audit firm and could affect audit partners position of prestige (Daugherty et al. 2012). Chi et al. (2012) studied audit partner rotation in Taiwan. Taiwan provides a unique setting for studying audit partner rotation because its financial reporting regulations require two audit partners to sign the audit report. Chi et al. (2012) raised a number of significant issues in relation to audit partner rotation and their ability to remain independent in the audit engagement process such as: job security, promotion opportunities, interorganisational power, and that the retention of an important client may be more beneficial at a partner level that at the audit firm level. The regulatory problem in New Zealand is that audit partners in remote location may have been carrying out engagements indefinitely, and that the

engagement partner may be financially dependent on the audit client. By implementing audit partner rotation, regulators are ensuring that engagement partners will not become dependent on individual clients.

### 2.3.7 Cooling-off periods

Following an audit engagement in New Zealand the engagement partner is required to observe a minimum number of consecutive years away from the audit before they are allowed to rotate back (Mattock et al. 2022). Section 290.164 (b) of PES 1 states that for the duration of the cooling-off period the individual shall not consult with the engagement team or client regarding technical or industry specific issues, transactions or events. This directive from the XRB inhibits the exiting engagement partner from being able to provide any expert knowledge to the new engagement partner once they have exited the engagement. What is not covered by regulations is that before the incoming audit partners becomes the engagement partner the incumbent audit partner is able to transfer knowledge of a client's specific matters.

Mattock et al. (2022) discusses how cooling-off periods result in the engagement partner losing client specific knowledge should they be rotated back on to the client. Suggesting that the longer the cooling-off period the more client specific knowledge is lost, but only if the audit partner is to rotate back onto the entity. Ernst & Young, (2018) questioned the relevance of cooling-off periods because there is no evidence to suggest that extended cooling-off periods increases audit quality. Mattock et al.'s (2022) study provides research evidence that extended cooling-off periods marginally increases audit quality. There is very little research evidence on the effects of extended cooling-off periods and regulators do not provide specific reasons for there implementation.

Audit partner rotation regulations were first introduced in New Zealand in 2004 which is now nearly nineteen years ago. Through this period of time there has been a generational

change in the workplace from Generation X to Millennials (Adkins, 2023). Millennials are being referred to as the job-hoping generation, staying on average less than five-years with an employer (Martic, 2023). The US Department of Statistics confirms that employees on average spend less than five-years with employers. The 2018 employer survey in New Zealand found that 38% of employees had spent more than five-years with their employer (StatsNZ). What this indicates is that there is less likelihood of a 'familiarity threat' developing between an engagement partner and a client and brings into question extended cooling-off periods because employees are spending less time with employers.

## 2.3.8 Auditor negotiations

The apotheosis moment in an audit frequently occurs in auditor client negotiations.

Negotiations generally involve senior members of the audit engagement team, representatives of the entity members and the audit committee. In auditor client negotiations the engagement team will look to challenge the key assumptions and judgments that have been made by the management of the entity. The critical issue is whether managements' interpretation of accounting standards could pose a material threat to the entity. The engagement partner, EQCR and senior members of the audit team have to decide if the assumptions made by management constitute the entity taking an overly aggressive position in the financial statements. Based the material risk assessment carried out by the engagement partner and audit team, the senior members of the audit team may have to concertedly challenge the management of the company till the engagement partner is comfortable issuing an unqualified audit opinion.

Beattie et al. (2000) researched the interactions between the two primary actors in the auditor–client relationship, the finance directors and audit engagement partner. Findings from this study are that: compliance issues dominate discussions, accounting and fee issues dominate negotiations, audit committees decrease negotiations and increase discussions and

most interestingly, in the majority of cases negotiations result in a change to the financial statements. This research identifies a tripartite of actors in auditor-client negotiations. The finance director putting forward an aggressive opportunistic financial position, the engagement partner looking to constrain any earnings management and the audit committee acting as an intermediary to work towards a negotiated outcome. The results of this study motivated Beattie et al. to conduct a further qualitative study interviewing finance directors and engagement partners. Beattie et al. (2004) study revolved around the research question "how do companies and their auditors resolve important audit issues?" focusing on what role auditors play in preventing an overly aggressive financial position? Findings from this study highlighted the importance of the engagement partner to act with integrity and that the primary relationship is critical to securing a quality outcome. The first study highlighted the importance of an audit engagement team being able to remain independent when going into a negotiated settlement. What was a little disturbing was that an auditor's fees were discussed in these negotiated settlement which could be levered against the auditors to accept an aggressive financial position. The conclusions in Beattie et al.'s (2004) study highlighted the potential for an age or experience power imbalance to occur in negotiations that may influence the outcome of the audit engagement. This is of particular concern to regulatory authorities in implementing rotation policies that by necessity brings in fresh engagement partners into engagements with experienced CFO's and finance directors.

#### 2.3.9 Auditor transitions

A quintessential argument is: Does audit partner rotation weaken or strengthen an engagement partner position in auditor negotiations? Prior research identifies that audit failures are more likely to occur in the first few years of an engagement. It can be deduced from this that audit partner rotation leads to the engagement partner being at a disadvantage in client negotiations in the first few years of an engagement. Dodgson et al. (2020) describes

the context in which audit partner rotation regulations have been implemented in the US. The SOX establishes that audit partners and concurring partners must rotate every five years but does not establish any parameters for other party interactions. This gives an audit firm a great degree of latitude to manage and implement rotations as they see fit. Dodgson et al. (2020) provides qualitative research evidence of how audit firms plan for upcoming rotations. This process starts with assigning a relationship partner for complex entities and the identification of appropriate candidate to go through an interview process. Dodgson et al. (2020) states that some interviewee's mentioned that there is senior partner involvement at the executive level, and that audit firms have internal processes to ensure continuity, commitment, trust and the sharing of knowledge through the transitional process. The significance of Dodgson et al. 's research is that rotations are not a single discrete event but a carefully staged process that may or may not be the intention of regulatory authorities. Considering regulatory authorities have not looked to interject themselves further into the process implies an understanding of the complexities involved in rotating audit partners especially for larger complex entities. Dodgson et al.'s (2020) study provides qualitative research evidence that audit firms consume resources in the management of audit partner rotation regulations.

Dodgson et al.'s (2020) research states that the audit partner rotation process starts eighteen months to two years before the incoming audit partner takes over from the incumbent engagement partner and that audit partner rotation is not a stand-alone event. Pittman et al. (2022) found that audit firms are more likely to assign an incoming audit partner that has jointly conducted a prior engagement with the incumbent audit partner. Audit firms look to minimise the loss of client specific knowledge when transferring engagement partners (Dodgson et al. 2020). By assigning colleagues that have connections, the audit firm can improve the transference of client specific knowledge enabling the incoming engagement partner to hit the ground running and maintain client satisfaction levels. Pittman et al. (2022)

suggests that the incoming engagement partner is more likely to be chosen when they have shared more teamwork experience with the incumbent engagement partner and concludes that individuals with stronger connections are more likely to be appointed especially on more complex audits. Evidence from this research supports the finding made by Dodgson et al. (2020) that audit firms train the incoming engagement partner to prevent knowledge loss and ensure a smooth partner transition.

## 2.3.10 Large client entities

Corporations have been growing in size over the century with their revenue now exceeding the GDP of a number of small countries (Johnston et al. 2022). Large corporations can be trans-national with multiple subsidiaries and divisions. Auditing a large complex entity often involves a hierarchical group of auditors that have to aggregate financial information form multiple components to form the consolidated financial statements of an entity (Stewart and Kinney, 2013). The IESBA define another key audit partner as a person that audits a subsidiary or division in an entity. An other key audit partner that audits a subsidiary is easily distinguishable because of their remoteness and identifiable contributions in a corporations annual report. An other key audit partner that audits a division or department is not as easily identifiable because their contributions to the financial statement is not explicitly stated.

Section 290.164 (b) of PES 1 states that engagement partner following an engagement should not have contact with the client or provide advise to the engagement team.

Regulations do not appear to prohibit an audit partner being involved in the audit prior to becoming the engagement partner. Dodgson et al. (2020) provides qualitative research evidence that incoming engagement partners may shadow the incumbent audit partner, and that audit firms may train an incoming audit partner into the engagement partners' position.

Auditing a large client entity involves multiple audit partners, regulations do not prohibit an

engagement partner being involved in the audit prior to becoming the engagement partner.

What this suggests is that audit firms when auditing a large client entity are able to have an audit partner present on the engagement prior to becoming the engagement. Which allows them to form relations and gain an understanding of the clients specific matters.

## 2.3.11 Managing audit partner rotation

New Zealand has implemented a multi-dimensional system of audit partner rotation for engagement partners, EQCR's, and other key audit partners for different entity types with corresponding cooling-off periods. Research done by Dodgson et al. (2020) and Pittman et al. (2022) have found that audit firms manage the rotation process to reduce the loss of client specific knowledge. Lennox et al. (2014) found that audit partner rotation results in increased effort from the exiting engagement partners and incoming audit partner. Sharma et al. (2017) states that transferring audit partners results in increased costs for the audit firm. The multi-dimensional system of audit partner rotation in New Zealand and research evidence indicates that audit firms need to construct a complex system for the management of audit partner rotation. New Zealand is currently experiencing an auditor shortage which is stressing the resources of audit firms.

Human resource management provides literature on how organisation can set up matrixes and rosters for the efficient allocation of their human resources and development of employee's skills (Baird and Meshoulam, 1988: Rogstadius et al. 2011). Audit firms in transitioning engagement partners are focused on not losing client specific knowledge and retaining institutional knowledge within the engagement. Human resource management involves a set of internal policies for a firm's human capital including collective knowledge, occupational specialisation, and knowledge-based intangibles such as the abilities of employees to understand the specific needs of a client (Baird and Meshoulam, 1988: Huselid et al. 1997: Schuler and Jackson, 1987: Quinn, 1992). The internal policies of human

resource management relate to the auditing principles of institutional knowledge, industry experience and client specific knowledge.

The basic structure of a matrix in human resource management is two dimensional. More complex matrix models can encompass three dimensions or more (Belout and Gauvreau, 2003) and can be seen as a necessity for managing large global organisations (Bartlett and Ghoshal, 1998: Galbraith, 1994: Kramer, 1994). New Zealand is experiencing an auditor shortage which requires audit firms to closely monitor the portfolios and capacity of licensed audit partners and be aware of the skills and experiences, so they have to fill the roles that are available (Lammintakanen et al. 2002). Effective human resource management includes the appropriate allocation of employees to specific tasks. Assigning workloads based on financial rewards can be detrimental to the employing organisation if the employee is likely to accept increased workloads even when the impact of these workloads may have a negative effect on the employee's well-being (Baird and Meshoulam, 1988: Rogstadius et al. 2011). Management in allocating workloads must consider if the acceptance of an increase workloads will result in poorer quality outputs (Toe, 2023).

## **2.3.12 Summary**

To be critical of the literature in this review, most of the literature on audit partner rotation is quantitative with the exceptions of Dodgson et al. (2020) and Beattie et al. (2004) that provide qualitative studies. Quantitative research on audit partner rotation uses proxies as measures for audit quality (Bamber and Bamber, 2009) such as income increasing abnormal accruals, restatements, going concern opinions, and just meeting or missing earning benchmarks (Carey and Simnett, 2006: Chi & Huang, 2005: Chen et al. 2008: Myers et al. 2003). The problem with the approach of this research is that it treats audit partner rotation as a stand-alone event. Research by Dodgson et al. (2020) and Pittman et al. (2022) have discovered that audit partner rotation is not a stand-alone event but is potentially managed by

the audit firm. Which potentially explains why prior research on audit partner rotation has not been able to justify the costs that it has imposed on audit firms and consumers.

### 2.7 Gaps in the research

Mandatory audit partner rotation has remained controversial because neither regulators or researchers have provided evidence of increased benefits to justify the costs incurred to the audit profession and clients from its implementation (Bamber and Bamber, 2009: Sharma et al. 2017) which motivates more research into the subject. The central debate in research is on how longer or shorter engagement tenures affect audit quality. Researchers that argue against audit partner rotation do so stating that in the early years of an engagement partners tenure the financial statements are more likely to be misstated (Chi & Huang, 2005: Chen et al. 2008: Jenkins & Velury, 2008: Knechel & Vanstraelen, 2007: Myers et al, 2003). While proponents for audit partner rotation argue that long tenures result in a diminution of audit quality from the engagement partner being less likely to challenge opportunistic accounting decisions, and that the 'fresh perspective' of the incoming audit partner will result in increased efficiencies and audit quality (Carey & Simnet, 2006: Chi & Huang, 2005: Chi et al. 2009: Fitzgerald et al. 2018).

Audit partner rotation research has been mainly carried out in large-scale economies such as the US, China, Taiwan, Australia, South Korea. New Zealand has implemented audit partner rotation regulations similar to these larger countries. Comparatively these countries have significantly larger audit firms with increased audit partner resources. There has been very little research done on transposing regulations from large-scale economies to a small-scale economy. This research looks to fill this gap in research by interviewing audit partners to gain and in-depth understanding of how they perceive the implementation of regulations from larger-scale economies into the New Zealand audit market.

Researchers such as Dodgson et al. (2020) who studied audit partner transitions in the US, finding that audit firms assign 'relationship managers' and there is senior partner involvements in the transition process. Pittmen et al. (2022) who researched audit partner rotation in China, suggesting that audit firms train incoming audit partners into the engagement partners' position. Lennox et al. (2014) who also investigates audit partner rotation in China, finding that audit partners work loads are increase in the early years of an engagement. And Sharma et al. (2017) that studied audit partner rotation in the US, providing evidence of increased fees around rotation. All indicate that audit firms manage the rotation process. To the researcher's knowledge no research has been conducted into the management of audit partner rotation, consequently this research will interpret the experiences of audit partners to provide qualitative research evidence on how audit firms manage audit partner rotation.

Regulators negatively interpret the social psychology perspective that long relationships result in changes in attitudes and behaviours. Yet, social psychology also extolls the virtues of 'trust' in long relationships, associating 'trust' to enhance communication between organisational actors (Ellemers et al. 2002: Hogg and Terry, 2000: Riketta, 2005). Audit partner rotation periodically ends auditor-client relationships. The consequential issue is whether audit partners in longer or shorter relationships are more or less likely to confront management on opportunistic management decisions. How audit partner rotation impacts the auditor-client relationship in auditor negotiations has not been studied. The is research will investigate this issue by using qualitative research methods to discover this change to engagements partners reality.

## **Chapter 3: Methodology**

#### 3.1 Introduction

This chapter explains how the research was designed and carried out. The paradigms used in this research will be discussed as well as the epistemological approach. This section includes the methods, interview design, interview questions, sampling, data collections, data analysis, and ethical considerations.

The purpose of this study is to discover how audit partner perceive the imposition of mandatory audit partner rotation regulations and the implications. Identifying the correct research methods is critical for providing credibility to the findings (Graneheim and Lundman, 2004). What has been identified, is that most prior research in the area is mainly quantitative (archival) and comes from countries that have larger economies than New Zealand. Qualitative research is a process of enquiry that looks to uncover shared understandings based on methodological traditions that explore a social or human problem (Khan, 2014). Throughout history, auditing has been perceived to be a socially constructed phenomenon (Andon et al. 2015: Power, 2003). Bryman and Bell (2015) state that qualitative research's main focus is on words with the objective to increase contextual knowledge by exploring a social phenomenon.

What is being studied are the implications to audit partners and audit firms from a regulatory change, which impacts their social and institutional realities. Contextual constructionism objectively recognises that reality is constructed from multiple realities all of which are meaningful (Burningham and Cooper, 1999). Social constructionism assumes that no reality has precedence over another in claiming truth about a social phenomenon (Andrews, 2012). Epistemology provides for a clear link between the researcher and participant, by talking and listening to a participant in an autonomous setting a more personalised model of data collection can be established (Sanders et al. 2012). Taking an

epistemological approach this research will look to discover how the increased regulatory environment affects individuals' actors as well as the audit firm they belong to, looking to discover how they have dealt with it without judging a specific method to be the correct one.

Gray (2009) argues that deductive reasoning moves towards hypothesis testing to verify, while inductive reasoning seeks to discover binding principles to construct generalisations. This study will use themes and follow–up questions to portray generalised principles related to identifiable topics. The establishment of these identifiable topics will be used to evolve the research question (Braun and Clarke, 2006).

This research is looking to interpret the perspectives of audit partners to gain a greater understanding of the realities that they and their audit firms now face.

## 3.2 Interview design

The identified research problem is that New Zealand has transposed a multi-dimensional audit partner rotation structure from large countries. There does appear to be any literature on the implementation of audit partner rotation into a small-scale economy such as New Zealand which is why the researcher asks the first research question "how do audit partners perceive the imposition of mandatory audit partner rotation regulations? Using the qualitative method of semi–structured interviews will allow the researcher to gain an in-depth understanding of the phenomenon being studied +(Bryman and Bell, 2015).

By synthesising literature done by Dodgson et al. (2020), Pittman et al (2022), Sharma et al. (2017), and Lennox et al. (2014) which suggests that audit firms use significant resources in managing audit partner rotation, the researcher asks the second research question "what are the implications of mandatory audit partner rotation in New Zealand" to discover the changes in audit partners realities the paradigm of social constructionism is used to interpret the meanings of experiences (Andrews, 2012).

The interview design for this study is to use semi-structured interviews to gain the perspectives of audit partners centred around the research problem and explore the experiences of the participants.

Based on prior research twelve initial interview questions were constructed to ask audit partners, however, two interview questions were later discarded following the first interview based on discussions between the primary supervisor and the researcher. These two questions were determined to be repetitive, interrupting the flow of the interview. The interviews were then structured around ten open—ended questions to ask participants. The openness of the questions allowed the participants to fully express viewpoints and experiences, follow—up or probing questions were then asked based on interesting responses to pre—constructed questions (Turner, 2010).

The first six interviews were conducted by the primary supervisor and the researcher, following each interview the primary supervisor and researcher debriefed the interview discussing different points that each individual audit partner had made. From these identified topics were identified to be followed up on in subsequent interviews to gain generalisations from audit partners and to gain additional knowledge. Interviews were free flowing with interview questions not necessarily being asked in the same order. The interviewers asked probing questions to explore topics as they were discussed.

Due to geographic disbursement of interview candidates, eleven of the interviews were conducted via an online video conferencing facility 'Zoom' which allows the researcher to pre–schedule one–on–one or group face–to–face interviews and five of the interviews were conducted in person. Eleven of the interviews were carried out one–on–one and two interviews consisted of two participants. One–on–one interviews took on average 52.3 minutes and interviews that consisted of two audit partners took on average one hour thirty six minutes.

Conducting the interviews face—to—face and one—on—one, including those via 'Zoom' allowed the interviewer to paraphrase interview questions and establish a rapport with the interviewee, which encouraged an open expression of views by taking note of non—verbal cues from interviewees (Baxter and Jack, 2008). To avoid normative answers and a reluctance to provide sensitive information to the research question interviewees were assured of both confidentiality; and a verbatim transcript for them to review. They were also able to withdraw any information that they didn't want used in the study, of which three made alterations to the transcriptions and one withdrew several statements. Saunders et al. (2012) argued that a semi—structures approach is more desirable allowing participants to express more of their experiences.

Each interview consisted of ten open—ended research questions to be asked participants based on prior research. Follow—up questions were used to explore interesting topics and previous interview evidence that had been identified by the primary supervisor and researcher. This subsection will provide the reasoning behind asking the research question followed by the research question that was asked in the study.

The first question was constructed from Daugherty et al. (2012) study. This research was based on the perception that more stringent mandatory audit partner regulations could affect the quality of an auditor's quality of life and provided evidence that audit partner rotation increased the likelihood that audit partners would have to relocate. Not knowing how audit partner rotation may have affected New Zealand audit partners quality of life we ask the question:

Please describe your experiences of audit partner rotation(s)? What had happened?
 Please describe this process.

The purpose of making audit partner rotation mandatory is to sever the relationship between an audit partner and a client. It can be presumed that audit firms generally want to retain their audit clients, which is why we asked the second question:

2. How did the above rotation(s) affect your relationships with clients?

By synthesising a number of research studies: Dodgson et al. (2020), Pittman et al. (2022). Sharma et al. (2017), and Lennox et al. (2014) the researcher identified that research should be done into how audit firms manage audit partner rotation. Which is why the researchers asks:

3. From your audit firm perspective what changes have you observed taking place following the introduction of audit partner rotation? What steps have management taken to ensure that audit quality and good client relationships have been maintained?

Rainsbury's (2019) research provides evidence that there has been a near 50% reduction in domestically registered audit firms. Which indicates that significant market concentration has occurred in the New Zealand audit market. The researcher looks to discover the route cause and implication to audit firms by asking:

4. From an industry perspective what changes have you observed taking place following the introduction of audit partner rotation? What steps have management taken to ensure that audit quality and good client relationships have been maintained?

Regulators in implementing audit partner rotation are presuming that it will increase audit quality from the auditor providing a 'fresh perspective' of the audit engagement (Carey and Simnett, 2006: Daugherty et al. (2012). Prompting the research question:

5. How do you evaluate whether audit partner rotation policies have achieved their objective(s)?

Question six could be described as a probing question. It was designed so that audit partners could relay any unintended consequences of audit partner rotation, that are currently unknown:

6. Are there any implications of mandatory audit partner rotation that you had not anticipated until you experienced the rotation yourself? Please describe.

DeFond and Lennox (2011) state that introduction of a regulatory agency significantly disadvantages small audit firms. To understand this phenomenon, the researcher asks:

7. The introduction of the FMA in New Zealand can be said to have reduced the number of domestically registered audit companies. How would you evaluate this statement? What do you think might be the reason?

Researchers of audit partner rotation use proxies for audit quality (Bamber and Bamber, 2009) without making any consequential findings. How audit partners perceive audit partner rotation affecting audit quality is relatively unknown, which is why research asks:

8. How do you relate audit partner rotation policies to audit quality you have described?

Many countries have implemented audit partner rotation policies in varying forms. New Zealand has aligned itself with the US, UK, Australia, and Canada. In contrast to this approach the European Union has mandated audit partner rotation at seven—years on followed by a two—year cooling off period. This motivates the research question:

9. Do you feel like audit partner rotation policies are suitable in their current form? What recommendations would you make to regulatory bodies regarding there suitability and the appropriateness of timeframes?

Following a standard qualitative research design, the last question that is asked is:

10. Before we conclude this interview, is there anything else you would like to add?

## 3.3 Data collection and analysis

## **3.3.1 Populations and Sample**

The regulatory situation in New Zealand is that domestic audit firms and auditors are registered through CAANZ and the FMA authorise the issuance of licenses to overseas auditors and audit firms. NZX, FMA and OAG's entities are the only entities that are subject to audit partner rotation regulations which must be carried out by a licensed audit partner and licensed quality control reviewer who belong to a registered audit firm. Consequently, the target population for this study are licensed audit partners who belong to a registered audit firm.

The New Zealand Companies Office maintains a register of all New Zealand licensed audit partners and registered firms which is supplied to them by CAANZ and the FMA. On the 1<sup>st</sup> of August 2022, a complete list of New Zealand licensed auditors was downloaded from the New Zealand Companies Office website. This sample included 143 Licensed audit partners from New Zealand, 28 licensed audit partners from Australia and one from Singapore.

The Auditors Register – New Zealand Companies Office provides the name of the licensed auditor, company which they work for, address of the company, employee status details (e.g., partner, director, employee) and email address. Each audit partner on the Auditors Register – New Zealand Companies Office was contacted by email and sent an introductory email that informed potential participates (that participation was voluntary and all information would be treated confidentially), a consent form, an information sheet describing the study, and a set of interview protocols.

The table below presents the results from emailing 172 audit partners from the Auditor Register at the Companies Office. Thirteen audit partners agreed to be interviewed, eleven of these interviews were conducted individually and one interview was carried out

with two participants. A referral resulted in another two licensed audit partners agreeing to be interviewed which was carried out simultaneously. This gave the researcher a total of thirteen completed interviews consisting of fifteen interviewees, a participation rate of 8.72% or 10.48% for audit partners that reside in New Zealand.

Table 4: *Interview Demographics* 

Panel A: Locations of audit partners		
	Number of Partner Interviewees	Percentage of Partner
	$\underline{(n=15)}$	<u>Interviewees</u>
Auckland	8	53.33
Christchurch	4	26.67
Wellington	2	13.33
Hastings	1	6.67
Panel B: Distribution between Big Four and non – Big Four		
	Number of Partner Interviewees	Percentage of Partner
	(n = 15)	<u>Interviewees</u>
Big Four	7	46.67
Non – Big Four	8	5.33
Panel C: Industry expertise*		
	Number of Partner Interviewees	Percentage of Partner
	$\underline{(n=15)}$	<u>Interviewees</u>
Financial Services	6	23.07
Insurance	4	15.38
Manufacturing	5	19.23
Government sector	4	15.38
Other (real estate, not-for-profit, Retirement villages, technology)	7	26.92
*Some audit partners may have more than one area of expertise		

The primary empirical data for this research project was obtained from conducting semi–structured interviews with New Zealand audit partners. Semi–structured interviews allow for flexibility in the interview approach and the possibility to obtain rich information and a deep understanding of the issues at hand (Bryman and Bell, 2015). Interviews were transcribed by the researcher and sent to participant for their approval. These finalised transcriptions provided the data for results and analysis.

## 3.3.2 Data Analysis

This research took a thematic analysis approach for processing the results of the raw data. Thematic analysis was used for its flexibility in processing raw data and approach to meaning generation from patterns within the data set (Braun and Clarke, 2017). Following each interview, the researcher and primary supervisor discussed the interesting topics that had been raised within the interview, the researcher then looked to expand on these topics with future participants. By using this method distinctive patterns evolved within the research with audit partners providing perspectives that could be generalised or contrasted within different categories. In many cases this represented a tangible–extensions of knowledge on the identifiable topic with audit partners communicating different aspects on the topic unique to their own or audit firms' perspective.

Ten open—ended questions had been sent to participant in advance and it was conceived (though not confirmed) that the participants had prepared for the interview. The interview method meant that not all participants responded to the follow—up questions.

Though not all interviewees directly responded to these questions interviewees had often referred to these topics in the conversations, which allowed the researcher to generate themes inside the topic areas discussed.

Thematic analysis has been used to analyse the data, the first step in analysing the data was to search the interviewee transcriptions to identify generalised topics of discussion, these generalised topics of discussion were put together in the excel sheet to form themes. The open-ended research questions were structured based on gaps in the literature and identified demographic problems. To answer the research questions, themes were put into buckets to form central topics that were present in the open-ended research questions that were asked in the interviews with audit partners. This process resulted in three central topics: Implications to the audit firm, planning and engagement mechanisms, and engagement partner client relationships.

Themes were developed by conceptualising 'central organising concepts' that came from the constructed open—ended questions and probing questions. Qualitative research can produce large amounts of data (Pope et al. 2000). Being an inductive exploratory study transcripts were read and then re—read for any identifiable themes and then coded diversely not taking into account themes that previous research had identified on the topic (Hollway, 1989). Codes were collated into central themes in systematic fashion across the entire data set to collect as much relevant data to each code. Following the constructionist paradigm, a latent approach is adopted. Themes and patterns were 'mashed up' together to gain a greater understanding on the conceptualised topic and a description of the stated reality (Braun and Clarke, 2006).

The analysis resulted in four identifiable categories amongst participating audit partners: Big Four, non-Big Four, managing partners, and new audit partners. The objective of identifying these was to perform additional analyses and present different perspectives within the dataset. This also allowed the researcher to make comparisons of how these different categories had dealt with the imposition of a mandatory audit partner rotation policy and present different viewpoints of the construction of a regulatory agency.

To gain a deeper understanding of the data, how participants phrased certain answers was analysed with the categories of positive, negative, moderate being assigned.

#### 3.3.3 Ethical considerations

This research was approved by the University of Canterbury's Human Research Ethics Committee and considered to be low risk research. Though this research was considered to be low risk it was assessed that some questions or topics discussed may involve sensitive information. The introduction of specific regulations may have led to negative situations occurring in the interviewee's personal lives. Information was provided to the interviewees in the information sheet explaining not required to answer any questions that they consider sensitive in nature or make them feel uncomfortable in any way. Participants were also advised that participation was voluntary, and they had the wright to withdraw from the research project at any stage, up until a six—week period following the interview and they had been given the opportunity to comment on the transcript where they were able to withdraw all or parts of their information.

To ensure the reliability and validity of the research data the aims, purpose, and interview process were well presented on first contact with the participants. Emails to potential participants contained an introduction to the research project, consent form, information sheet, and interview protocols. Participants were required to return the consent form before the interview took place and the information sheet contained information that advised the participant of the interview process. Should a participant have withdrawn their interview information the raw materials collected in the interview would have either been returned or destroyed.

To ensure confidentiality and anonymity codes were assigned to replace the names of the participants and firms (e.g., interviewee 1 of Company A). Furthermore, no locations or identifying material will be revealed. To ensure the identity of participants wasn't known to anyone outside the research team, we will keep your signed consent form in a file separate from your interview transcript. All data will be destroyed after five years following the completion of the study.

Interviews were either recorded using a portable recorder or using the Zooms audio–recording feature. One interviewee did not initially consent to be recorded via the Zooms audio–recording feature, but then gave consent at the time of the interview after the interviewee explained that all information was to be kept confidentially and his identity would remain anonymous.

## **Chapter 4: Findings**

#### 4.1 Introduction

The purpose of this chapter is to present the results from the semi-structured interviews with New Zealand audit partners. The research question "how do audit partners perceive the imposition of mandatory audit partner rotation regulations?" will be answered by the responses of audit partners from the different categories and their perceptions of the imposed regulations. The second research question "What are the implications of mandatory audit partner rotation in New Zealand?" will be answered by providing qualitative research evidence that describes the consequences to audit partner and audit firms from regulators implementing rotation requirements.

Participant responses have been grouped together into threads of conversation to provide perspectives on prior literature and in some instances, new qualitative research evidence has been contributed. What the researcher envisioned was that each interviewee would add to the body of knowledge on a subject area. Interviewee responses have threaded together to form meaning around a subject area. This is way the interview evidence has been presented. Where more than one interviewee has been sited (Interviewee 1NonBig4, 2Big4, 3NonBig4, 4Big4) indicates that each of these audit partners have said something very similar or have confirmed a participant's response.

### 4.2 Implications of audit partner rotation on audit practice in New Zealand

Before audit partner rotation was made mandatory, audit firms were required to mitigate the 'familiarity threat' and long association risk as part of their internal controls. This requirement in the ethical standards, while not mandating rotation created an understanding that over familiarity was a threat to independence. The UK's Financial Reporting Council (FRC) provides statistical evidence that the number of registered audit firms worldwide have been reduced. Rainsbury (2019) states that there has nearly been a 50%

reduction in New Zealand domestically registered audit firms providing evidence of significant market concentration.

In New Zealand there approximately 500,000 small businesses, 10,000 medium sized businesses and 2,500 large businesses (Roberts, 2021). Making audit firm demographic environment one of mainly small to medium-sized entities with a few large client entities. New Zealand has transposed audit partner rotation regulations from large-scale developed economies. Regulators should consider 'right-sizing' regulations before they are implemented (Quack and Shubler, 2015). Rainsbury (2019) and Haughton et al. (2013) found that regulations disproportionality disadvantage small and medium-sized audit firms due to increased costs. DeFond and Lennox (2011) states that regulators are likely to target small audit firms in the first few years that they commence operations.

The section of the chapter will be set out as follows: Firstly, that audit firms were already rotating their audit partners. Secondly, audit partner rotation and market concentration. Thirdly, audit partner rotation effects both Big Four and Non-Big Four audit firms. Fourthly, the practicality of audit partner rotation regulations. Fifth, audit partner rotation puts pressure on an audit firms licensed audit partner resources. Sixth, large complex clients require significantly more audit partner resources.

#### 4.2.1 Solidifying audit partner rotation policies

Making audit partner rotation mandatory is based on the notions that engagement partners in long relationships are less likely to challenge management on an aggressive financial position. Interview participants communicated that audit firms in New Zealand were already rotating their audit partners based on this notion: "So, rotation as a concept is something that most firms had an internal policy on even if there wasn't necessarily a hard and fast external policy" (Interviewee 1, NonBig4). Audit regulations pre—audit partner rotation in New Zealand stated that audit firms must mitigate a long association risk. Audit

partners confirmed that this was being implemented: "We have a policy of reviewing rotation after 7 to 10 years" (Interviewee 11, NonBig4) and they perceived long tenure to be of concern "familiarity over ten years we say you can't do it" (Interviewee 2, Big4). Interviewee 9, NonBig4 provided the perspective that: "There are a few people out there that operate the same client for twenty years, so I think that is a bad thing". Participants provided the perspective that the implementation of audit partner rotation regulations was a solidification of the regulations, and the regulatory environment was just catching up with what was happening in the industry. Participants in this study, shared the same values as regulators perceiving that long tenure is a threat to independence.

## 4.2.2 Audit partner rotation and market concentration

Prior research by Rainsbury (2019) associated the reduction of domestically registered audit firms in New Zealand to the implementation of a regulatory regime. Participants in the research resonated such views: "What you're seeing these days is that there are [sic] less and less firms doing audits...it's quite hard for small firms and they have started just saying no to doing audits" (Interviewee 4, Big4). "The FMA said at the start we're not going to force the small players out. They've just made the rules such that people decide to, because it's too difficult to comply" (Interviewee 9, NonBig4). "For firms like us, if they keep making the rules harder, it may get to the point that firms in the mid-tier start to drop out. I think certainly the firms that have dropped off so far have probably been the smaller ones" (Interviewee 10, NonBig4) and it is audit partner rotation that is impacting market concentration "So, what they're doing is limiting the licensing market to the Big Four....you're talking five or six firms that can probably manage the rotation easily, for the rest it's challenging" (Interviewee 11, NonBig4). "There's been a significant decrease in the volume of small firms doing audits" (Interviewee 10, NonBig4). Non-Big Four audit partners comments were critical of the effect that audit partner rotation had on small audit firms. Audit

partner rotation can also be considered a tool for the regulatory regime to use, to increase market concentration in a small-scale economies.

Haughton et al. (2013), Rainsbury (2019) state that the implementation of a regulatory regime disadvantages small audit firms: "I think certainly the firms that have dropped off so far have probably been the smaller ones that maybe on hindsight, decide they didn't want to be part of the regime, and you can well understand that, because it's very onerous" (Interviewee 10, NonBig4) "there's a couple of people that we know, that have withdrawn from being licensed..... both of them in fact, are mid-tier. If they're not top 10, the top 12 firms globally that are represented in New Zealand" (Interviewee 11, NonBig4). The consequences for some entities that need to be audited are: "They are going around town almost saying, will anyone do my audit and all the Big Four are going 'no' we won't" (Interviewee 5, Big4) "We have turned clients away" (Interviewee 10, NonBig4). The entities that appear to be getting mostly effected are: "We have turned down some very small, not for profits" (Interviewee 11, NonBig4) "it's very uneconomical for us to audit a lot of charities" (Interviewee 4, Big4). What can be interpreted from the comments made by participants is that regulators regime has had unintended impacts on the supply of audit services in New Zealand. Entities that are legally required to be audited should have access to auditors at a reasonable price and have their audit completed within a reasonable time frame.

## 4.2.3 Audit partner rotation effects both Big Four and Non-Big Four audit firms

Prior research identifies that the inception of a regulatory regime significantly impacts smaller to mid-sized audit firms. Participants from non–Big Four audit firms put forward the critical perspective: "From a practice management point of view it has been difficult for the mid, smaller sized firms simply because you don't have that many partners in each geographic location" (Interviewee 1, NonBig4)."We can make it work, it's not ideal, if you were a small firm, you've just got to give up jobs" (Interviewee 6, NonBig4). "Life would be

simpler without this" (Interviewee 9, NonBig4). Non–Big Four audit partners were in negative agreement that audit partner rotation adversely affected them. These comments also indicated that audit partner rotation was causing small audit firms to leave the domestically registered audit market.

Participants from Big Four audit firms confirmed that audit partner rotation was disproportionately affecting small audit firms: "It would be more challenging for a small firm with fewer licensed auditors" (Interviewee 2, Big4). "Maybe in smaller firms it's quite difficult, when you've only got one or two partners" (Interviewee 4, Big4). Managing partners from Big Four audit firms provided the perspective that it: "Is making sure that you've got a sufficient partner resource to be able to rotate on and off appropriately, and to have the right skill sets, to deliver high quality audits" (Interviewee 5, Big4) stressing the importance that "being able to have enough sufficiently qualified and experienced people to do the various roles within the context of partner rotation" (Interviewee 2, Big4). The context of Big Four audit partners was one of providing the right audit partner based on experience and industry knowledge, which could be contrasted against the views of non-Big Four audit partners whose concern was having sufficient partner resources to provide their clients with an audit partner and EQCR.

## 4.2.4 The practicality of audit partner rotation policies in New Zealand

Audit partners questioned whether audit partner rotation regulations were practical in the New Zealand setting, conveying that: "New Zealand is a small economy that has small scale commercial firms...... and smaller scale audit firms and practices" (Interviewee 1, NonBig4). Interviewee 10, NonBig4 provided the perspective that audit partner rotation is likely to work well for audit firms with significant partner resources but caused problem for audit firms with limited partner resources. Interviewee 1, NonBig4, made the generalised comment regarding implemented regulations: "Some of the stuff they are trying to implement

here, which really has an international focus and frankly is impractical to do in New Zealand commercially" "because it's just hard to see the relevance" (Interviewee 3, NonBig4).

Participants critically evaluate the implementation of audit partner rotation regulations because of the international focus of regulations and audit firm size. What these comments express is that regulators in implementing audit partner rotations regulations need to consider the demographic variables of a country, such as GDP and population size.

Participants perceived that New Zealand follows overseas when implementing regulations: "The bigger thing I'm struggling with in New Zealand, we are standard takers not makers" (Interviewee 6, NonBig4). "New Zealand does tend to follow overseas. I think being a follower in this stuff is quite good because we're smaller anyway" (Interviewee 9, NonBig4). "One thing else I've noticed over the past ten years plus......whenever a new standard comes out, we want to adopt it, we don't want to be dragging the chain or being seen as a second-class citizen" (Interviewee 10, NonBig4). The question is whether the regulations are suitable for the New Zealand audit market: "Those people that are driving it are sitting in very large firms from large countries and the application in New Zealand is quite different and quite difficult" (Interviewee 11, NonBig4). "An international standard comes in and we talk to our clients, but we have to adopt it, yet sometimes it doesn't quite fit New Zealand" (Interviewee 6, NonBig4). Audit partners were not in agreement "I believe that New Zealand should be a standard taker....we should just take international standards and double-check to make sure it applies to New Zealand" (Interviewee 9, NonBig4) because "you can kind of observe what's happening overseas" (Interviewee 4, Big4) contrasted against: "Both in accounting standards and auditing standards we are far too fast to implement things that are international, that are written for large scale international markets and corporates and are really not applicable" (Interviewee 1, NonBig4). Some participants perceived that the New Zealand regulators were implementing regulations without

considering whether they are practical in the New Zealand setting. While other participants perceived that there were advantages from adopting regulations from overseas.

# 4.2.5 Audit partner rotation puts pressure on the audit firm to increased their licensed audit partner resources

Audit partners communicated how the management of audit partner rotation was directly related to the amount licensed audit partners that an audit firm has and that audit partner rotation significantly impacted audit firms with only a small number of licensed audit partners: "The reasons that we merged was that we actually saw the regulations, rotations and FMA coming into play as being a need to really have more partners, because of the rotation issues" (Interviewee 11, NonBig4) "two things have tended to happen, one is there has either been some mergers or moving around of the firms, or they have loosely joined together to contract each other in" (Interviewee 1, NonBig4). Interviewee 9, NonBig4 potentially expressed this best, stating: "For auditor regulations our belief is that without four licensed auditors you can't comply. So, we have four licensed auditors and I want to bring in a fifth". This point of view was reiterated by Interviewee 6, NonBig4: "Because to have auditor rotation you would need to have at least four audit partners in a firm, because.... it's not just the lead audit partner, it's the quality control audit partner that has to rotate as well". The consequence for audit firms with less than four audit partners was that: "You get one of the small firms saying, I'll be the quality control partner on yours and you can be the quality control partner on mine" (Interviewee 6, NonBig4) also expressed by "I am aware that some of those firms use partners from another firms to be the EOCR for example on their jobs" (Interviewee 1, NonBig4). Interviewee 6, NonBig4 then made the critical observation: "Where does the liability land if things go wrong" highlighting potential litigation problems in the event of an audit failure. The requirements for becoming a licensed audit partners is 3000 hours auditing work, 750 of those hours has to be spent auditing FMC

or NZX entities creating difficulties for non-Big Four audit that have limited NZX and FMC clients (Interviewee 11, NonBig4).

## 4.2.6 There are significantly more resources involved in auditing large complex entities

New Zealand can be described as having mainly small and mid-sized entities with a few large commercial entities. Participants discuss the amount of audit partner resources required for different sized entities: "Most audits would have one partner, maybe with a EQCR as well" (Interviewee 4, Big4) "in New Zealand you'd have a signing partner.....and the quality control partner" (Interviewee 5, Big4). Participants from Big Four audit firms explain how larger more complex clients involve significantly more audit partner resources: "You have one signing partner, and then you have about six to seven partners supporting them" (Interviewee 4, Big4). A significant issue for Big Four audit firms was that audit partner rotation reduced their ability to provide the most suitable engagement partner with industry experience: "It's quite a small pool of partners that can actually do some of these jobs" (Interviewee 12, Big4) "we'd have probably four partners in New Zealand to be honest, with sufficient maturity and experience to be able to be put up for that panel" (Interviewee 2, Big4). "I doubt there is more than ten of us that are truly experienced in auditing a bank or maybe it's fifteen" (Interviewee 5, Big4). What can be interpreted from participant responses is that small and mid-sized entities only require and engagement partner and EQCR. Larger more complex clients involve multiple partners, significant to audit partner rotation is that New Zealand has a very small pool to auditors that can audit large complex clients.

Auditing large complex multinational entities involves group aggregating information from different audit teams across different industries (Stewart and Kinney, 2015). Group auditing research provides example of auditors drawing on specialists within the industry (Trotman et al. (2015). Participants provided qualitative research evidence that large complex clients involve several audit partners, with specialist skill sets: "I normally have a tax

specialist, a treasury specialist, we call the financial risk management specialist, an accounting specialist for some of our complex accounting standards, and we might have a corporate finance specialist" (Interviewee 8, Big4). "When you think of specialists, partner inputs, some of these audits will have tax technology professionals, valuation specialists" (Interviewee 5, Big4). Bank audits are configured: "Let's use the bank for example, we would typically have the lead partner/signing partner, we might have a second audit partner as well, who would probably have responsibility for treasury, we would then have a valuations partner, so they would value all the derivatives, then also but not always a partner...... loan provisioning and then depending on if they've got insurance.....an IT partner...... we don't have tax partners the same way" (Interviewee 12, Big4) and large complex audits have a second partner who signs, we've got a second partner who's involved in quite a lot of details" (Interviewee 2, Big4). Participants from generally Big Four audit firms provide a structure for auditing large complex clients consisting of an engagement partner, an audit partner second in charge and other audit partners that head specialist areas within the engagement.

Managing partners perceive: "We have problems or difficulties in terms of coordinating audit partner rotation just with the number of different roles that we have, and then also the different roles in an engagement" (Interviewee 2, Big4) "because you've got multiple partners in there, you'd never design your model as a firm where you had all of them starting at the same time, all leaving at the same time" (Interviewee 5, Big4). The purpose of which: "So that a lot more institutional knowledge is present in the year that the partner comes on" (Interviewee 5, Big4) because "if they all have to rotate roughly around the same time, you have a little bit of institutional knowledge loss" (Interviewee 7Big4). What the audit firm is trying to achieve is: "They will have brought through a more junior partner who can be the continuity" (Interviewee 8, Big4, 5). Interviewee 5, Big4 explains how: "You might be

losing partner A as your signing partner, but actually, there's every chance one of the other partners you've known for the last three years might be coming into that signing partner role, or if a new pattern is coming in, you've still got partner B or partner C, who was the second partner on the job anyway" and "You don't want your EQCR and your lead partner rotating off at the same time" (Interviewee 8, Big4). large entity audits can be considered to be: "They are complex beasts" (Interviewee 7, Big4) "if you think about it logically on a big job, it takes you longer to get to grips with it" (Interviewee 8, Big4). To summarise, audit firms have multiple audit partners involved in auditing a large complex clients that rotate on and off at different stages, so that the audit firms retain institutional knowledge in the engagement, providing an example of how audit firms manage their human resources (Rogstadius et al. 2011)

## 4.3 Planning and coordination mechanisms

Through the last section of this research, participants provided qualitative research evidence that audit partner rotation disproportionately affects small and mid-sized audit firms because of limited licensed partner resources. At the other end of the spectrum Big Four audit firms have difficulty providing audit partners for all of the roles they have available. Both Big Four and non-Big Four audit firms focus on providing audit partners with industry experience and retaining institutional knowledge within the engagement. What participants have communicated is that audit firms have to carefully manage audit partner rotation so that audit quality is not affected.

This sections of the chapter will be set out as follows: Firstly, audit partner rotation involves a complex matrix. Secondly, transitioning audit partners is complex. Thirdly, an engagement partners capacity and portfolio. Fourthly, rotating the EQCR challenges an audit firms' resources. Fifth, engagement partners and client specific knowledge. Sixth, the audit

team drives audit quality. Seventh, the more complex the entity the more time on an engagement partner needs.

## 4.3.1 Audit partner rotation involves a complex matrix

What was identified early on in this study was that audit firms have implemented varying forms of matrixes or logs to manage audit partner rotation: "It's actually quite a complex matrix to get it to work, so there's actually more complexity than I think the audit regulators probably envisaged" (Interviewee 1, NonBig4). "I think it's a complex matrix.... even in a large firm like ours it is complex" (Interviewee 12, Big4). "We have created a log to monitor movements of rotating partners" (Interviewee 3, NonBig4). The complexity of the matrix comes from: "Managing the myriad of different rules" (Interview 7, Big4). "It's added a layer of complexity for the firm to manage.... if it's a public listed company, a five-year plan, if it's an OAG audit a six-year plan, and for an FMC audit a seven-year plan" (Interview 8, Big4). Audit firms also have to manage rotation for the EQCR (Interviewee 1, NonBig4) "and other key audit partners on the engagement" (Interviewee 7, Big4). That audit firms use a matrix to manage audit partner rotation is new to research and a key finding of this study.

Big Four audit partners describe how they use: "A global rotation database" (Interviewee 7Big4). "We also run a global rotation process" (Interviewee 5, Big4) that can capture "ten years for your signing partners, your engagement quality review partner, any other key audit partners involved in the engagement, and also increasingly the managers on the engagement" (Interviewee 7, Big4) and it is a "reasonably sophisticated systems where we're planning our rotations, two or three years out" (Interviewee 5, Big4). Consistently, both Big Four and non-Big Four audit partners talk about planning for future rotations: "We are looking two to three years out" (Interviewee 2, Big4). "What is forecast to happen in the next three or four years" (Interviewee 3, NonBig4). "We will plan out the next five or seven

years" (Interviewee 6, NonBig4,) then "every three months it gets reviewed and approved" (Interviewee 5, Big4). because partners rotate off and retire (Interviewee 7, Big4). Audit partners from Big Four audit firms discuss the: "Difficulties in terms of coordinating audit partner rotation just with the number of different roles and different engagement types" (Interviewee 2, Big4) and "even though we are one of the bigger firms, it's still a finite group" (Interviewee 4, Big4). The findings from this section identify that audit firms use a complex matrix to manage the multi-dimensional structure of rotation in New Zealand. Consistently, both Big Four and non-Big Four audit partners talk about planning rotations years in advance, and regularly updating the matrix. Where these two groups differed: Big Four audit partner discuss a global rotation system, while some non-Big Four audit partner refer to the matrix as a log, indicating that Big Four audit firms have a three-dimensional matrix while non-Big Four audit firms are likely to just have two (Bartlett and Ghoshal, 1998: Galbraith, 1994: Kramer, 1994).

# 4.3.2 Transitioning the engagement partners involves a number of complexities for the audit firm

Research done by Dodgson et al. (2020) investigated how audit firms transition engagement partners from client to client in the US, finding that audit firms use a relationship manager and that other senior audit partners are involved in the transition process.

Participants stated that audit firms were involved in transitions of audit partners. Findings suggest that there were distinctive differences between how Big Four and non–Big Four audit firms managed the process and that transitioning audit partners was incorporated into the matrix of audit partner rotation.

The process for non–Big Four audit firms started with: "The new partner sitting down with the existing partner or senior manager, who is most familiar with a client and works through what's involved" (Interviewee 3, NonBig4). The process involves talking about

differences and questioning the treatment off balances: "I'd definitely be talking to the other guy and saying why did you do it that way, have I missed something" (Interviewee 6, NonBig4). Non–Big Four audit partners also thought it was: "Getting the matrix right so you've always got some consistency of the skills" (Interviewee 1, NonBig4) stressing the importance of having either the engagement partner, EQCR, or audit manager having industry experience "we try and keep the engagement manager and team.... on the job.... that helps with institutional knowledge" (Interviewee 11, NonBig4). Interviewee 11, NonBig4 describes how the incoming partner will attend the: "Planning meeting, go to the concluding meeting, and meet the board.... then rotate onto the engagement the following year". The above interview data shows that there is a transfer of knowledge between the partners and that they prioritise retaining experience and institutional knowledge within the engagement. By transitioning the engagement partner audit firms are minimising the loss of client specific knowledge and threats to audit quality (Dodgson et al. 2020: Pittman et al. 2022).

Participants from Big Four audit firms describe a more formalised approach to transitioning engagement partners. Managing Partners select candidates for upcoming engagements based on the audit partner: "Having the right industry experience" (Interviewee 5, Big4) "availability, background and experience" (Interviewee 2, Big4) stressing that "it's important for us that we manage the transition well in terms of both client expectations and timing" (Interviewee 2, Big4) having "a conversation with management and the audit committee chair around the types of skill sets that they have found valuable" (Interviewee 5, Big4) "giving them a panel to look at for example" (Interviewee 2, Big4). Clients: "Understand the importance of having a relationship that they can work with" (Interviewee 2, Big4) "the incumbent partner would be involved in that process to the extent that they have a good feel for the type of personality, and the chemistry that's going to be needed for a particular client" (Interviewee 5, Big4). This studies concurs with Dodgson et al. (2020)

research that audit firms provide clients with a selections of audit partners to choose from and that senior partners are involved the traditional process. What participants are repetitively emphasising is the replacement partner must have industry experience and be able to have a relationship with the client (Abbott and Parker, 2000).

A significant finding from this study is that audit firms match the personality type of the engagement partner to the client. Interviewee 4, Big4 explains: "You get to meet them, just to make sure they're a good fit, because of different personalities, we're putting the right personalities with the right clients. Because what you want is tension, but not too much tension" and having a relationship is important for completing the audit: "We can understand that the audit partner needs to be a cultural fit and it's important that you have a relationship, otherwise you can't do the audit" (Interviewee 7, Big4). The engagement partner must also have the industry experience and be able to constrain a clients aggressive financial position: "I've seen situations where a partners comes off rotation, ideally suited technically, but also ideally suited personality wise to a client that pushes the envelope, and that client needs a strong partner..... we've rotated them off, and the option is a partner that perhaps is less well-suited" (Interviewee 8, Big4) "we do a lot of internal work here and we call it business chemistry" (Interviewee 4, Big4). Big Four audit partners through this sections describe some of the complexities to the engagement partner client relationship. The engagement partners must have a relationship with the client to transfer information and complete the audit, have the technical knowledge or industry experience, and be able constrain a client's aggressive financial position. Emphasising the importance of matching personalities and a matrix for audit partner rotation.

The audit partner then meets with the client: "Clients like to consider who will come on and meet with them as an audit committee chair, just to make sure their comfortable, that works well" (Interviewee 2, Big4) "then the firms would typically share candidates with them

to either meet" (Interviewee 5, Big4) "it could be as simple as bringing them along to a meeting" (Interviewee 4, Big4). Though critically: "I am reluctant to get too much into a beauty parade, here are three you can pick from. I'm not saying we don't do that, because sometimes there is a need" (Interviewee 7, Big4). Big Four audit firms start the process: "We would be planning our rotations twelve to eighteen months out" (Interviewee 7Big4) "we would typically start that process, twelve to twenty-four months ahead" (Interviewee 12, Big4). Participants in this study echo Dodgson et al. (2020) research that transitions start eighteen months to two years out.

Once it's established who the new partner will be on the engagement, that partner will then be involved in shadowing the incumbent partner to get familiar with each other (Interviewee 12, Big4). "What we would like to do is involve them in a couple of shadow meetings, they won't charge any time and won't make any decisions, to make sure that the transition works" (Interviewee 7, Big4) so that they become aware of the issues and to "start building those relationships" (Interviewee 12, Big4) "it makes sense that both the transition partner and the current partner are involved in the judgments" (Interviewee 13, Big4). "It gives the partner a chance to come to clearance meetings and have a little bit of an overlap, so you can do an efficient hand over" (Interviewee 8, Big4). What's important: "Is when we manage the process of transition is to ensure that when we go from one partner to another partner, we are not losing knowledge, the really fundamental knowledge of the issues" (Interviewee 5, Big4). Interviewee 4, Big4 provides the perspective that: "Often the exiting partner is still around, it's not like knowledge is lost, but you do actually put up those walls as well with the previous partner because you're a new partner coming in". The purpose of the transition process is so that the new partner: "Is ready to pick it up straight away, so they are not starting cold" (Interviewee 12, Big4). Audit firms in managing the transitional process are assuring the client that knowledge is not being lost, that the incoming audit

partner will have already established relationships and have understanding of the judgments being made. The purpose of this is so that the incoming audit partner will hit the ground running when they become the engagement partner (Pittman et al. 2022).

### 4.3.3 Audit firms need to consider an engagement partners capacity and portfolio

In constructing a matrix to manage audit partner rotation the audit firm must consider an audit partners portfolios: "As you start to broaden it out, you are starting to look at the manageability of total portfolios for audit partners" (Interviewee 1, NonBig4) which is determined by "the capacity to come on to that job, how does that fit with their other portfolios" (Interviewee 12, Big4) because it's all about timing and the portfolio's we have (Interviewee 3, NonBig4). What has to be considered is: "How many clients can an audit partner realistically have or should have" (Interviewee 1, NonBig4). Big Four audit partners communicated that an audit partners portfolio is managed by a: "Risk and quality team" (Interviewee 2, Big4) to ensure "we don't go over our five years or seven years" (Interviewee 12, Big4). Managing partners are involved in the allocation of engagement partners to clients: "We make a recommendation, we'll look at people's portfolios, availability, background and experience, and make that assessment" (Interviewee 5, Big4). "One of the things you've got to do is actually find out what experience you've got" (Interviewee 6, NonBig4) and that they "haven't got any conflicts" (Interviewee 12, Big4). To accommodate a significant client, auditors may be required to: "Go through processes where you might rotate early....to free up capacity" (Interviewee 13, Big4) with the planning starting "five years out" (Interviewee 12, Big4) to ensure that the partner has the right experience to go onto an engagement. A consequence of audit partner rotation and the auditor shortage is that audit firms need to closely manage and audit partner portfolio capacity to ensure that the partners work life balance is maintained (Lammintakanen et al. 2002).

## 4.3.4 Rotating the engagement quality control reviewers challenges an audit firms resources

Audit regulations stipulate that the audit partners, engagement quality control reviewers (EQCR), and other key audit partners are subject to rotation regulations adding additional complexity to the matrix. Audit partners discuss the impacts of rotating the EQCR: "I think it's made managing the business hard, because what you need to think about is, are the audit partner and EQCR going to have to roll off in two years' time" (Interviewee 8, Big4). "The other thing is timing if your quality review partner and the engagement partner have to rotate at the same time, that's challenging.... and the other really problematic thing is on your really large audits, is you also have other key audit partners" (Interviewee 7, Big4) increasing the complexity of the rotation matrix. Audit firms also need to consider client competitors when replacing audit partners, Interviewee 8, Big4 explains how a client will not appreciate an audit partner being moved onto a direct competitor following rotation.

Little research has been done on the involvement of the EQCR in the audit engagement. Interview participants discuss the role of the EQCR: Interviewee 1, NonBig4 stated that in the engagement process the EQCR should have no contact with the client, which was then confirmed by: "The other partner is what we call an EQCR who isn't client facing" (Interviewee 12, Big4) and what regulators need to consider is "the risk with them becoming too close to the crime doesn't exist" (Interviewee 5, Big4) relating the EQCR to becoming over familiar. Audit partners perceived that having an EQCR was one of the tools to address familiarity: "The aspect of familiarity is addressed in different ways, part of that is having the EQCR" (Interviewee 13, Big4) because the purpose of the EQCR is to: "Challenge the engagement team and ensure quality" (Interviewee 12, Big4). Managing partners provided the perspective that: "It's important that you have another perspective, we have that to a certain extent with the EQCR" (Interviewee 2, Big4). "Having a slightly longer

period and more familiarity with the entity and more history with the issues that have arisen is probably going to enhance your audit quality" (Interviewee 5, Big4). Participants in this study provide qualitative research evident that the EQCR has no contact with clients and that the EQCR's purpose is to challenge the engagement partner and team. Participants describe the EQCR's position as being independent and that there is no threat of the EQCR becoming familiar with the client.

Audit partners didn't see the usefulness of long cooling—off periods for EQCR's: "You can't flip from being an engagement partner to an EQCR and I think that's a shame"

(Interviewee 9, NonBig4). "Intuitively to me it feels like I could probably do a great job of being an EQCR, so I'm not sure why I would need two more years off" (Interviewee 5, Big4) "I think you need two years minimum, you could have three for the partner and two for the EQCR" (Interviewee 11, NonBig4). Interviewee 13, Big4 explains that some of the supply issues with audit partners comes from their capacity to be an EQCR on client engagements which limits their ability to be the engagement partner. Participants viewed the cooling—off periods for EQCR's critically which potentially comes from regulators not providing a justification for why EQCR's need to cool—off. The complexities of the rotation requirements for EQCR's is likely to be contributing to the auditor shortage in New Zealand.

## 4.3.5 Audit partners with client's specific knowledge increase audit quality

Researchers that argue against audit partner rotation do so on the basis that by rotating the audit partner you lose client specific knowledge (Myers et al. (2003): Several interview participants provided the perspective that industry experience improved the judgements that audit partners made an improved audit quality through knowledge of the client operations (Interviewee 7, Big4). The argument as stated by Interviewee 6, NonBig4: "You might go, there so specialized maybe he's become complacent" then expanded on "the risk is, you get complacent and not challenge things enough" (Interviewee 11, NonBig4). New Zealand audit

partners agree that industry knowledge does improve an engagement partners decision making, but also perceive a risk in longer engagements of becoming complacent.

Interviewee 11, NonBig4 provides the perspective that: "The biggest risk is that you lose a lot of institutional knowledge when you rotate the engagement team" and that "a lot of institutional knowledge is kept within the audit firm". That there is a lot of knowledge within the audit team is confirmed by: "You retain all of that collateral knowledge within your manager and audit seniors within the firm" (Interviewee 4, Big4) "we make sure there is continuity of the audit manager or senior staff on the audit" (Interviewee 1, NonBig4) and that rotation "puts more responsibility on the manager.....it's really because of that base knowledge, that understanding is often with the manager" (Interviewee 4, Big4). What can be identified from these comments is that changing the engagement partner is less of a risk than changing the audit team, though: "You do have some risks when you rotate partners" (Interviewee 11, NonBig4) "because you've been forced to rotate to someone that has to learn the whole industry" (Interviewee 1, NonBig4).

The size of New Zealand's audit market is a factor that influences industry specialization: "New Zealand [SIC] too small to have specialists in industries" (Interviewee 6, NonBig4). The identifiable problem with rotation is: "Because we're so small you can lose specialization" (Interviewee 6, NonBig4) "you've lost your industry specialisation because you've been forced to rotate" (Interviewee 1, NonBig4) "were losing industry specialization, which is critical to audit and that's really, really important....very specialized businesses may be a challenge for audit partner rotation because is that new partner really adding value or reducing value to the audit function" (Interviewee 6, NonBig4) because "you can't just go and pull in a random partner and say you're now a (industry 1) specialist, when they've never done that" (Interviewee 12, NonBig4). New Zealand audit partners were in agreement that engagement partners with industry specialist knowledge outperform engagement partners

without industry experience confirming research by (Knechel et al. 2013: Messier et al. 2008).

## 4.3.6 The audit team drives audit quality in an engagement

Carcello et al. 1992 categorised audit team and audit firm experience, client knowledge and industry expertise as one of the four most important factors for determining audit quality. New Zealand audit partners provided their perspectives on how the audit team impacts audit quality: "The team of people together is what gets that audit to the result" (Interviewee 8, Big4). "If you've got a good director a good team.....combined with a good [sic] client management, you are a long way on the road to a good quality audit" (Interviewee 12, Big4). Because: "The practical reality is the works being done at a manager and team level" (Interviewee 1, NonBig4). Then linking the audit team to audit partner rotation: "That partner rotation period is quite a good time to refresh the team to make sure that we're still being current" (Interviewee 2, Big4). "By rotating the partner, you've still got a lot of the previous knowledge retained within other team members, which is really *important*" (Interviewee 4, Big4). The quality of the people within the team is important: "That meaningful work ethic is huge...... all our people should be high quality delivering the best" (Interviewee 13, Big4). "The way we frame it with our clients, is we expect them to be looking at the quality of service, the audit quality, the challenge that we provide as an audit team to your management team" (Interviewee 2, Big4) "the incremental difference, is having really good quality people and good quality teams" (Interviewee 1, NonBig4) because "it's a joint thing between a whole lot of people" (Interviewee 8, Big4) "they're working out, what's the opinion? have we complied" (Interviewee 2, Big4). Interview participants were in agreement that the audit team is one the biggest determinant of audit quality and that the fundamental knowledge held within the audit team is really important for the audit firms to maintain audit quality.

### 4.3.7 The more complex the business, the more time on an audit partner needs

Audit firms and engagements partners need to control for entity size and complexity in an audit. Larger more complex entities can have increased numbers of transactions, divisions and multiple subsidiaries adding layers of complexity to the audit engagement and increasing the likelihood that the financial statements could be misstated, as the engagement partner becomes more remote from the transactional data being produced. Audit partners discuss the impact of complexity on the audit: "It's always been a debate, what's the appropriate time for rotation, and I don't know what the answer to that is, it really depends on the complexity" (Interviewee 4, Big4). "I think if you look at an audit, it's an incredibly complex thing...... on a big job that is very complex, it takes you a lot longer, to make sure you do it well" (Interviewee 8, Big4). "FMC reporting entities are not simple, they've usually got some areas of complexity in them, and so it does often take you some years to really get a thorough understanding of how they work and how issues have been responded to" (Interviewee 10, NonBig4)". Participants agreed that size and complexity are significant factors for determining when engagement partners should be rotated, though critically, new audit partners make the observations: "When you think about why it's there, it's the familiarity risk, so that doesn't change depending on complexity" (Interviewee 13, Big4) "I think it depends on the complexity of the client, how many partners are involved" (Interviewee 4, Big4).

Audit partners questioned cooling-off periods: "It's probably on the cool off stand down part of it which is a bit onerous in the New Zealand setting" (Interviewee 1, NonBig4, 6NonBig4) "I don't think you need to be off for five years.....personally I think if you have no involvement at all, two years is fine" (Interviewee 8, Big4) and that the cooling-off periods present the biggest challenge to the audit firms: "I think probably the cool down periods probably are the biggest challenge for us in the local market given our size, five years off is a

long time" (Interviewee 12, Big4). Critically raising the question: "Why five years versus two years.... does that five years actually benefit anybody" (Interviewee 13, Big4). Participants also provided perspectives on cooling-off periods for the EQCR: "The EQR is three, I'm not really sure why the EQR is three and not two years like the other key audit partners, because I think two years gives you an appropriate time to refresh, but also gives you the advantage of coming back with some knowledge" (Interviewee 7, Big4).

Participants in this study were critical of cooling-off periods for engagement partners and EQCR's suggesting that they were too long and could be reduced to two-years for both the EQCR and the engagement partner. Participants also communicated that New Zealand regulators were determining the length of cooling-off periods to align our regulations with international standards. The implications of the managing partners comments (Interviewee 6, NonBig4) are that increased rotation policies and extended cooling-off periods are levers for increased market concentration, which implies that reducing these levers will decrease market concentration. Evidence from prior research found marginal improvements in audit quality from extending the cooling-off period.

## 4.4 Relationships between engagement partners and their clients

That last sections of this chapter discovered that audit firms match personalities between the engagement partners and clients. Participants in this research provided qualitative research evidence that auditor client relationships include gaining information to complete the audit and gain an understanding of the entity, technical abilities or industry experience to maintain audit quality, and an engagement partner must be able to constrain management taking an aggressive financial position. Audit partner rotation periodically ends these relationships, which potentially effects audit quality in the first years of the engagement (Carey and Simnett, 2006).

This section of the chapter will be set out as follows: Firstly, the early years of an engagement partners tenure. Secondly, the incoming audit partners 'fresh perspective'.

Thirdly, audit partner rotation enhances the new partners ability to challenge management.

Fourthly, audit partner rotation reduces the 'familiarity threat'. Fifth, management perceive audit partner rotation as an inconvenience. Sixth, boards of directors' view audit partner rotation as good governance. Seventh, organisational employees are constantly changes.

# 4.4.1 The first years of an audit partner's engagement

Researchers argue that in the first few years of an audit partner's engagement the financial statements are more likely to be misstated while an audit partner builds up knowledge and experience. Audit partners confirm that the engagement partner in the first few years on an engagement is on a learning curve: "You've got your first year when you're learning the client" (Interviewee 2, Big4) "in your first year you might ask a lot of stupid questions because you're on a learning curve" (Interviewee 4, Big4) "you have your first year of getting up to speed, building relationships" (Interviewee 7, Big4). "What I find is that in year one on an audit you learn all the big picture stuff' (Interviewee 9, NonBig4) "the most dangerous year...... it's always a big learning experience" (Interviewee 10, NonBig4). "You're only really understanding and hitting your straps by year two" (Interviewee 12, Big4) the "first year is always a bit lumpy" (Interviewee 3, NonBig4). Audit managers provided the perspective that, an engagement partner will not be as familiar with an organisations affordable risk in the first year of an engagement which reduces audit quality (Interviewee 5, Big4) "firstly, you challenge them, and you get some response, and you accept it" (Interviewee 2, Big4). These statements from audit partners confirm the research perspective that the financial statements are more likely to be misstated in the first year of an audit partners engagement while they build up knowledge and experience.

Interviewee 5 Big4 suggested that: "Your second year is when you think your starting to get on top of things" "I do an audit, in the second year I pick up on things I didn't pick up in the first year" (Interviewee 6, NonBig4) "year two you're filling in gaps....you're finding out stuff which you didn't realize you got wrong in year one" (Interviewee 9, NonBig4) "I don't know how you can get a grasp of a company within a year or two, and then you get a year or two of actual quality" (Interviewee 4, Big4). To manage this, managing partners encourage in-depth discussion between the exiting partner and incoming partner before they pick up the audit, so that they are aware of what to challenge management on (Interviewee 5, Big4). Interviewee 2, Big4 states how in the second year you confront management and tell them that their key assumptions aren't working. What's important is that: "By rotating the partner, you've still got a lot of the previous knowledge retained with another team members, which is really important" (Interviewee 4, Big4) "it's helpful having a manager that's got a relationship with them" (Interviewee 11, NonBig4) "there is continuity of the audit manager and senior staff on the audit, so the actual audit work is being done by people who have some continuity and experience" (Interviewee 1, NonBig4). Collectively audit partners perceive that: "you typically tend to find as you work through your second year, or maybe even your third, that you learn more and more" (Interviewee 10NonBig4). What can be concluded from this is that engagement partners rely heavily on their audit managers in the first year and look to rectify any mistakes that may have been made in the second year of an engagement.

Interestingly Interviewee 5, Big4 describes how: "One year in five, the audit is not as good as the other four potentially, then you could easily say, with a five-year rotation, that in the last year how interested is the partner going to be? Are they going to be focused on their next rotation, or are they going to be focused on their last audit? So, if you took a really negative view of it, you'd argue that two of the five years, forty percent are not at the optimal level" Interviewee 5, Big4's critical analyses of audit partner rotation relates directly to

research evidence that states that audit quality is not as high in the first year of an engagement, and confirms Winn (2021) research that provides evidence of audit partners not being focused on an audit in the year before rotation.

### 4.4.2 The engagement partners 'fresh perspective' can lead to increased audit quality

Regulators believe that the incoming audit partners 'fresh perspective' or 'fresh set of eyes' will improve audit quality, which was confirmed by interview data of this research: "The new pair of eyes, the new experience, the different expectations the slightly different approach, that's how you hope to see some quality, potentially some quality improvements" (Interviewee 1, NonBig4). "There is no doubt there is some benefit [SIC] of having fresh eyes in there" (Interviewee 8, Big4). "My own personal experience has always been positive around having a fresh perspective on things" (Interviewee 13, Big4). Audit partners are in agreement that audit partner rotation brings a 'fresh perspective' to the engagement which can lead to positive outcomes, affirming the regulatory regimes position.

Participants explain the 'fresh perspective' starts with: "I've unapologetically said to the management of the entity, I'm new in here and I'm going to ask some questions on things" (Interviewee 8, Big4) "there's always more questions asked and different aspects that come out of the audit" (Interviewee 11, NonBig4). Leading to: "It does create the opportunity to have a fresh look at some of those key judgments" (Interviewee 12, Big4). "Potentially you could get a new partner or new firm coming in saying, actually, I don't think that treatment was correct, or they're asking questions to try and understand it" (Interviewee 4, Big4) "someone coming in, doing it fresh might think well I had a client that had this sort of transaction a couple of years ago, and they counter for it this way, is this client doing it right or not" (Interviewee 3, NonBig4). Interviewee 10, NonBig4 explains: "You look at the balance and think that's fine, we already dealt with that last year......you don't necessarily question it again, whereas bringing another partner on after a period of time means they will

look at it with a fresh set of eyes, and they will question it again......you get to challenge things that have been done in the past". Which: "Takes people out of their comfort zones, I think that has definitely worked, if that was the objective of it" (Interviewee 7, Big4). This research confirms prior research done by Daugherty et al. (2012) that audit partner rotation is a re–examination of the key risks and Lennox et al. (2014) that engagement partners peer review, each other's work. Then adds to this body of knowledge by providing details of how engagement partners critically re–examining treatments, judgments and balances through posing questions to the client, which makes audit partners rotation a process for the continual re-examinations of risks that may threaten the entity.

A key construct of the 'fresh perspective' is that it enhances the challenge to management: "Five years of challenging a client? I think it's appropriate that you have someone new come in to continue the challenge or to continue the focus. So, the new partner comes in, and doesn't know the track record, the discussions, the detail" (Interviewee 2, Big4) "you're definitely asking all those questions about, why do we do it this way, I understand what the risks are, so you definitely are challenging" (Interviewee 12, Big4). Interviewee 9, NonBig4 summarises how the engagement partner applies the 'fresh perspective': "It's back to that whole fresh pair of eyes, risk assessment, audit focus, then is there a proper review of controls....... a different firm may have a different approach, so it's very complex". Audit partners agreed that the 'fresh perspective' and audit partners rotation increases the ability of the engagement partner to bring a fresh challenge to the management of an entity (Interviewee 6, NonBig4).

### 4.4.3 Audit partner rotation enhances an engagement partners ability to challenge

Beattie et al. (2004) provides research evidence that auditor—client negotiations often lead to adjustments being made in the financial statements. What can be concluded from this is that the auditor has challenged management on a specific treatment, judgment, or balance,

which has resulted in an adjustment being made. Audit partners explain that: "Your job is to challenge people, so you are naturally going to be in debates, some debates you're going to win and some you're going to lose that's the job of the audit partner is to change positions" (Interviewee 6, NonBig4) "I guess it's how you assess and challenge the key assumptions" (Interviewee 7, Big4). The point is to: "If we find an issue, I explain to the client why we've raised the issue, you say this is what the rules say, this is what we're doing, this is the problem that we have" (Interviewee 10, NonBig4). Then linked to audit partner rotation: "I guess the point of audit [SIC] rotation, as an auditor our role is really to hold the management of the company to account and challenge a lot of the assumptions that they've made in the financial statements" (Interviewee 4, Big4) and "audit quality comes down to being courageous, having courageous conversations with clients, and challenging them, and being able to demonstrate and to continue to challenge them in difficult circumstances" (Interviewee 2, Big4). This research confirms that auditor negotiations can lead to the adjustments being made in the financial statements through the audit partner challenging management on the key assumptions that have been made.

A critical part of audit partner rotation is that the incoming engagement partner: "You're not just blindly coming on going, because partner 4 did it that way, it must be the right way, going back to individual partner risk" (Interviewee 12, Big4) and bringing "it is also the element of new different experiences and challenges, and dealing with different clients and different audit teams as well through that process" (Interviewee 13, Big4). The incoming engagement partner needs to ask the question: "Why do we apply this materiality benchmark? Why are these the risks" (Interviewee 12, Big4). "Have we really addressed the right risks? Have we set the right materiality?" (Interviewee 13, Big4) with the purpose being to: "Challenge our own previous judgments, not necessarily to result in a significant change, but to just revaluate and confirm what those judgement were" (Interviewee 12,

Big4). Interviewee 7, Big4 explains how the peer review process can lead to internal challenges with the audit firms, and that it can takes confidence to come to another view that the previous engagement partner. Because "you're always going to have judgment calls when producing financial statements" (Interviewee 6, NonBig4). Audit partner rotation initiates: "Getting to challenge things that have been done in the past and see if you agree with them or not" (Interviewee 10, NonBig4) and "you're going to go straight in and challenge, because you kind of know that's what your role is" (Interviewee 6, NonBig4). Engagement partners in negotiations have to: "Make judgments as they go, a lot of that is about looking someone in the whites of their eyes and just challenging them, asking them hard probing questions" (Interviewee 4, Big4) and "clients should be thinking of tendering or rotating when they don't believe they can be sufficiently challenged and having robust conversations with their auditors, that's ultimately what it is...... providing you don't lose that knowledge" (Interviewee 2, Big4). This research provides evidence that audit partner rotation is a critical reassessment of the material risks that the entity faces and that audit partners rotation initiates engagements partners challenging each other's judgements resulting in sharing of knowledge and increased audit quality.

### 4.4.4 Audit partner rotation reduces the familiarity threat

Mautz and Sharaf (1961) were the first to suggest that a corrosive element could emerge in longer tenures. Regulatory agencies would develop the construct of a 'familiarity threat' based on the perception that engagement partner in longer tenures is less likely to challenge management on aggressive financial positions. Audit partners provide the perspective that: "I think there is a familiarity threat that comes along if you're just on the job for years and years" (Interviewee 1, NonBig4). "If I was being honest, there were a few clients I was getting a little familiar with" (Interviewee 6, NonBig4). "Familiarity over ten years we say you can't do it. So, we've introduced the principles based on familiarity and

independence" (Interviewee 5, Big4). "When you think about why it's there, it's the familiarity risk" (Interviewee 13, Big4). "I think the intention of rotation is to ensure quality through removing familiarity and that's a really important thing" (Interviewee 6, NonBig4). Audit partners were in agreement that familiarity can be a corrosive element in an engagement, supporting the position taken by regulators in implementing audit partner rotation. Interviewee 13, Big4 states that: "The objective of partner rotation is to meet the ethical requirements to reduce the familiarity threat".

Participants also associated overfamiliarity with potentially missing something in the financial transactions: "You do find that you get familiar with your clients, so whether that means you miss things I don't know" (Interviewee 3NonBig4). "The partner overlooked that because he was perhaps too familiar" (Interviewee 8Big4). Perceiving that familiarity may inhibit an engagement partner's ability to challenge management: "The risk is that you aren't asking those challenging questions" (Interviewee 4Big4). However, familiarity can also be positive: "You do get quite familiar with them which actually gives you an insight into audit risk and some of the fraud risk" (Interviewee 1NonBig4). Interviewee 1, NonBig4 explains that as you accumulate audit knowledge of a client, the history behind transactions, you have increased knowledge of the risks that threaten the entity "There's always also a risk.... that possibly they're not as familiar, and therefore they may miss something" (Interviewee 11NonBig4). Participants argued the critical view that familiarity lessens the likelihood that engagement partners will challenge management on aggressive financial positions supporting research by (Chia-ah & Karlson, 2010) but they also support the view that there is a risk to audit quality in the first few years of an engagement that an audit partner may miss things from not being as familiar (Chi & Huang, 2005: Chen et al. 2008: Jenkins & Velury, 2008: Knechel & Vanstraelen, 2007: Myers et al, 2003).

### 4.4.5 Clients perceive audit partner rotation as an inconvenience

Proponents of audit partner rotation make the argument that by changing audit partners the incoming audit partner will be critical of managements taking opportunistic accounting decisions. The participants in this research perceived that audit partner rotation can frustrate management: "They have to explain the same things all over again, which really frustrates clients" (Interviewee 4, Big4) "some clients see it as just another inconvenience, because doing an audit it is an inconvenience, and we go in there and when you have new people, they do ask silly questions" (Interviewee 8, Big4) "I think in general though there is quite a bit of resistance" (Interviewee 9, NonBig4) "I think from a management perspective, it's probably seen as more of a negative because there obviously used to dealing with the same people, and they have that relationship" (Interviewee 11, NonBig4). Interviewee 6, NonBig4 explains why: "I finally teach you about my business and you can't do my audit anymore and now I've got to get someone else in that doesn't know my business and I've got to teach them" "What we see is that even if a company or management is getting bad audit service, there's a perception that there's too much work, they'll create more work for themselves by changing audit firms" (Interviewee 9, NonBig4). Prior research focuses on a perception that in longer engagements, audit partners may become sympathetic to management opportunistic decisions, this study finds no evidence to support that. But does provide evidence that the reason management are resistant to changing engagement partners is because it represents increased work volumes from having to teach auditors that don't understand their business. However, these perspectives are potentially influenced by the audit partner lacking knowledge about the client in the early years of the engagement.

Several participants perceived that consistency was important to the management of the entity: "From a client perspective, they like having consistency" (Interviewee 2, Big4). "As the audit partner rotates, hopefully the team below remains reasonably consistent"

(Interviewee 5, Big4), which can be considered a quality issue from the audit firms' perspective. "We always try to have the consistency of either having the engagement partner, quality reviewer or the audit manager....so you've always got some consistency of the skills" (Interviewee 1, NonBig4) because "those are the people that they want to see every day, and they're the ones that understand ...... how it works" (Interviewee 3, NonBig4). These participant perspectives confirm the audit quality aspect of having an audit partner and team that have client specific knowledge.

### 4.4.6 Boards of directors' view audit partner rotation as good governance

Dependent on the type of entity, either the audit committee or board of directors are responsible for appointing the audit firm. Part of this responsibility is to ensure that the auditors are independent and remain objective in the engagement. Participants suggest that directors viewed audit partner rotation as good governance: "A lot of them like to apply best practice" (Interviewee 4, Big4) "So they didn't need to rotate, but they wanted to" (Interviewee 10, NonBig4) "it's seen as a good thing to do, its good hygiene, it's a positive" (Interviewee 11, NonBig4) "I think from the directors perspective on rotation it is very good for them, and it helps them from the governance perspective" (Interviewee 13, Big4). Because it brings a 'fresh perspective': "It was the board that said we would like a fresh set of eyes" (Interviewee 10, NonBig4). "I asked questions which perhaps hadn't been focused on before...... at the board level that was refreshing for them, it was a different discussion of different points" (Interviewee 9, NonBig4). Interviewee 13, Big4 provided the perspective that directors were pushing for quality by ensuring that management provided paperwork regarding judgments and taking responsibility for the independence of auditors, because it's important that "you've got an audit partner that will be able to challenge your management, and work through issues with them" (Interviewee 13, Big4). Participants in this research

agreed that boards of directors' view audit partner rotation as providing assurance that they are gaining a 'fresh perspective' on the key judgment that have been made.

## 4.4.7 Organisational employees are constantly changing

According to the US Bureau of Statistics the average length of time an employee stays within an organisation is five—years (Sturt, 2015). What can be deduced from this, is that the audit partner will often remain with an entity for a similar amount of time as the employee, making audit partner rotation policies based on close client relationships superfluous. Audit partners perceive that: "Most of these types of entities have had the same level of turnover that we would have had. Some of them would have twenty-five to thirty percent turnover in their finance team in the last two years, so it's a real challenge for everyone" (Interviewee 2, Big4). "I think probably from the people that we're working on, when I was there, they will be gone anyway, they will have completely changed their staff, so it'll be like a new relationship anyway" (Interviewee 3, NonBig4) Interviewee 6, NonBig4 stated that on an entity he was auditing, the CFO had change and had a completely different board of directors, but would have still been to familiar because of the audit partner rotation regulations. "In this case the clients had dramatic change within that period" (Interviewee 9, NonBig4). "I think the other thing that we've observed is that there's a lot of change at clients as well, you would very rarely have the same CEO, CFO for the whole rotation period, anyway" (Interviewee 12, Big4) because "where in a generation, where people staying in a job for more than five years doesn't happen much these days because the younger generation want variety" (Interviewee 6NonBig4). Which leads to: "In New Zealand there's so much changes happening at the organizations that we audit, you've already got a natural control there around the familiarity risk" (Interviewee 13, Big4) reducing consistency "it's not like you've got the same management team, same board, same audit team, it's very unusual to have that consistency through, even a five-year period" (Interviewee 12, Big4). Bringing into question the coolingoff period: "Does that five years actually benefit anybody" (Interviewee 13, Big4). Audit partner rotation regulations are implemented to reduce familiarity between the engagement partner and client. Since or during their inception, a generational change has occurred resulting in employees spending less time with an organisation, which brings into question the relevance of rotation.

## 4.5 Chapter summary

This chapter presented the qualitative findings on the three central topic. What was discussed by participant in implications of audit partner rotation on audit practice in New Zealand was that audit partner rotation requires and audit firm to have increased audit partner resources to manage imposed regulations. Planning and coordination mechanisms identified how audit could potentially manage audit partner rotation. Relationships between engagement partners and their clients provided qualitative research evidence on the affects of audit partner rotation in the early years on an engagement, and how the fresh perspective can improve audit quality.

The next chapter will discuss the key findings of this study and relate literature to the findings that have been made in this chapter, while identifying gaps for future research.

# **Chapter 5: Discussions**

#### 5.1 Introduction

The chapter discusses the empirical findings from the interviews with New Zealand audit partners. The method used in this research was thematic analysis. Following this method semantic analysis was used in the findings to interpret the significance of patterns into broader meanings and implications (Braun and Clarke, 2006). This chapter will move to the latent level to conceptualize the underlying ideas and ideologies, presenting the data three dimensionally to identify the features that gave the data particular form and meaning.

The discussions section of the chapter will be separated into three sections: Firstly, the implication of audit partner rotation on audit partner rotation. Secondly, planning and coordination mechanism. Thirdly, engagement partner client relationships.

### 5.2 Implications of audit partner rotation on audit practice in New Zealand

The findings section of this researched discovered that audit firms were already using their own internal controls to manage the 'familiarity threat'. By making audit partner rotation mandatory regulators have increased market concentration and centralised audit services to larger audit firms with increased partner resources, the implications of this have been that small audit firms have either been forced to merge or leave the domestically registered audit market.

This section of the chapter will discuss: Firstly, that New Zealand audit firms were already rotating their audit partners. Secondly, how audit partner rotation has increased market concentration. Thirdly, the practicalities of audit partner rotation regulations in the New Zealand. Fourthly, audit partner rotation and non-Big Four audit firms. Fifth, audit partner rotation in large client entities will be described. Sixth, audit partner rotation and Big Four audit firms, the last section will present the conclusions.

# 5.2.1 Audit firms in New Zealand were already rotating their audit partners before audit partner rotation became mandatory

Before audit partner rotation was made mandatory in New Zealand audit firms were meant to use their own internal controls to mitigate a long association risk between the engagement partner and client. The New Zealand audit industry has been described as being largely unregulated until the FMA was establishment in 2011 (Rainsbury, 2019). The implementation of a regulatory regime and a three-dimensional system for audit partner rotation can be considered a regulatory intervention into the New Zealand audit market. Audit partners confirmed that audit firms in New Zealand were already rotating their audit partners generally between seven to ten years and that by making audit partner rotation mandatory was a solidification of the rules rather than the implementation of new regulations (Interviewee 1, NonBig4, 2, Big4, 9, NonBig4, 11, NonBig4). Audit partners perceived that by implementing audit partner rotation, the regulatory regime in New Zealand was catching up to the values of leading audit firms that identify long tenures as a threat to independence and over familiarity in the engagement as a problem (Interviewee 6, NonBig4).

Audit firms are naturally geographically disbursed to facilitate the needs of remote industries. New Zealand has a lot of small towns often centred around one large industry. The problem for audit firms in small towns is that they have only one or two audit partners servicing one large client. The perception of audit partners from larger audit firms is that these small audit firms lack the knowledge or the resources to apply International Financial Reporting Standards (IFRS) to an acceptable standard (Interviewee 6, NonBig4, 14, NonBig4). The problem from a regulatory perspective is that small audit firms are financially incentivised to keep major clients. Audit partners perceived that some audit partners were operating clients on a continual basis due to a lack of enforceable regulations (Interviewee 2, Big4, 9, NonBig4) negatively impacting the audit industry.

# 5.2.2 Mandatory audit partner rotation has resulted in increased market concentration in New Zealand

Rainsbury (2019) found that the introduction of a regulatory regime disproportionately affects small audit firms. This research agrees with Rainsbury (2019) but found that the predominant reason for the reduction in domestically registered audit firms was mandatory audit partner rotation. The imposition of audit partner rotation regulations significantly affected small and mid-tier audit firms in New Zealand based on a simple dynamic. Mandatory audit partner rotation regulations require small and mid-tier audit firms to have increase audit partner and EQCR resources to service their clients through time-on periods and extended cooling-off periods. This has significant implications for audit firms with limited partner resources. Participants from a number of different firms perceived that audit firms requires a minimum of four licensed audit partners to manage audit partner rotation with additional partner resources being required from other geographic locations. Audit firms that foresaw the effects of audit partner rotation merged or made arrangements with other firms to contract audit partners in, otherwise the audit firm had to opt out of auditing NZX, FMC, and OAG entities.

Four participants confirmed that audit firms with less than four audit partners were contracting in either engagement partners or EQCR's from other firms (Interviewee 1, NonBig4, 6, NonBig4, 9, NonBig4, 11, NonBig4). What these participants stated was that small audit firms were entering into mutual arrangements with other small audit firms for the supply of audit partners and EQCR's. Audit firms swapping and changing audit partners appears to be an unintended consequence of audit partner rotation.

# 5.2.3 The practical implications of audit partner rotation regulations in the New Zealand environment

Audit partners perceived that rotation regulations have caused some audit firms to leave the registered audit firm market and caused other audit firms to contract in licensed audit partners. This brings into question the practicality audit partner rotation settings in New Zealand. Audit partners from non-Big four audit firms were critical of audit partner rotation regulations based on the size and scale of commercial entities and audit firms. Interviewee 10, NonBig4 provided the perspective: "We are a very small country, with a very small number of auditors, whilst partner rotation might work great in a US firm which has one hundred and fifty audit partners. It doesn't necessarily work well in smaller firms in New Zealand". Confirming the problem identified by the researcher and reiterating that to manage audit partner rotation effectively is dependent on the number of audit partners in the audit firm.

Audit partners in this research perceived that New Zealand was now a standard taker rather than a standard maker. It is very quick to adopt regulations from larger capital markets from around the world (Interviewee 1, NonBig4, 5, Big4, 6, NonBig4, 9, NonBig4). By doing so New Zealand is aligning the quality of auditing standards to international standards, the controversial topic then becomes whether audit firms in New Zealand have the partner resources to implement these standards without adversely affecting audit quality. Adopting a regulatory intervention from overseas may lead to a number of benefits to New Zealand businesses such as attracting foreign investment and conversely making it easier for New Zealand companies to operate in other countries. However, some participants perceived that this approach from regulators was not consistent with the size of New Zealand's audit firms and commercial entities, because they were designed for large scale international markets and corporations making their application difficult to fit.

### 5.2.4 Audit partner rotation and non-Big Four audit firms

The participants in this study were made up of eight audit partners from non-Big Four audit firms and seven audit partners from Big Four audit firms. All the audit partners in this study came from audit firms that had at least four audit partners. This limited the perspectives to audit partners from audit firms that could effectively manage audit partner rotation though two partners involved were from companies that had been forced to merge due to audit partner rotation. It was clear that the current rules around rotation were both challenging to both of these identifiable categories. So far, the discussion has been focused on the most affected group of small audit firms. Naturally the discussion now moves to non-Big Four audit firms with at least four audit partners.

The table below presents the how audit partner resources are distributed amongst New Zealand audit firms. Big Four audit firms have between sixteen to forty-six licensed audit partners including audit partners that are located in Australia. Non-Big Four participant in this study came from audit firms from between four to thirteen licensed audit partners. The overarching difference between non-Big Four audit firms and Big Four audit firms in dealing with audit partner rotation was that Big Four audit firms have sufficient partners to resources to supply audit partners to all their clients, thus Big Four audit firms focus on matching the engagement partners to their clients based on industry experience and the specific roles within the engagement. Whilst non-Big Four audit firms focus on consistency of experience through the engagement team ensuring that either the engagement partner, EQCR, or audit manager has the appropriate industry experience.

Table 5: New Zealand domestically registered audit firms and audit partner distribution

Audit Firm	NZ audit partners	Australian audit partners	Total
PWC	32	14	46
Deloitte	22	7	26
Ernst & Young	14	2	16
KPMG	24	0	24
BDO	10	3	13
Grant Thornton	5	0	5
Baker Tilly Staples Rodway	10	0	10
William Buck Audit (NZ) Limited	4	0	4
RSM Hayes Audit	5	0	5
Silks Audit Chartered Accountants Limited	4	0	4
Crowe NZ Audit Partnership	4	0	4
KS Black & Co	0	2	2
PKF Goldsmith Fox Audit Limited	2	0	2
LNP Audit and Assurance PTY LTD	2	0	2
MNSA	1	0	1
Crowe Horewarth	2	0	2
Triple Ledger Limited	1	0	1
IRCS	1	0	1
AD Danieli Audit Pty Limited	0	1	1
Totals	143	29	172

Note 1: PWC has one licensed audit partner in Singapore, this person was counted into audit partners from Australia.

Note 2: The data for this table was taken from the Licensed Auditors Register at the New Zealand's Companies Office on the 1<sup>st</sup> of August 2022.

EQCR's and engagement partners rotating at the same was considered a problem by participants which was initially caused by the way the regulations were brought in. The predicament that audit firms faced, was that within a relatively short period of time both the

engagement partner and EQCR would be ineligible to be on the engagement. Bringing the problem back to audit partner resources in New Zealand and that implementing audit partner rotation policies forces an audit firms to change audit partners and EQCR's. The consequence of this to client audits was the potential loss of institutional knowledge for a particular client and or industry. Audit partners from both Big Four and non-Big Four audit firms have included both the engagement partner and EQCR into the rotation matrix.

Research identifies that audit partner rotation is a problem for specialist audit partners in specialist industries, requiring partners to specialise in multiple industries (Hammersley, 2006). New Zealand like most countries have several niche industries in different geographical locations (Pells, 2022). The consequence for non-Big Four audit firms was that the imposition of audit partner rotation regulations may have resulted in audit firms rotating audit partners with specialist skills off their specialist client industries and replacing them with an engagement partners that does not have such knowledge, thus potentially lowering audit quality (Interviewee 1, NonBig4, 6, NonBig4).

### 5.2.5 Large client entities and audit partner rotation

To understand the significance of transposing audit partner rotation regulations from large-scale economies to a small-scale economy the researcher investigated large client entities. Client entity size in the US, UK, Canada, and Australia can be considered to be a lot larger than in New Zealand as are their audit firms. Large client entities generally contain different departments and multiple business units (Hammer, 2001: Kaplan and Norton, 2005). The designation of a other key audit partner is that they audit a subsidiary or division within an entity IESBA (2007). Big Four audit partners discussed auditing large complex clients both in New Zealand and overseas, providing evidence that large entity audits consist of between many more audit partners than the engagement partner and EQCR.

Dodgson et al. (2020) states that audit standards in the US merely state that the maximum number of years an engagement partner is allowed to stay on an engagement, followed by the length of the colling-off period. In the New Zealand context, rotation policies set out by the XRB state that engagement partner following rotation should have no contact with the client following rotation (XRB, 2022). This leaves the door open in larger client audits for the second audit partner or an audit partner from a division to become the signing partner when the engagement partner is rotated. Interviewee 5, Big4 explains that in large client audits they stagger the rotation of audit partners to conserve institutional knowledge within the engagement and that the second partner or partner from a division that has already been involved in the audit may move into the engagement partners role. Big Four audit firms are effectively managing audit partner rotation to retain institutional knowledge in the engagement. The controversial issue is what was this the intention of the regulators in implementing audit partner rotation policies. The regulatory position is that the incoming partner will bring a 'fresh perspective' to the engagement. Big Four audit firms in bringing through a partner from either being the second partner or partner in charge of a division may have influenced that partners perspective of the engagement through an induction process (Pittman et al. 2022). This research provides evidence that an engagement partner may have been in the audit for two or three years prior to becoming the engagement partner (Interviewee 5, Big 4). While this gives the audit partner time to learn the entities client specific matters, it also gives the incumbent partner time to influence the partners perspective towards how the judgements of accounting standards have been determined which inhibits the 'fresh perspective' that they may bring to the engagement.

Regulators should assess a countries demographics before implementing regulations.

Regulators in large economies conceive larger client entities and large audit firms. They then design regulations to meet those demographic conditions. In the context of audit partner

rotation, regulators should consider (1) whether the audit firms in a country have sufficient audit partner resources to be able to service their clients whilst allowing for time-on periods and cooling-off periods; and (2) whether the regulations could affect audit quality. By considering these demographic variables, the regulators would be 'right sizing' the regulations to fit the environment that they are being applied to (Quack and Schubler, 2015). That small audit firms have left the domestically registered audit market due to audit partner resource limitations brings into question whether the New Zealand regulators have 'right sized' the regulations to meet the demographics of the economy. Regulators can argue that matching your trading partners auditing regulations can increase trade between countries. Many countries have adopted audit partner rotation regulations since the corporate collapse of Enron and the passing of the Sarbanes Oxley Act 2002.

# 5.2.6 Audit partner rotation and Big Four audit firms

Audit partner rotation relies on there being a sufficient supply of audit partners with the appropriate level of experience and industry knowledge. From the audit partners perspective, they require an interchangeable supply of clients. Big Four audit partners perceived that there may not be comparable entities for an audit partners to take on after they have been rotated off from a high-profile client. Interviewee 12, Big4 discussed how audit partners may have to replace one large client with ten smaller ones. The effect of this on the audit partner is that they may lose specialized skills (Interviewee 6, NonBig4). Auditing and accounting standards are constantly being implemented and updated. Audit partner rotation in a country like New Zealand can cause audit partners to be forced out of auditing the industry that they are specialized in. The consequence of this is that audit partners may re-enter the industry with deteriorated expertise.

Audit firm and audit partner specialisation increases earnings quality (Arthur et al. 2017: Balsam et al. 2003). The concept of auditor specialisation is that as an audit partner

spends more time in an industry, they gain more industry experience and specialist knowledge. Big Four audit partners perceived that there were very few audit partners in New Zealand with the industry experience to audit certain entities, especially large complex entities. Industry experience gives the engagement partner an advantage in determining the correct treatments of accounting standards (Garcia–Blandon and Argiles, 2017: Carcello and Nagy, 2004). Making audit partner rotation mandatory is an intervention into longer engagement partner client relationships, the consequence of this in New Zealand is that without an interchangeable supply of clients in specialist industries is to erode the gene pool of specialist audit partners to becoming general partitioners (Interviewee 1, NonBig4, Interviewee 6, NonBig4) which may affecting the quality of earnings (Balsam et al. 2003). Managing partners from Big Four audit firms stated how audit partner rotation caused difficulties for the audit firms in filling all of the available roles and allocating the most suitable audit partner to the engagement with the appropriate industry knowledge, providing qualitative research evidence that audit partner rotation may not consistently increase audit quality.

### **5.2.7 Conclusions**

Audit partners perceived that regulators in New Zealand are matching audit partner rotation regulations to larger countries that have large audit firms and increased partner resources. The consistent view amongst both Big Four and non-Big Four audit partners was that regulators are not 'right sizing' regulations to the New Zealand demographic environment and the audit partner resources available to New Zealand audit firms. Audit partners perceived that specialisation was eroding at the audit partner level while uplifting the skills of audit manager and team members.

The implications of audit partner rotation have been that there has been centralisation of audit services to larger audit firms with increased resources. Which has potentially

increased audit quality and improved compliance to IFRS. This research has provided qualitative evidence that there are only few audit partners in New Zealand capable of auditing a large complex entities. The problem that audit partner rotation creates in a small-scale economy is that due to extended cooling-off periods, audit firms are unable to allocate the most suitable audit partner to the engagement. The problem as perceived by some audit partners was that there isn't an interchangeable supply of clients for an engagement partner to rotate onto in New Zealand.

# 5.3 Planning and coordination mechanism

A significant finding from this study is that audit firms use general planning and coordination mechanisms, namely a matrix for audit partner rotation. Participants indicted audit firms use a complex matrix for audit partner rotation to ensure that the incoming audit partner has industry experience; institutional knowledge is retained within the engagement; and that the audit manager and team have relevant knowledge of the entities clients' specific matters.

This section of the chapter will discuss: Firstly, the differences between how Big Four audit firms and non-Big Four audit forms will be discussed. Secondly, how transitioning audit partners is part of the matrix. Thirdly, audit partners portfolios and capacity. Fourthly, The EQCR and cooling-off periods. Fifth, client specific matters and the audit team. Sixth, that cooling-off periods pose the biggest challenge to audit firms, and the last section will present the conclusions on the matrix of audit partner rotation.

### 5.3.1 Audit firm size and the matrix of audit partner rotation

Constructing a matrix for audit partner rotation can be directly related to the rostering of staff in human resource management. Rosters include assigning staff to specific roles based on the position they hold within the entity and skills and experience they have to effectively complete the designated role. Setting up matrix for audit partner rotation involves

a number of complexities. Firstly, a computer model needs to be generated to cover the myriad of different time-on periods and cooling-off periods for engagement partners, EQCR's, and other key audit partners. Then client information should be entered with all of the different roles that are required for each engagement. Secondly, the matrix needs to be able to identify engagement partner with relevant industry experience to replace the position of the engagement partner and EQCR. Prior literature states that engagement partners with industry specific knowledge perform higher quality audits. Evidence from this research confirms that in identifying potential engagement partners, audit firms look to provide an engagement partner with industry experience. Engagement partners and EQCR's should have similar levels of experience and age so that one cannot dominate the other when assessing significant judgements made (Beattie et al. 2004, Interviewee 5, Big4). Thirdly, the audit firms then look to infuse institutional knowledge into the engagement and an awareness of the client specific matters. For mid-size engagements this is done through the audit manager and engagement team, for large client entities the institutional knowledge and client specific knowledge is more likely to be held at the audit partner level.

Audit partners from non-Big Four audit firms provided evidence that to retain knowledge of the client specific matters in the audit, they constructed there matrix so that either the engagement partner, audit manager or EQCR has knowledge of the client specific matters that are unique to the entity. The institutional knowledge can therefore be determined to be within the audit firm and engagement team for mid-size entities. Non-Big Four audit firms are involved in auditing all of the three types of entities, though generally engagement partners from non-Big Four audit firms related audit partner rotation to FMC entities.

Discussions regarding other key audit partners was absent from conversation with non-Big Four audit firms. What was discussed is that non-Big Four audit firms have to source engagement partners from other geographic locations around New Zealand and on occasion

Australia making the construct of a matrix either national or international in some instances and due to limited audit partner resources, needed to be extended to audit managers. This means that the matrix for non-Big Four audit firms have to cover five-year, six-year, and seven-year engagements and cooling-off periods of five-years, three-years, and two-years.

A frustration for Big Four audit partners was that there are only few high-profile clients in New Zealand, relegating them to auditing much smaller entities on the completion of auditing a much larger entity. The matrix for a large entity audit includes an engagement partner, a partner that is second in charge, and then a number of other audit partners dependant on the size and complexity of the entity being audited.

## 5.3.2 Transitioning engagement partners within the matrix of audit partner rotation

A significant difference in the way non-Big Four audit partners were transitioned into the next engagement was determined by the size of the entity being audited. The interviewee evidence from this research suggested that non-Big Four audit partners were rarely involved in engagements that involve multiple audit partners, though it can be considered that non-Big Four audit partners often consult tax, treasury, and governance specialists for advice on specific matters in the engagement. Audit partners from both non-Big Four and Big Four audit firms considered New Zealand audits to be predominantly made up of an engagement partner, EQCR, audit manager, and team. With both the engagement partner and EQCR having to be licensed and observe cooling-off periods, engagements of this size cannot facilitate the grooming of an incoming audit partner into the signing partners role position. The transitional process where there is not multiple partners involved is having conversations with the exiting incumbent partner around client specific matters and the material threats to the entity. Prior to research being done by Dodgson et al. (2020) and Pittman et al. (2022), it was assumed that rotating the engagement partners was a standalone event which appears to be relatively true for most auditor engagements in New Zealand. The effect this has is that the

incoming engagement partner can provide a 'fresh perspective' to the audit engagement without being influenced by the incumbent audit partner on the judgements and treatments that have been made.

Big Four audit firms start the transition process with managing partners identifying several candidates for the upcoming engagement based on industry knowledge and experience. Pittman et al. 's (2022) research found that audit firms groom a partner into the signing role position. This research found some evidence to support this, Interviewee 2, Big4, 4Big4 stated that the audit firm does know who they would prefer the incoming partner to be. Multiple Big Four audit partners confirmed the process set out in Dodgson et al. (2020) research which stated that Big Four audit firms provide clients with a selection of audit partner to choose from with some audit committees wanting to interview all of the potential candidates while others prefer the incoming engagement partner to attend meetings to ensure a working relationship can be achieved. A Managing partner stated that in matching the EQCR to the engagement partner imbalances in age and experience should be considered so that a more experienced engagement partner could not dominant the EQCR is assessing the judgments that they have made.

Based on research done by Dodgson et al. (2020) this study investigated the transition process of audit firms. The findings from this study confirmed a number of the processes that Dodgson et al. (2020) describes. Then added to this body of knowledge by providing evidence that audit firms look to match personalities in the transitions process. It is important that engagement partner and client has a productive working relationship, the ease of this relationship can be of benefit in gathering the information required to complete the audit (Dodgson et al. 2020: Richard, 2006). A problem that the engagement partner faces is that they are required to sceptical and objective of a clients treatments of accounting standards, yet they are also want to retain the client, be able to negotiate fees and add value to the entity

they are auditing. The nature of an audit engagement is that there will inevitably be disagreements and potentially confrontations in auditor negotiations, with the engagement partner having to challenge the entity on there treatments of accounting standards. The outcomes of these negotiated settlements can be determined by the relationship that the engagement partner and the client have (Beattie et al. 2004). Audit firms in assessing personalities do so on the basis that the engagement partner will a good cultural fit with the client and be able to resolve any potential issues, so they don't have concerns in issuing an unqualified audit opinion. The resolution of issues can be determined by how an aggressive financial position the client has taken. It should be assessed that entities taking highly aggressive financial positions is rare from the low rate of audit failures (Francis, 2004). The difficulty for an engagement partner in this position is that they have to constrain an entities aggressive financial position whilst maintaining a relationship with the client, so they are not obstructed in the gathering information in future audits or loose the client from not being able to negotiate a settlement on specific judgements that have been made. This could be described as an extreme circumstance in allocating an audit partner to an engagement. The consequence of limited partner resources is that New Zealand audit firms may not have many alternatives in allocating an engagement partner. Interviewee 8, Big4 described a circumstance where a less appropriate engagement partner was allocated to an engagement, due to audit partner rotation regulations and a lack of audit partner resources, which should be of concern to regulatory agencies.

# 5.3.3 Audit firms consider an audit partners entire portfolio before allocating them to an engagement

Auditing a large complex entity takes a significant amount of planning. The acceptance of a client can be dependent on whether a licensed audit partner has the industry knowledge, experience, and capacity in their portfolio. Human resource development

provides literature on how organisations develop the expertise of their employees to meet the organisations objectives (Torraco and Swanson, 1995). To expand an audit partners portfolio, audit firms assign them to different positions in the engagement to build up their knowledge and experience, by doing this they are able to groom their audit partners into the engagement partners positions in larger more complex entities (Pittman et al. 2022, Interviewee 12, Big4). This grooming makes the audit partner the preferred candidate for the engagement partner position, though this may not eventuate if the client does not perceive them to have the right chemistry for a workable relationship. Managing partners oversee the portfolios of audit partners and liaise with employees that implement the global rotation database. What should be considered is that even in the larger audit firms they are still dealing with a finite group of audit partners and for entities that require specialization that group is even smaller (Interviewee 5, Big4,12, Big4). This study has identified a complex structure for inducting audit partners into specific industries and slowly building up their portfolios so they can eventually take on larger more complex clients.

### 5.3.4 Rotating the EQCR in a small-scale economy creates problems for the audit firm

Audit firms are required to rotate the EQCR on all NZX, FMC, and OAG audits with varying applicable timeframes. EQCR's need to have industry experience so that they can competently challenge the engagement partner and members of the audit team (PCAOB). As stated by Interviewee 5, Big4, 6, NonBig4 the matrix of audit partner rotation should identify any experience and power imbalances and identifying the most suitable person for the position. The key difference between the engagement partner and EQCR's in the matrix are the time-on periods and cooling-off periods that have to be observed. Before 2018, EQCR's were required to observe two-year cooling-off periods, this was then extended to three-years for NZX, FMC, and OAG entities. The purpose of cooling-off periods for engagement partners is to increase independence and reduce familiarity between the engagement partner

and the client. It is also perceived that cooling-off periods bring forth an invigorated 'fresh perspective' on an engagement partners return to the engagement.

The reasoning for EQCR's to observe cooling-off periods is not so clear. The reasoning from a regulator's perspective could be that they don't want the EQCR's switching directly into the engagement partners position after completing their engagement as the EQCR. This perception could be assimilated to the perspective that management are able to influence an engagement partner decisions in long tenures. Implying that the engagement partner may influence the EQCR's perceptions on judgements made and treatments of accounting standards. Thus, inhibiting the outgoing EQCR's ability to bring a 'fresh perspective' to the audit engagement when they became the engagement partner.

The EQCR's role in an engagement is to challenge the engagement team and discuss the key judgments made with the engagement partner, they are not client facing and do not interact with the client. Some audit partners in this research were critical of extended cooling-off periods for EQCR's and questioned the reasoning behind EQCR's having to observe cooling-off periods. Based on the perception that overfamiliarity between the EQCR and client does not exist and that their accumulated knowledge of the client could increase audit quality. Regulators could argue that an engagement partner should not move into the position of being EQCR because a relationship bias may still exist. This research does not advocate for no cooling-off periods for EQCR's but does argue that regulators should provide the reasoning for extended cooling-off periods.

# 5.3.5 Institutional knowledge is kept within the audit team

The purpose of creating a matrix is to ensure that there is consistency of knowledge retained within the client engagement. Audit partners did not provide evidence that the matrix of audit partner rotation included individual team members. But did provide the significant perspective that the changing of the audit team was the biggest threat to audit quality

(Interviewee 1, NonBig4, 11, NonBig4, 12, Big4, 13, Big4). What needs to be differentiated is that in a large client entity audit the knowledge of client specific matters is held by the multiple audit partners involved in the engagement. In smaller entity audits the knowledge of client's specific matters is kept within the audit manager and audit team. Litt et al. (2014) research presents the possibility that changing the engagement partner and EQCR may have little impact on audit quality. This concept is at odds with the regulatory approach of singling out the engagement partner for rotation and requiring engagements partners name stated on the audit report. Engagement partners do take overall responsibility, yet what must be considered is that an audit is a team effort involving many people across different areas of the entity and audit team (Interviewee 8, Big4). This practical reality is that in smaller to mid-size engagement the audit work is being carried out at the audit manager team level (Interviewee 1, NonBig4). Audit partners provided the perspective that the quality of those people and the work they carry out determines the quality of the financial statements, with the engagement partner and EQCR being involved at the entity, commercial risk level.

Audit partners considered that the fundamental knowledge in an audit was kept within the audit team. The subject of audit firm rotation was a constant theme throughout this research with some engagement partners seeing this as inevitable (Interviewee 7, Big4).

Interviewee 4, Big4 perceived the first year of an audit firm auditing an entity as the hardest. The issue of engagement partners signing the audit report and regulators being able to conduct criminal and civil prosecutions against the engagement partners is potentially out of congruence when the construct of an audit is taken into consideration. Carcello (1992) states that the characteristics of the audit team are important to audit quality. This research has found that the practical reality is that the audit manager and team complete the audit work. The quality of the work carried out is dependent on the quality of information supplied by the entity. In larger client entities the engagement partner can be considered to be significantly

removed from the transactional data being obtained from the client. This research identified that it is the audit manager who is the go-to person for the management of the firm and is more likely to have closer relationship with management. Making them an integral part of the matrix for mid and smaller sized engagements. Audit partners perceived that audit partner rotation has resulted in engagement partners being more dependent on the work carried out by the audit manager and team in the first year of the engagement. And more reliant on the audit managers and teams' relationships with client due to the periodic breaking of the engagement partner client relationship. That engagement partner is singled out to take accountability does potentially create additional assurance that the financial statement is less likely to be misstated. The argument put forward by participants in the study is that singling out the engagement partner is at odds with the construct of an audit and the many variables that go into producing audited financial statements and the contractual basis of audit firms providing reasonable assurance.

### 5.3.6 Cooling-off periods are the biggest obstacle for New Zealand audit firms

A significant problem in constructing a matrix for audit partner rotation is that audit firms have to manage a myriad of different rules for engagement partners, EQCR's and other key audit partners. Audit partners in this study were in agreement that cooling-off periods in the New Zealand were too long and that the threat of overfamiliarity did not exist after a cooling-off period of three-years. They also believed that cooling-off periods for EQCR's were to long and could be reduced to two-years to match other key audit partners.

Interviewee 13, Big4 questioned the relevance of extending cooling-off periods from three-years to five and whether these timeframes were suitable for the New Zealand economy with limited partner resources. Interviewee 6, NonBig4 potentially provided the answer to this by stating that the suitability of timeframes should be determine by the availability of audit services. New Zealand is currently experiencing an auditor supply shortage with the

government being forced to pass the Annual Reporting and Audit Time Frames Extensions

Legislation 2021 giving OAG entities a two-month extension on filing there audited financial
statements. From December 2023 OAG entities will also be able to apply for a reduction to
the cooling-off periods for the engagement partner. On acceptance by the regulator,
engagement partners cooling-off periods will be reduced from five-years to three-years

(XRB, 2022). Audit partners stated that certain entity types were struggling to gain audit
services such as charities and not-for profits, indicating a shortage of audit firms. What this
suggests is that reducing cooling-off periods will increase the availability of licensed audit
partners to complete audits. Making increased cooling-off periods levers for audit partner
supply in New Zealand.

### **5.3.7 Conclusions**

The purpose of setting up a matrix for audit partner rotation is to manage the myriad of different regulations that have been put in place. The complexity of the matrix comes from (1) replace the incumbent partner with an audit partner with industry experience (2) managing an audit partners portfolio and capacity (3) retaining institutional knowledge in the engagement, and (4) have an audit manager and audit team members present in the engagement that have knowledge of an entity's clients' specific matters. Audit partner agreed that extended cooling-off periods were the biggest challenge to audit firms and that certain entity types were struggling to gain the services of audit firms. A significant conclusion from this research is that extended cooling-off periods in a small-scale economy are levers for market concentration and that by reducing cooling-off periods this will allow audit firms to better serve the entities that are legally required to be audited.

This research followed Dodgson et al. (2020) research in investigating audit partner transitions. Most of the findings from this study concurred with Dodgson et al. (2020), then but added to by finding that audit firms match personality types when assigning engagement

partners to clients and experience levels when matching the EQCR to the engagement partner. The purpose of matching personality types is so that an engagement partner can constrain a client's aggressive financial position while also facilitating the ease of information transference from the client to the audit team to ensure audit quality. Audit firms match age and experience levels between EQCR's and engagement partners so that an engagement partner (and vice versa) cannot dominate an EQCR when making significant judgements and so that the EQCR can make a credible challenges to the engagement partner and team without being at experience or age disadvantage.

### 5.4 Relationships between engagement partners and their clients

The purpose of audit partner rotation is to periodically end the engagement partner client relationship. Beattie et al. (2004) researched auditor client negotiations finding that primary auditor client relationships are important for gaining a quality. Following this research this study looks to explore the implications of audit partner rotation on auditor client negotiations.

This section of the chapter will discuss: Firstly, the impact of audit partner rotation on auditor-client relationships in the first years on an engagement. Secondly, the 'fresh perspective' and material risk assessments. Thirdly, that audit partner rotation leads to internal and external challenges. Fourthly, audit partners perceive multiple aspects of familiarity. Fifth, how audit partner rotation effects the relationship between the engagement partner and the management of the entity. Sixth, that directors perceive audit partner rotation as good governance. Seventh, that employees in client entities are constantly changing, and lastly the conclusions on engagement partner client relationships.

# 5.4.1 Audit partners face a knowledge and experience imbalance in the first year of the engagement

Researchers that argument against audit partner rotation do so on the basis that the financial statements are more likely to be misstated in the first few years of an audit partners engagement, while the engagement partner builds up knowledge and experience. This is a critical issue of audit partner rotation, audit partners in this study confirmed what was already known in research by providing evidence that in the first year of the engagement the audit partner is on a learning curve (Interviewee 2, Big4, 4, Big4), that this is a period of building key relationships (Interviewee 7, Big4), learning the big picture stuff (Interviewee 9, NonBig4) and potentially asking a lot of stupid questions (Interviewee 4, Big4). Interviewee 5, Big4 explains: "That in the first year of performing an audit, an audit partner is not going to be as familiar with the organization and the affordable risk, they won't perform as higher quality audit". The reference to affordable risk relates to the threats that an entity faces, and that in the first year of an audit partners engagement they are not yet familiar with the clients' specific matters which can potentially decrease audit quality. Concerningly, Interviewee 2, Big4 states that in the first year of an engagement an audit partner is likely to accept management interpretations of the key assumptions that have been made. This has significant implications in the auditor negotiations. What audit partners were communicating was that there is a knowledge and experience imbalance in the first year of an audit partners engagement creating a power imbalance in the relationship between the engagement partner and client. This knowledge and experience power imbalance can lead to management being able to exploit the engagement partners inexperience to potentially gain financial rewards.

The second year of the engagement is when the audit partner looks to rectify any mistakes that have been made in the first year of the engagement and address the relationship power imbalance. Research does not provide specific evidence on the second year of an audit

partners engagement. This study contributes to research by describing this process.

Engagement partners in the second year of the engagement starts picking up on things they have missed in the first year (Interviewee 6, NonBig4), the focus goes from the big picture to learning the details of the entity and gaining a greater understanding of their client specific matters (Interviewee 5, Big4). Interviewee 9, NonBig4 states that the second year is when you might fix anything that you may have got wrong, confirming the research perspective that the financial statemented are more likely to be misstated in the first year of an engagement. Participants described how in first years of an engagement the engagement partner is highly reliant on the knowledge retained amongst the engagement team and the relationship held between the audit manager and the client (Interviewee 4, Big4, Interviewee 11, NonBig4). Auditors challenge management in auditor negotiations. Interviewee 2, Big4 states how in: "The second year you go this is not working, your assumptions aren't working". This statement summarizes how the engagement partner in the second year of the engagement looks to rectify the power imbalance created by audit partner rotation by directly challenging the management of the entity on the assumptions that they have made. The implications of audit partner rotations are that management may take a more aggressive interpretation of the accounting standards in the first year of an audit partners engagement to gain financial rewards, reliant on an information and experience imbalance occurring through the transitional process. Critically, audit partner rotation puts the engagement partner in a position of unease to challenge management on the assumptions that they have made. The engagement partner in the second year has to then overcome this period of unease and show courage by challenging management on the assumptions they do not agree with.

# 5.4.2 The 'fresh perspective' is a re-assessment of the material threats that may threaten the entity

The purpose of implementing of audit partner rotation regulations is based on two definable concepts (1) a sociological perspective that attitudes and behaviours change in longer engagements (2) and that the incoming audit partner will bring a 'fresh perspective' to the audit engagement. The argument against mandatory audit partner rotation and a 'fresh perspective' is that due to the incoming partner being less informed about the client there is an increased likelihood that they may not identify a financial reporting problem. This study provided research evidence that the financial statements are more likely to be misstated in the first year of audit partners engagement. The argument for audit partner rotation and a 'fresh perspective' is that the incoming engagement will carry out a powerful peer review of the exiting engagement partners work (Lennox et al. 2014). Audit partners in this study provided the perspective that audit partner rotation resulted in them questioning balance sheet items and the judgements of exiting partners whilst increasing their ability to challenge the management of the firm.

Researchers that argue in favour of audit partner rotation, do so on the basis that the incoming partner will bring a 'fresh perspective' to the engagement, increasing audit quality by detecting and correcting mistakes in the financial statements. This statement may be true but implies that audit partner rotation is a stand-alone event. Dodgson et al. (2020) provides evidence that audit partner rotation is not a stand-alone event but is managed by the audit firm with the incoming engagement partner being involved in the audit process for as long as eighteen months to two years before they become the engagement partner, this research concurs with this literature providing further evidence that audit partner rotation is not a stand-alone event.

Dodgson et al. (2020) states that incoming audit partners may shadow the incumbent partner prior to becoming the engagement partner with additional resources being applied to larger more complex engagements. Pittman et al. (2022) provides evidence of connections between the incoming audit partner and the incumbent engagement partner and that audit firms may choose to groom an incoming audit partner into the engagement partner's position. This research finds evidence that on large client entity audits the engagement partner will have an audit partner that is second in charge and likely to ascend to the engagement partner's position. Pittman et al. (2022) states that an audit partner that is familiar with the entities client specific matters and has already established relationships, is likely to ascend to the engagement partner's position. That an incoming partner is involved in an audit for up to two years before hand brings into question whether they can bring a 'fresh perspective' to the audit engagement, it also implies that they are able to contribute and be involved in the judgements that have been made (Interviewee 13, Big4). This process allows the incoming audit partner to question the incumbent partner on the treatments of accounting standards and how they have determined that they are the correct. While this process may be highly valuable for the incoming partner it also provides a period of time when the incumbent partner is able to persuade the incoming partner that the judgements, they have made are the correct ones and dissuade the incoming partner from taking a fresh approach to previous judgments made.

There are now three sets of research that have found that audit firms manage audit partner rotation and that it is not a stand-alone event (Dodgson et al. 2020: Pittman et al. 2022). This brings into question prior research done by Myers et al. (2003), Carey and Simnet, (2006), Chi and Huang, (2005) and Chi et al. (2009) that have tried to discover the effects of audit partner rotation based on the principle that audit partner rotation is a stand-alone event and used quantitative measures, such as the proxies of abnormal accruals, going

concern, and misstatements to determine if there have been changes to audit quality. That audit partner rotation is not a stand-alone event explains why these researchers were not able to gain conclusive results. This highlights the importance of qualitative studies and the importance of having conversations with professionals in the area of research that is being studied.

Audit firms manage audit partner rotation to ensure audit quality in the engagement. This practice potentially inhibits the incoming engagement partner from bringing a 'fresh perspective' to the engagement. Whether regulators are aware of it or not they have countered audit firms inhibiting the 'fresh perspective' by requiring engagement partners to sign the audit report which can have a negative reputational consequences in the event of an audit failures. Though signing the audit opinion can also have positive reputational implications for the successful completion of larger more complex audits. In the event of a corporate collapse or where fraud can be established engagement partners can also be held personally liable and face criminal or civil prosecutions. What this creates is an environment of 'individual partner risk' motivating the incoming engagement partner to re-examine the material risks that threaten the entity and not take for granted the material assessments that have been done in the past. The construct of 'individual partner risk' also shifts the litigation risk away from the audit firm, placing increased responsibility on the engagement partner in the event of an audit failure. The effect of 'individual partner risk' is that it emboldens the engagement partner to challenge other or senior partners interpretations of accounting standards as well as being less likely to accept an aggressive financial position held by management, changing the relationship dynamic. Audit partners in this research were in agreement that the 'fresh perspective' leads to positive outcomes in the audit engagement, providing evidence that the incoming engagement partner will take the opportunity to reevaluate the material risks that the entity faces (Interviewee 13, Big4), questions the previous judgements made (Interviewee

12, Big4), use their experience from other engagements to question the treatments of accounting standards (Interviewee 4, Big4), increase the challenge that the engagement partner brings to the management of the entity (Interviewee 2, Big4) and change the relationship dynamic in auditor negotiations.

# 5.4.3 Audit partner rotation leads to internal and external challenges

The relationship between an auditor and a client is one where the auditor, when necessary, has to hold the management of the entity to account on their interpretations on the accounting standards and the judgements that they have made, which can potentially lead to conflict between the two parties. The engagement partner in this circumstance has to consider whether the impact of the management's interpretation is material to the entity and whether it would influence the audit opinion (Interviewee 6, NonBig4). If the engagement partner determines that the interpretation is material, then they must be prepared to have a courageous conversation with the client even when the potential change to that interpretation may place the entity in a difficult situation (Interviewee 2, Big4) such as reporting a loss instead of a profit. In this circumstance the relationship between the engagement partner and the client can have a significant impact of the potential outcome of these negations (Beattie et al. 2004). A newly rotated on engagement partner in this situation faces a power imbalance due to their lack of knowledge of client specific matters, inhibiting their ability to challenge management in auditor negotiations and having to rely on the audit manager and teams' relationships in mid to small-size engagements and other audit partners in large client entity audits.

Challenging a client on the assumptions that they have mad is the external outcome of the audit. Lennox et al. (2014) provides evidence of engagement partner internally peer reviewing each other's work. This study confirms a peer review process providing research evidence of incoming engagement partners reassessing the risks that may threaten the entity

and questioning the material benchmarks that have been applied (Interviewee 12, Big4, 13Big4). So far what has been discussed is the challenge that auditors present to the client. Changing engagement partners leads to the incoming engagement partner challenging the incumbents' judgements and decisions which can lead to changes in relationships between audit partners and power dynamics within the audit firm. Interviewee 7, Big4 explains: "One of the internal challenges is obviously having the confidence to come to a different view than your previous partner had, which happens a lot". The foreseeable difficulty for a newly promoted audit partner is challenging a senior partner in the audit firm that has a distinctive power advantage, potentially dissuading the junior partner from challenge the senior partner should the junior audit partner determine that challenging the senior audit partner could have negative career consequences. Following Lennox et al.'s (2014) research and assuming citrus paribus in audit partner relationships, audit partner rotation should promote the incoming engagement partner challenging the assumptions and judgements made by the incumbent's partner. Audit partners in this study perceived that the incoming engagement partner will bring their experiences and knowledge from previous client engagements to bare when assessing the material risks of the entity and judgements made by the incumbent engagement partner (Interviewee 4, Big4, Interviewee 13, Big4). Making audit partner rotation an audit quality increasing exercise from the continual reassessment of the material assumptions made by the entity.

The New Zealand regulatory agency has created a regulatory environment of 'individual partner risk' this environment has meant that both the engagement partner and audit firm face consequences in the event of an audit failure. Audit firms generally have increased resources compared to an individual audit partner. The reputational damage done to an individual engagement partner from an audit failure may mean that they will not be able to work in the industry again and was described as a highly personal matter (Interviewee 12

Big4). The creation of 'individual partner risk' by the regulatory agency changes relationships within the audit firm. Audit partners need to consider the risk to themselves when deciding whether to challenge the judgments of the exiting partner. The combination of regulatory actions that could be taken by the regulator, motivates audit partners not to blindly accept the judgements that has been made in the past but to be sceptical of the judgements promoting a revaluation and confirmation process to evaluate how those judgements have been derived at before challenging the judgement if the incoming audit partner disagrees with the exiting partners interpretation.

The combination of audit partner rotation and other regulations changes the relationships not just between the engagement partner and the client but also impacts partner to partner relationships. An additional consequence of 'Individual partner risk' is that not only does it motivates junior partners to challenge senior partners on the judgements they have made but creates an environment where the engagement partner is less likely to acquiesce to pressures within the audit firm. The growing power of large corporate entities has been of concern to many people since the 1980's (Frick, 2018) gaining increased political power (Young et al. 2020). The dominance of Big Four audit firms and their alignment with large corporate entities is of serious concern in both the UK and US (Lagace, 2013: Williams-Alvarez, 2022). Regulators by creating 'individual partner risk' are perceptibly altering the large corporate entity, large audit firm relationship by making the engagement partner personally accountable in the engagement. Because of the reputational risk the individual partner faces they are less likely to give into pressures from large corporate entities and except the aggressive treatments of accounting standards. This alters the power dynamics within the audit firm, the successful challenge of an interpretation may elevate that partner within the firm gaining prestige and financial rewards, though power structures in audit firms

are likely to be complicated and could result in the opposite effect especially if it effects a large powerful entity.

# 5.4.4 Audit partner rotation ends the relationship between the engagement partner and the client

The ethical principle of audit partner rotation is based in social psychology, which provides evidence of changing attitudes and behaviours in longer relationships. Based on these principles, the construct of a 'familiarity threat' has been developed that implies that long relationships lead to engagement partners been susceptible to the opportunistic interpretations of accounting decisions by management. This research provides evidence that most of the audit partners perceived familiarity to be a risk when engagement partners carry out engagements for years and years, though some partners were critical doubting that audit partner rotation was suitable for the New Zealand environment based on the size of the economy (Interviewee 1, NonBig4, 5, Big4, 6, NonBig4). An interesting finding in this study was that audit partners perceived multiple dimensions of familiarity, the first being that over familiarity may cause engagement partners to miss things in the engagement and that familiarity may inhibit an engagement partner from asking challenging questions. Interviewee 6, NonBig4 assertively puts it as: "Because it's a judgment call someone should challenge that judgment and if someone is too familiar, they may not challenge". This statement confirms the social psychology perspective that long relationships influence attitudes and behaviours, in this instance decreasing the engagement partners willingness to challenge judgements. That engagement partners are likely to miss things is more likely to come from the engagement partners being susceptible to a self-serving bias of believing that they know all the threats that an entity faces.

Audit partners in this study provide research evidence that agrees with both sides of the argument surrounding audit partner rotation. Firstly, the position held by regulators that long relationships threaten an engagement partners ability to challenge aggressive financial positions and secondly, researchers that argue that familiarity with a client's specific matters enhances the engagements partners abilities to provide a higher quality audit. In direct contrast to over familiarity causing audit partners to potentially miss something, they also provide the perspective that a lack of familiarity is a risk to audit quality. This aspect of familiarity is where an engagement partner gains an insight into the entity so that they have a greater understanding of audit and fraud risks. Interviewee 1, NonBig4 explains: "You do get an awful lot of accumulative audit knowledge of clients when you are working on them, you know where the risks are, you know the history of some of the transactions they have been involved with in the past". This implies a deeper understanding of the entity and the risks they may be exposed to. A common proxy used to measure audit quality in research is abnormal income increasing accruals. An historic understanding of an entities transactions gives the engagement partner an understanding of what the acceptable level is of an abnormal income increasing accrual which could safeguard the investor against aggressive financial decisions made by management. Audit partners perceived that the biggest risk to audit quality was changing the entire audit team, directly relating this to the loss of institutional knowledge and the disruptions to relationships. Social psychology provides evidence of attitudinal and behavioural changes in long relationships. Regulatory agencies associate these behavioural traits negatively towards a perceived 'familiarity threat' which is confirmed by audit partners. The contrasted viewpoint in social psychology is that these behavioural changes can also positively impact relationships through trust. Increased trust in a relationship can lead to a greater insights into the entity from individual actors in the organisation being willing to divulge more information's giving the engagement partners a greater understanding of their client specific matters, with the consequence being increased audit quality (Wilson et al. 2017).

# 5.4.5 Audit partner rotation disrupts the relationship between the engagement partner and management

Literature on auditor client relationships focuses on how long tenures result in management being able to influence the engagement partner into accepting aggressive financial decisions. The reality as perceived by New Zealand audit partners is that the engagement partner and audit teams need good relationships with management to ensure a quality audit. This research provides evidence that the management of the entity view audit partner rotation as an inconvenience because of the practical reality that they have to teach everything again to the incoming engagement partner who is likely to ask a lot of stupid questions (Interviewee 4, Big4, 6, NonBig4, 8, Big4). Interviewee 4, Big4 explains: "The reason I'm asking questions is because I'm now the partner and I need to understand it.... that's part of rotation". Audit partners provided the perspective that the incoming partner has to spend a lot of time asking questions about balances that have already been signed off by previous partners, potentially for several years, frustrating and confusing the client as to why the engagement partner is interested in balance sheet items that have been there for years.

This research provides evidence of how management like consistent relationships, across the whole engagement team. Because from a management perspective they want auditors that know how their entity works (Interviewee 3, NonBig4). Engagement partners do not interact with management on a daily basis. Most interactions occur between audit manager and team that are completing the practical work of proving balances, testing transactional data and checking contracts, while the engagement partner and EQCR are more involved at the entity, commercial risk level (Interviewee 1, NonBig4). Contrary to the focus of a lot of research it is the audit manager that has a close relationship with management, they are likely to see them every day and can be described as the go to person when management have questions or problems (Interviewee 1, NonBig4, 5, Big4). In line with the management

of the entity the audit firm wants consistency of skills and knowledge within the engagement and the continuance of established relationships that ensure a quality audits (Interviewee 1, NonBig4, 2, Big4, 4, Big4, 5, Big4).

## 5.4.6 Boards of directors perceive audit partner rotation as a best practice

The establishment of a regulatory regime and audit partner rotation has had a significant effect on audit partners and the directors of New Zealand entities. Directors can be held personally liable and be sued for a breach in their duties (Rankin, 2022). The duties on a director include ensuring the compliance of the financial statements and that they represent a true and fair view of the entity (FMA, 2018). A number of participants in this study perceived that there was a disparity between the obligations of management (Interviewee 12, Big4, 13, Big4) and those of the directors and auditors (Interviewee 8, Big4, 9, NonBig4). Conceiving that directors and auditors have 'skin in the game' in the event of corporate failure.

This research has provided evidence that the management of the entity perceive audit partner rotation to be an inconvenience. In contrast to this the perceptions of audit partners in this study were that directors perceive audit partner rotation to be good governance.

Participants made reference that some directors had little understanding of audit partner rotation but the boards that do wanted to implement best practices (Interviewee 4, Big4).

Interviewee 13, Big4 explains: "There the ones pushing for the quality, they're the ones pushing for the right disclosures and the papers supporting judgments..... it's their responsibility to assess the independence of the auditors". It is a director's responsibility to ensure that the audit firms meet its compliance requirements (FMA, 2020). From a directors' perspective audit partner rotation achieves two governance goals (1) that it increases the independence of the audit firms and (2) that the material risks to the entity have been reassessed.

A positive from the new regulatory regime is that it has increased the communications between engagement partners and directors. Interviewee 8, Big4 stated that directors are now seeking time alone with the engagement partner. This can be interpreted as directors seeking additional assurance that their obligations to the investors are being met. Interviewee 7, Big4 explained how the establishment of a regulatory regime has had the effect of aligning the goals between the directors and auditors to protect the shareholders of the entity. What the regulatory regime has achieved is to change the relationship dynamic between the engagement partner, directors and management of the entity. The perceptual difference is that due to directors and auditors having 'skin in the game' directors are now acting as intermediaries in auditor negotiations and have an increased understanding of their responsibilities to support the audit function (Beattie et al. 2004). This research provides evidence that the new regulatory regime and audit partner rotation has strengthened the relationship between directors and auditors by increasing their responsibilities making both parties more aligned in protecting shareholder wealth.

## 5.4.7 Employees in client entities are constantly changing

The passing of the Sarbanes Oxley Act in 2002 resulted in the tightening of audit partner rotation regulations and increasing of cooling-off periods in the US. New Zealand started implemented audit partner rotation regulation in 2004. In congruence to this there has been generational change in workplace demographics. This demographical change is that the workplace is now dominated by Millennials as opposed to Generation X (Martic, 2023). This generational change has resulted in employees spending less time with individual employers, on average five- years. This has significant implications for audit partner rotation policies and cooling-off periods. This research provides evidence that due to this generational change it is very unlikely to have consistency of employees amongst directors, managers or audit teams. Which has resulted in a natural control of the familiarity risk (Interviewee 2, Big4, 3,

NonBig4, 6, NonBig4, 12, Big4, 13, Big4). Interviewee 6, NonBig4 explains: "I had to come off one of my jobs last year, where the CFO had changed and the complete board was different, I didn't really know any of those people at all, but I would have been too familiar". Evidence from this research posses' specific problems for regulators. The generational change that has occurred directly brings into question the regulators reasoning for extended cooling-off periods but not audit partner rotation because audit partner rotation can be considered an audit quality increasing exercise.

Audit partners are subjected to a five-year cooling-off period following an engagement for NZX and FMC entities. The reasoning that regulators give for audit partner rotation and extended cooling-off periods is that a self-interest threat might develop from an individual concern about losing a longstanding client or an interest in a close personal relationship may develop between a member of the senior management or those in charge of governance (XRB, 2022). The interpretation of this is that a long and close relationships can affect an engagement partners judgement and that the regulatory agency considers a cooling-off period of five-years sufficient time for this threat to be extinguished. The problem with the regulatory reasoning is that according to statistical evidence supplied by the US

Department of Statistics, research information and evidence provided by audit partners in this research is that this threat no longer exists or has been significantly reduced, with a natural control developing from a generational change that prevents long relationships developing in the engagement leading to the conclusion that five-year cooling-off periods are no longer necessary and should be revised by the regulatory agency.

#### **5.4.8 Conclusions**

This research investigated the first and second year of the audit partners engagement providing qualitative research evidence that audited financial statements are more likely to be misstated in the first year of the engagement and that audit partner looks to rectify any

mistakes made in the second year of the engagement. This research concurs with prior research that due to an engagement partner not being as familiar with an entity's client specific matters, engagements partners are more accepting of managements interpretations of accounting standards in the first year of an engagement which suggest management are more likely to management earnings while the engagement partner builds up knowledge and experience. The 'fresh perspective' of the incoming audit partner counters the loss of client specific knowledge in the first year of an engagement and can be described as a critical reassessment of the material risks that threaten the entity and previous judgements that have been made. The implications of this are that audit partner rotation and the 'fresh perspective' becomes a governance issue ensuring independence and that the risks an entity faces are continually being re-assessed.

Audit partners in this research confirmed the social psychology perspective that long relationships lead to behavioural changes between engagement partners and clients. Finding that engagement partners are less likely to challenge managements treatments of accounting standards in auditor client negotiations when they have become over familiar with the client. However, it can be concluded that engagement partners face an information's and experience imbalance in the first year of an engagement creating a power imbalance that favours the management of the entity in auditor negotiations. Regulators have introduced 'individual partner risk' into the audit engagement by (1) making audit partner rotation mandatory (2) requiring audit partners to sign the audit opinion for NZX and FMC audits, and (3) making the engagement partner liable in criminal and civil prosecutions. This has effectively isolated the engagement partner from the audit firm in event of audit failure or corporate collapse. Regulators have changed the relationship dynamics both internally and externally by creating 'individual partner risk'. Internally this means that the engagement partner is more likely to

challenge other or senior partners judgments, externally this means that the engagement partner will be less likely to except judgements by the entity that can be considered material.

This research has provided evidence that auditors are often going into an engagement where there has been significant change within the entity. Which is backed up by data from the US department of statistics and research evidence that Millennials changes jobs on average every five years. This has significant implications for cooling-off periods. A generational change has resulted in long auditor client relationships becoming less likely to occur. This research has concluded that extended cooling-off period pose the biggest problem to audit firms in New Zealand. This finding has significant implications to regulators.

Regulators should consider reducing cooling-off periods to increase the availability of licensed audit partners.

# Chapter 6: Conclusions, limitations and future research.

The purpose of this research was to gain the perspectives of audit partners on the implementation of audit partner rotation regulations. This chapter of the research will: Firstly, draw conclusions to directly answer the research questions. Secondly, it will highlight how the observations made in this research can contribute to theory. Thirdly, it will explore areas of future research, and fourthly it will explain the limitations of this research.

#### **6.1 Conclusions**

The first research question: "How do audit partners perceive the imposition of mandatory audit partner rotation regulations?" can be answered by communicating the shared perspectives of audit partners on the identified research problem.

This research found that regulators design regulations based on their perceptions of the environment they are regulating. The audit partner rotation regulations that have been implemented in New Zealand are designed for large-scale commercial entities and large-scale audit firms. This research has provided qualitative research evidence that reduced time-on periods are suitable to large client entity audits because they have increased institutional knowledge present in the engagement. New Zealand has mainly small to medium-sized commercial entities with a few commercial entities and a relatively small number of audit firms. The reason audit partners perceived audit partner rotation regulations to disproportionately effect small and mid-sized audit firms is because the regulations are not designed for small and mid-sized audit firms and have not been 'right sized' to meet New Zealand's demographic environment. What audit partners expressed was that the regulatory approach is disconnected from the practical realities of New Zealand audit firms. Which has resulted in some small and mid-sized audit firms not having sufficient audit partner resources to stay in the domestically registered audit market, which has created the perception that the current regulatory environment may not be sustainable.

The implications of introducing regulation based on standardization rather than on the demographic environment is that audit firms have to compensate for the regulatory disconnection, to the practical reality of limited audit firm resources. Positively mandatory audit partner rotation has led to a more even distribution of skills amongst audit partners and audit team members, negatively it leads to audit firms not being able to appoint the most suitable audit partner to the engagement. Audit partners perceived that mandatory audit partner rotation leads to increased reliance on the audit manager and audit team. Audit quality at this level could be assumed to be constant or improving. Appointing an engagement partner with no or limited industry experience is a threat to audit quality and requires the audit firm to apply more resources to the engagement to ensure that audit quality isn't compromised at the commercial risk, entity level.

To second research question "What are the implications of making audit partner rotation mandatory in New Zealand?" is answered by how audit firms have responded to the imposition of mandatory audit partner rotation regulations.

New Zealand has implemented a complex structures of audit partner rotation policies, in response to this regulatory structure audit firms have set up planning and coordination mechanism (e.g., matrixes and logs) to manage audit partner rotation. The purpose of a matrix is to provide practical solutions to the myriad of regulations that have been introduced, while also incorporating the more nuanced aspects of audit partner rotation. The practical application of a matrix is to ensure that there is institutional knowledge and client specific knowledge present in the engagement. It should also determine who the most suitable replacement audit partner is based on industry experience, while matching age and experience levels of the EQCR and engagement partners so that one cannot dominant the other. The more nuanced aspects of audit partner rotation are in matching the personalities of engagement partners to clients. On large client entity audits, audit firms are able to groom

audit partners into the engagement partners role (Dodgson et al. 2020: Pittman et al. 2022), though this may not always work out if their client does not think there is the right chemistry for a relationship to develop. Regulators should understand the industry they are regulating. This research provided qualitative research evidence that audit firms were already rotating audit partners before the regulations were made mandatory.

The difference between Big Four and non-Big Four audit firms is that Big Four audit firms having significantly more audit partner resources changing the dynamic, to matching personalities and being able to provide client entities with a selection of engagement partners to choose from, though they may not be able to provide the most suitable engagement partner due to extended cooling-off periods. Big Four audit firms have an advantage in transitioning audit partners in large client entity audits where there are multiple partners involved. This allows the audit partner to be involved in the audit for potentially a number of years in advance before becoming the engagement partner so that the audit partners can become familiar with the clients' specific matters and form relationships. This effectively counters the lack of knowledge and experience power imbalance that can occur in the first year of an engagement and ensures that the engagement partner can present a robust challenge in auditor negotiations, but negatively impacts the 'fresh perspective' that an engagement partner should impart.

# **6.2** Contributions to practice

Firstly, this research contributes to practice by identifying how regulations should 'right sized' when transposing regulations from large economies to small economies.

Secondly, this research contributes to theory by providing research on how audit firms should develop a matrix to manage audit partner rotation.

### 6.2.1 Right sizing regulations to an economies GDP and population

The identifiable problem in this research was that New Zealand has transposed audit partner regulations from countries with significant larger GDP's and populations. The implications of this have been that imposed regulations have disproportionately disadvantaged small and mid-tier audit firms that haven't got the audit partner resources to be able to manage the regulations. Audit partners perceived that extended cooling-off periods for engagement partners and EQCR's provide the biggest challenge for audit firms in a small-scale economy.

This research was a qualitative inductive study that has been able to contribute to the theory development of transposing regulations from a large-scale developed economy to small-scaled developed economy (Morgan and Quack, 2006). The purpose of an inductive study is to generate theory (Gasson, 2004). New Zealand regulators initially matched cooling-off periods to the size of the New Zealand audit market then extended them to match regulations imposed on large-scale economies. This has resulted in increased market concentration and certain entity types that are legally required to be audited having trouble gaining the services of audit firms. The theoretical contribution from this research is that imposed regulations need to be 'right sized' to meet the resources available to the audit firms that are operating in the market. By extending cooling-off periods regulators forced some small and mid-tiered audit firms out of the audit market or merge to increase their audit partner resources.

Like most countries New Zealand legally requires certain entity types to be audited.

This implies a responsibility on the regulators part to not impede the supply of audit firms providing that service. In implementing regulations, regulators should identify the market participants that are most likely to be affected by the regulations and the sector of the market that those participants supply services too. Analysis should them be performed to determine if

the market is able to absorb the excess demand without imposing significant cost increase on entities that have been affected by the regulations.

# 6.2.2 Implementing a matrix for audit partner rotation management

This research discovered that audit firms use a matrix to manage audit partner rotation. Because this was new to research the researcher matched the matrix of audit partner rotation to human resource management to provide literature on how organisations incorporate matrixes for employee management. Combining human resource management with the qualitative research evidence in this research allows the researcher to develop a theory on how audit firms could manage a complex multi-dimensional structure of audit partner rotation.

The first step in constructing a matrix for audit partner is to create a computer model for all of regulations that have been put in place. For New Zealand audit firms this would include a three-tiered system for time-on periods followed by five-year cooling-off periods. Time-on periods and cooling-off periods for EQCR's and other key audit partner would be the next step in generating a computer model. The roles that are required by each entity types should then be incorporated into the matrix as well as the industry experience and specialist skills of each audit partner as well as audit managers and team members with specialist knowledge. Age and years of experience in specific industries will allow the correct matching of audit partners and EQCR's.

The purpose of matrix is so that the audit firm can maintain institutional knowledge within the engagement. Audit partner discussed expanding the matrix to cover audit managers but not team members. Incorporating audit managers into the matrix will allow an audit firms to increase the knowledge of client specific matters into the engagement. Based

on industry experience the matrix should identify potential candidates for replacing engagement partners when they are required to rotate.

This research provided qualitative evidence that audit firms profile the personality type of their clients so that they can match the engagement partner to the client. The matrix should contain personality type data on both clients and engagement partners to achieve business chemistry in the auditor client relationship and so that the engagement partner can challenge management on their assumptions and gain favourable outcomes (Beattie et al. 2004). The matrix should also provide the most suitable EQCR's based on personalities, age, and experience so that they can provide a suitable challenge to the engagement partner and audit team.

#### **6.3** Future research

Two future research projects were identified in thesis from discussions with audit partners. The first was mandatory audit firm rotation which can be considered a natural progression from audit partner rotation. The second study is designed to establish if engagement partners are completing their engagements and whether gender influences.

## 6.3.1 Mandatory audit firm rotation

The topic of mandatory audit firm rotation was a constant theme throughout this research. Some audit partners perceived that its implementation was inevitable in New Zealand and that mandatory audit firm rotation posed a threat to audit. Interviewee 4, Big4 stated that the first audit a company does for a client entity is the hardest. Prior research done on the subject of mandatory audit partner rotation has been mainly quantitative using proxies for audit quality and based on the assumptions that audit partner rotation is a stand-alone event. The results from quantitative prior research have not provided definitive results. This research Dodgson et al. (2020) and Pittman et al. (2022) provide evidence that the reason no

definitive results have been gained from quantitative research is because audit partner rotation is not a stand-alone event but managed by the audit firm. Mandatory audit firm rotation be considered a stand-alone event. Prior research has found that the long tenure of audit firms is associated to increased audit quality (Jackson et al. 2008). Capital market actors such as banks and insurance companies associate long tenure to financial reporting accuracy and credibility (Ball et al. 2008). This indicates that if the regulatory framework progresses towards mandatory audit firm rotation the result could be a decrease in audit quality.

Four audit partners in this research volunteered to help with future research into mandatory audit firm rotation. This would be a mixed methods study that uses qualitative semi-structure interviews with audit partners to determine the correct proxies to use in a quantitative study. Audit partners indicated that the proxies to use for audit firm rotation were the impairment of intangibles measured against information's contained in the cash flow statement. The purpose of this study would be to determine if earnings quality has been affected by changing the audit firm.

In 2016 audit tendering was made mandatory in the UK for FTSE 350 entities (FRC, 2014). Between 2013 and 2016 a hundred and twenty-two FTSE 350 entities put there audits out for tender (Agnew, 2016). Which FTSE 350 entities have changed audit firms since 2016 is publicly available information through the FTSE index and the entities annual report, which provides a sample for this research. The approach to this research could be to emulate the research done by Myers et al. (2003) and Carey and Simnet (2006) using descriptive statistics, regression analysis, and a modified Jones Model.

### 6.3.3 The implications of gender on audit engagement completions

Audit partner in this research provided qualitative research evidence that there is often significant change amongst directors, managers and staff members of the client entity they are auditing as well as audit team members. That employees are changing employers more

regularly is backed up by the US department of statistics and prior research establishing a five-year career itch. That employee no longer spend as long with employers has been attributed to a generational change from Generation X to Millennials (Martic, 2023). The 2018 working life survey reported that 38% of employees had been with there current employer for more than five-years in New Zealand (StatsNZ).

Audit partner rotation regulation in New Zealand range from five to seven-years for time-on periods. This research and the generational change raises a number of questions: "Are engagement partner on average completing their time-on engagement periods?" And "how long are engagement partners on average spending in engagements". Research done by Carey and Simnet (2006) and Chai and Huang (2005) provide research evidence that the optimal time-on periods for engagement partners is seven-years. Initially like the UK, New Zealand, Australia, US, and Canada implemented audit partner rotation regulations at five-year time-on periods for its listed entities. In 2010 Canada reversed its five-year time-on periods and implemented seven-year time-on periods for engagement partners auditing listed entities (Daugherty et al. 2012).

Similar to New Zealand audit reports in Canada contains the independent engagement partner's name. The proposed sample for this research are Toronto Stock Exchange (TSX) entities because of Canada's seven-year time-on periods. The increased time-on periods will allow the researcher to provide research evidence on whether engagement partners are more likely to complete five or seven-year engagements. This research project could also provide gender specific information such as: What percentage of male and female engagement partners audit TSX entities? Does gender impact the number of years the engagement partner spends in the engagement? and are males or females more likely to complete time-on periods?

Chin and Chi, (2008) Gold et al. (2009) and Ittonen and Pen, (2012) identified sex as an individual characteristic worthy of examination. Landivar (2015) research found that females work fewer hours than their spouses. While Slone (2017) provides research evidence that females are more committed to state employment. What is being proposed is a quantitative research study that uses descriptive statistics to provide quantitative research evidence on average time-on periods for female and male engagements partners using the annual reports of TSX entities.

#### **6.4 Limitations**

A critical question in a qualitative research project is whether saturation has been reached. The researcher did not consider that saturation had been reached but did find that participants near the end of this research project were generally providing views that had already been expressed by other participants. The sample for this research was taken from the New Zealand Auditors Register limiting the study to licensed auditors. This excluded a significant number of auditors from this study including auditors that may have left the licensed audit regime due to overregulation.

Taking the sample from the New Zealand's Auditor Register limited the sample to 172 licenced audit partners 28 of which resided in Australia and 1 was located in Singapore. From this sample 13 interviews were conducted consisting of fifteen interviewees, a participation rate 8.72% or 10.48% if audit partners from Australia and Singapore were removed. No audit partners from Australia or Singapore took part in the research limiting the study to perspectives from New Zealand audit partners that resided and were employed within New Zealand. The researcher sent out 26 follow up emails to participants that didn't respond to the first email that was sent, this did not result in any further participants taking part in the research.

This study failed to gain a demographic mix of participants. Only one female took part in the research and most participant ages were between forty-five to sixty. What this meant was that all participants in this study apart from two had experienced the self-regulated environment, followed by the implementation of a state regulatory regime including audit partner rotation. The implication of this is that certain biases may have been exhibited through this study from participants that had experienced both regimes.

This research was subject to financial, word count, and time limitations. The time and word count limitations restricted the researcher from being able to explore the research in full. The financial constraints impacted the amount of travel that the researcher could undertake.

This thesis is for a master's degree in accounting, a theoretical framework is not a requirement for master's degree. Consequently, this research lacked conceptualisation in its research design and research questions (Grant and Osanloo, 2016) connections to the literature review (Rocco and Plankhotnik, 2009) and viewing the data through a theoretical lens (Collins et al. 2018). With additional time and resources, the researcher could have used sociology to gain a greater understanding of the social relationships and interactions that were described by audit partner between, clients, and audit team members (Pentland, 1993).

#### References

- Abbott, L. J., & Parker, S. (2000). Auditor selection and audit committee characteristics.

  \*Auditing: A Journal of Practice & Theory, 19(2), 47–66.

  https://doi.org/10.2308/aud.2000.19.2.47
- Andon, P., Free, C., & O'Dwyer, B. (2015). Annexing new audit spaces: challenges and adaptations. *Accounting, Auditing & Accountability Journal*. https://doi.org/10.1108/AAAJ-01-2015-1932
- Andrews, T. (2012). What is social constructionism?. *Grounded Theory Review*, 11(1).
- Archer, C., & de Bussy, N. (2006). Reputation or relationships: what really drives perceptions of employer attractiveness in the health industry?. *Asia Pacific Public Relations Journal*, 7(2006), 209-221.
- Arthur, N., Endrawes, M., & Ho, S. (2017). Impact of partner change on audit quality: An analysis of partner and firm specialisation effects. *Australian Accounting*\*Review, 27(4), 368-381. https://doi.org/10.1111/auar.12150
- Baird, L., & Meshoulam, I. (1988). Managing two fits of strategic human resource management. *Academy of Management Review*, *13*(1), 116-128. https://doi.org/10.2307/258359
- Ball, R., Bushman, R. M., & Vasvari, F. P. (2008). The debt-contracting value of accounting information and loan syndicate structure. *Journal of Accounting Research*, 46(2), 247-287. https://doi.org/10.1111/j.1475-679X.2008.00273.x
- Bamber, E. M., & Bamber, L. S. (2009). Discussion of Mandatory Audit Partner Rotation,

  Audit Quality, and Market Perception: Evidence from Taiwan". *Contemporary*Accounting Research, 26(2), 393-402. https://doi.org/10.1506.car.26.2.3

- Barth, M. E., Landsman, W. R., & Lang, M. H. (2008). International accounting standards and accounting quality. *Journal of Accounting Research*, 46(3), 467-498. https://doi.org/10.1111/j.1475-679X.2008.00287.x
- Bartlett, C. A., & Ghoshal, S. (1993). Beyond the M-form: Toward a managerial theory of the firm. *Strategic Management Journal*, *14*(S2), 23-46. https://doi.org/10.1002/smj.4250141005
- Baxter, P., & Jack, S. (2008). Qualitative case study methodology: Study design and implementation for novice researchers. *The Qualitative Report*, *13*(4), 544-559.
- Beattie, V., Fearnley, S., & Brandt, R. (2000). Behind the audit report: A descriptive study of discussions and negotiations between auditors and directors. *International Journal of Auditing*, 4(2), 177-202. https://doi.org/10.1111/1099–1123.00312
- Beattie, V., Fearnley, S., & Brandt, R. (2004). A grounded theory model of auditor-client negotiations. *International Journal of Auditing*, 8(1), 1-19. https://doi.org/10.1111/j.1099-1123.2004.00225
- Beattie, V., Fearnley, S., & Hines, T. (2013). Perceptions of factors affecting audit quality in the post-SOX UK regulatory environment. *Accounting and Business Research*, 43(1), 56-81.
- Bedard, J. C., Deis, D. R., Curtis, M. B., & Jenkins, J. G. (2008). Risk monitoring and control in audit firms: A research synthesis. *Auditing: A Journal of Practice & Theory*, 27(1), 187-218. https://doi.org/10.2308/aud.2008.27.1.187
- Bedard, J. C., & Johnstone, K. M. (2010). Audit partner tenure and audit planning and pricing. *Auditing: A Journal of Practice & Theory*, 29(2), 45-70. https://doi.org/10.2308/aud.2010.29.2.45

- Bell, E., Bryman, A., & Harley, B. (2022). *Business research methods*. Oxford university press.
- Bharath, S. T., Sunder, J., & Sunder, S. V. (2008). Accounting quality and debt contracting. *The Accounting Review*, 83(1), 1-28. https://doi.org/10.2308/accr.2008.83.1.1
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research* in *Psychology*, 3(2), 77-101. https://doi.org/10.1191/1478088706qp063oa
- Braun, V., & Clarke, V. (2019). Reflecting on reflexive thematic analysis. *Qualitative Research in Sport, Exercise and Health*, 11(4), 589-597. https://doi.org/10.1080/2159676X.2019.1628806
- Burningham, K., & Cooper, G. (1999). Being constructive: Social constructionism and the environment. *Sociology*, *33*(2), 296-316. https://doi.org/10.1177/S0038038599000280
- Cai, Y., & Li, M. (2022). CEO-CFO tenure consistency and audit fees. *Pacific-Basin Finance Journal*, 73, 101779. https://doi.org/10.1016/j.pacfin.2022.101779
- Cameran, M., Prencipe, A., & Trombetta, M. (2016). Mandatory audit firm rotation and audit quality. *European Accounting Review*, 25(1), 35-58. https://doi.org/10.1080/09638180.2014.921446
- Carcello, J. V., Hermanson, R. H., & McGrath, N. T. (1992). Audit quality attributes: The perceptions of audit partners, preparers, and financial statement users. *Auditing*, *11*(1), 1.
- Carey, P. & Simnet, R. (2006). Audit partner tenure and audit quality. *The Accounting Review*, 81, 653–676. https://doi.org/10.2308/accr.2006.81.3.653

- Chen, C. Y., Lin, C. J., & Lin, Y. C. (2008). Audit partner tenure, audit firm tenure, and discretionary accruals: Does long auditor tenure impair earnings quality?. *Contemporary Accounting Research*, 25(2), 415-445. https://doi.org/10.1506/car.25.2.5
- Chi, W., & Huang, H. (2005). Discretionary accruals, audit-firm tenure and audit-partner tenure: Empirical evidence from taiwan. *Journal of Contemporary Accounting & Economics*, 1(1), 65-92. http://doi.org/10.1506/car.25.2.5
- Chi, W., Douthett Jr, E. B., & Lisic, L. L. (2012). Client importance and audit partner independence. *Journal of Accounting and Public Policy*, *31*(3), 320-336. https://doi.org/10.1016/j.jaccpubpol.2011.08.009
- Chi, W., Huang, H., Liao, Y., & Xie, H. (2009). Mandatory audit partner rotation, audit quality, and market perception: Evidence from Taiwan. *Contemporary Accounting Research*, 26(2), 359-391. https://doi.org/10.1506/car.26.2.3
- Chia-Ah, E., & Karlsson, J. (2010). The impact of extended audit tenure on auditor independence: Auditors perspective.
- Choi, J. H., Kim, J. B., & Zang, Y. (2010). Do abnormally high audit fees impair audit quality?. *Auditing: a Journal of Practice & Theory*, 29(2), 115-140. https://doi.org/10.2308/aud.2010.29.2115
- Christensen, B. E., Newton, N. J., & Wilkins, M. S. (2021). How do team workloads and team staffing affect the audit? Archival evidence from US audits. *Accounting*,

  Organizations and Society, 92, 101225. https://doi.org/10.10.16/j.aos.2021.101225
- Christfort, K., & Vickberg, S. (2018). Business Chemistry: Practical Magic for Crafting Powerful Work Relationships. John Wiley & Sons.

- Cloutier, O., Felusiak, L., Hill, C., & Pemberton-Jones, E. J. (2015). The importance of developing strategies for employee retention. *Journal of Leadership, Accountability & Ethics*, 12(2).
- Coates, I. V., & John, C. (2007). The goals and promise of the Sarbanes-Oxley Act. *Journal of economic perspectives*, 21(1), 91-116. https://doi.org/10.1257/jep.211.91
- Cohen, D. A., Dey, A., & Lys, T. Z. (2008). Real and accrual-based earnings management in the pre-and post-Sarbanes-Oxley periods. *The Accounting Review*, 83(3), 757-787. https://doi.org/10.2308/accr.2008.83.3.757
- Collins, C. S., & Stockton, C. M. (2018). The central role of theory in qualitative research. *International Journal of Qualitative Methods*, *17*(1), 1609406918797475. https://doi.org/10.1177/1609406918797475
- Daniels, K., & Harris, C. (2000). Work, psychological well-being and performance.

  \*\*Occupational Medicine (Oxford), 50(5), 304-309. https://doi.org/10.1093/occmed/50.5.304
- Daugherty, B. E., Dickins, D., Hatfield, R. C., & Higgs, J. L. (2012). An examination of partner perceptions of partner rotation: Direct and indirect consequences to audit quality. *Auditing: A Journal of Practice & Theory*, *31*(1), 97-114. https://doi.org/10.2308/ajpt–10193
- Daugherty, B. E., Dickins, D., Hatfield, R. C., & Higgs, J. L. (2013). Mandatory audit partner rotation: Perceptions of audit quality consequences. *Current Issues in Auditing*, 7(1), P30-P35. https://doi.org/10.2308/ciia-50446
- Davies, C. (1996). The sociology of professions and the profession of gender. *Sociology*, 30(4), 661-678. https://doi.org/10.1177/0038038596030004003

- De Smet, A., Dowling, B., Hancock, B., & Schaninger, B. (2022). The Great Attrition is making hiring harder. Are you searching the right talent pools. *McKinsey Quarterly*, 1-13.
- DeAngelo, L. E. (1981). Auditor size and audit quality. *Journal of Accounting and Economics*, 3(3), 183-199.
- DeFond, M. L. (2010). How should the auditors be audited? Comparing the PCAOB inspections with the AICPA peer reviews. *Journal of Accounting and Economics*, 49(1-2), 104-108. https://doi.org/10.1016/j.jacceco.2009.04.003
- DeFond, M. L., & Francis, J. R. (2005). Audit research after sarbanes-oxley. *Auditing: A Journal of Practice & Theory*, 24(s-1), 5-30. https://doli.org/10.2308/aud.2005.24supplement .5
- DeFond, M. L., & Lennox, C. S. (2011). The effect of SOX on small auditor exits and audit quality. *Journal of Accounting & Economics*, 52(1), 21–40. https://doi.org/10.1016/j.jacceco.2011.03.002
- DeFond, M. L., Francis, J. R., & Wong, T. J. (2000). Auditor industry specialization and market segmentation: Evidence from Hong Kong. *Auditing: A Journal of Practice & Theory*, *19*(1), 49-66. https://doi.org/10.2308/aud.2000.19.1.49
- Dettmer, J. (2011). Enron Collapse: The scandal that shook wall street. Retrieved from: https://www.thisismoney.co.uk/money/markets/article-2071852/Enron-collapse-The-scandal-shook-Wall-Street.html
- Dex, S., & Bond, S. (2005). Measuring work-life balance and its covariates. *Work, Employment and Society*, 19(3), 627-637. https://doi.org/10.1177/0950017005055676
- Dhas, B. (2015). A report on the importance of work-life balance. *International Journal of Applied Engineering Research*, 10(9), 21659-21665.

- Dodgson, M. K., Agoglia, C. P., Bennett, G. B., & Cohen, J. R. (2020). Managing the auditor-client relationship through partner rotations: The experiences of audit firm partners. *The Accounting Review*, 95(2), 89-111. https://doi.org/10.2308/accr-52556
- Dowling, C., Knechel, W. R., & Moroney, R. (2018). Public oversight of audit firms: The slippery slope of enforcing regulation. *Abacus*, *54*(3), 353-380. https://doi.org/10.1111/abac.12130
- Ellemers, N., Spears, R., & Doosje, B. (2002). Self and social identity. *Annual Review of Psychology*, 53(1), 161-186. https://doi.org/10.1146//annurev.psych.53.100901.135228
- Evans, P. A. (1986). The strategic outcomes of human resource management. *Human*\*Resource Management, 25(1), 149-167. https://doi.org/10.1002/hrm.3930250111
- External Reporting Board. (2011). International standards on auditing (New Zealand) 220.

  Retrieved from: https://www.xrb.govt.nz/dmsdocument/1540
- External Reporting Board. (2018). Professional and ethical standard 1. Retrieved from: https://www.xrb.govt.nz/dmsdocument/4890
- Ferguson, A., Francis, J. R., & Stokes, D. J. (2003). The effects of firm-wide and office-level industry expertise on audit pricing. *The Accounting Review*, 78(2), 429-448. https://doi.org/10.2308/accr.2003.78.2.429
- Financial Markets Authority. (2012). Financial Markets Authority Liechtenstein. Retrieved from: https://www.fma-li.li/files/fma/fma-mm-ifiar-e.pdf
- Financial Markets Authority. (2016). Regulatory Response Guidelines. Retrieved from: https://www.fma.govt.nz/assets/Policies/160824-Regulatory-response-guidelines-policy.pdf

- Financial Markets Authority. (2020). The auditor regulation act (prescribed minimum standards and conditions for licensed auditors and registered audit firms). fma.govt: https://www.fma.govt.nz/assets/Notices/Auditor-Regulation-Act-Notice-2020.pdf
- Financial Markets Authority. (2021). Audit quality monitoring report. Retrieved from: https://www.fma.govt.nz/assets/Reports/Audit-Quality-Monitoring-Report-2021.pdf
- Financial Markets Authority. (2021). Temporary extension for certain reporting deadlines.

  Retrieved from: https://www.fma.govt.nz/news/all-releases/covid-19/temp-extension-reporting/
- Financial Reporting Council. (2021). Key facts and trends in the accountancy profession.

  Retrieved from: https://www.frc.org.uk/getattachment/e976ff38-3597-4779-b192-1be7da79d175/FRC-Key-Facts-Trends-2021.pdf
- Fitzgerald, B. C., Omer, T. C., & Thompson, A. M. (2018). Audit partner tenure and internal control reporting quality: US evidence from the not-for-profit sector. *Contemporary Accounting Research*, 35(1), 334-364. https://doi.org/10.1111/1911–3846.12348
- Foley, S. (2022). Accountants work to shed 'boring' tag amid hiring crisis. Retrieved from: https://www.ft.com/content/4d014b21-c4f6-4562-ab33-48a51d0312b0
- Ford, J., & Marriage, M. (2018). The big flaw: auditing in crisis. Retrieved from: https://www.ft.com/content/29ccd60a-85c8-11e8-a29d-73e3d454535d
- Francis, J. R. (2004). What do we know about audit quality?. *The British Accounting Review*, *36*(4), 345-368. https://doi.org/10.1016/j.bar.2004.09.003
- Frankenfield, J., & Kindness, D. (2020) Waterfall payment: definition, benefits, how it works and examples: Retrieved from:

  https://www.investopedia.com/terms/w/waterfallpayment.asp

- Frick, W. (2018. The conundrum of corporate power: Retrieved from: https://hbr.org/2018/05/the-conundrum-of-corporate-power
- Galbraith, J. R. (1971). Matrix organization designs How to combine functional and project forms. *Business Horizons*, *14*(1), 29-40. https://doi.org/10.1016/0007–6813(71)90037–1
- Gao, L., Janssen, O., & Shi, K. (2011). Leader trust and employee voice: The moderating role of empowering leader behaviors. *The Leadership Quarterly*, 22(4), 787-798. https://doi.org/10.1016/j.leaqua.2011.05.015
- Gasson, S. (2004). Rigor in grounded theory research: An interpretive perspective on generating theory from qualitative field studies. In *The Handbook of Information Systems Research* (pp. 79-102). IGI Global.
- Ghouri, F. (2021). Audit reforms: IoD call for caution in plans to break Big Four dominance:

  Retrieved from: https://www.cityam.com/audit-reforms-iod-call-for-caution-in-plans-to-break-big-four-dominance/
- Goelzer, D. (2019). The impact of disclosing engagement partner identity: no clear answer.

  Retrieved from: https://medium.com/the-audit-blog/the-impact-of-disclosing-engagement-partner-identity-no-clear-answer-290ee5f58ac0
- Graneheim, U. H., & Lundman, B. (2004). Qualitative content analysis in nursing research: concepts, procedures and measures to achieve trustworthiness. *Nurse Education Today*, 24(2), 105-112. https://doi.org/10.1016/j.nedt.2003.10.001
- Gray, T. (2023). Avoid employee poaching by competitors. Retrieved from: https://recruit-nz.co.nz/avoid-employee-poaching-by-competitors/

- Guénin-Paracini, H., Malsch, B., & Tremblay, M. S. (2015). On the operational reality of auditors' independence: Lessons from the field. *Auditing: A Journal of Practice & Theory*, *34*(2), 201-236. https://doi.org/10.2308/ajpt-50905
- Gul, F. A., & Tsui, J. S. L. (1997). A test of the free cash flow and debt monitoring hypotheses: Evidence from audit pricing. *Journal of Accounting and Economics*, 24(2), 219-237. https://doi.org/10.1016/S0165-4101(98)00006-8
- Gunny, Katherine A., and Tracey Chunqi Zhang. "PCAOB inspection reports and audit quality." *Journal of Accounting and Public Policy* 32, no. 2 (2013): 136-160. https://doi.org/10.1016/j.jaccpubpol.2012.11.002
- Hammer, M. (2001). *Harvard business review*, 79(8), 82-93.
- Hammersley, J. S. (2006). Pattern identification and industry-specialist auditors. *The Accounting Review*, 81(2), 309-336. https://doi.org/10.2308/accr.2006.81.2.309
- Harber, M., Verhoef, G., & de Villiers, C. (2022). Disputed interpretations and active strategies of resistance during an audit regulatory debate. *Accounting, Auditing & Accountability Journal*.
- He, X., Pittman, J., & Rui, O. (2016). Reputational implications for partners after a major audit failure: Evidence from China. *Journal of Business Ethics*, *138*(4), 703-722. https://doi.org/10.1007/s10551-015-2770-6
- Hellman, N. (2011). Chief financial officer influence on audit planning. *International Journal of Auditing*, 15(3), 247-274. https://doi.org/10.1111/j.1099–1123.2011.00433.x
- Hendry–Tennent. I. (2022). NZ has near–record low unemployment, severe staffing shortages so why are some Kiwis stil struggling to find work? Retrieved from:

  https://www.newshub.co.nz/home/new-zealand/2022/11/nz-has-near-record-low-

- unemployment-severe-staffing-shortages-so-why-are-some-kiwis-still-struggling-to-find-work.html
- Herrick, T., Barrionuevo, A. (2002). Were Enron, Anderson too close to allow auditors to do its job. Retrieved from: https://www.wsj.com/articles/SB1011565452932132000
- Hogg, M. A. (2000). Subjective uncertainty reduction through self-categorization: A motivational theory of social identity processes. *European Review of Social Psychology*, 11(1), 223-255. https://doi.org/10.5465/AMR.2000.2791606
- Holloway, I., & Todres, L. (2003). The status of method: flexibility, consistency and coherence. *Qualitative research*, *3*(3), 345-357. https://doi.org/10.1177/1468794103033004
- Houghton, K. A., Kend, M., & Jubb, C. (2013). The CLERP 9 audit reforms: Benefits and costs through the eyes of regulators, standard setters and audit service suppliers. *Abacus*, 49(2), 139-160. https://doi.org/10.1111/abac.12002
- Huang, T. C., Chang, H., & Chiou, J. R. (2016). Audit market concentration, audit fees, and audit quality: Evidence from China. *Auditing: A Journal of Practice & Theory*, *35*(2), 121-145. https://doi.org/10.2308/ajpt-51299
- Humphrey, C., Kausar, A., Loft, A., & Woods, M. (2011). Regulating audit beyond the crisis:

  A critical discussion of the EU Green Paper. *European Accounting Review*, 20(3),
  431-457. https://doi.org/10.1080/09638180.2011.597201
- Huselid, M. A., Jackson, S. E., & Schuler, R. S. (1997). Technical and strategic human resources management effectiveness as determinants of firm performance. *Academy of Management journal*, 40(1), 171-188. https://doi.org/10.2307/257025
- International Auditing and Assurance Standards Board. (2019). Conforming and consequential amendments to other international standards arising from ISA (Revised

- 2019). Retrieved from: https://www.ifac.org/system/files/publications/files/ISA-315-Full-Standard-and-Conforming-Amendments-2019-.pdf
- International Ethics Standards Board for Accountants. (2007). Key audit partner definition detailed comments. Retrieved form:

  https://www.ifac.org/system/files/meetings/files/3153.pdf
- International Federation of Accountants (IFAC). (2003). Coe of Ethics. New York, NY: International Federation of Accountants.
- Jackson, A. B., Moldrich, M., & Roebuck, P. (2008). Mandatory audit firm rotation and audit quality. *Managerial Auditing Journal*, 23(5), 420-437. https://doi.org/10.1108/02686900810875271
- Jenkins, D. S., & Velury, U. (2008). Does auditor tenure influence the reporting of conservative earnings?. *Journal of Accounting and Public Policy*, 27(2), 115-132. https://doi.org/10.1016/j.jaccpubpol.2008.01.005
- Jones, C. (2022). The NZ auditor shortage. Retrieved from: https://tylerwren.co.nz/the-nz-audit.
- Kallapur, S., Sankaraguruswamy, S., & Zang, Y. (2010). Audit market concentration and audit quality. *Available at SSRN 1546356*.
- Kaplan, R. S., & Norton, D. P. (2005). The office of strategy management. *Harvard business* review, 83(10), 72.
- Khan, S. N. (2014). Qualitative research method: Grounded theory. *International Journal of Business and Management*, *9*(11), 224-233.

- Khurana, I. K., & Raman, K. K. (2004). Litigation risk and the financial reporting credibility of Big 4 versus non-Big 4 audits: Evidence from Anglo-American countries. *The Accounting Review*, 79(2), 473-495. https://doi.org/10.2308/accr.2004.79.2.473
- Kim, J. B., Song, B. Y., & Tsui, J. S. (2013). Auditor size, tenure, and bank loan pricing. *Review of Quantitative Finance and Accounting*, 40, 75-99. https://doi.org/10.1007/s11156-011-0270-z
- Knechel, R., Ghandar, A. (2021). Do audit inspections improve audit quality. Retrieved from: https://www.ifac.org/knowledge-gateway/supporting-international-standards/discussion/do-audit-inspections-improve-audit-quality
- Knechel, W. R., & Vanstraelen, A. (2007). The relationship between auditor tenure and audit quality implied by going concern opinions. *Auditing: A Journal of Practice & Theory*, 26(1), 113-131.
- Knechel, W. R., Krishnan, G. V., Pevzner, M., Shefchik, L. B., & Velury, U. K. (2013).
  Audit quality: Insights from the academic literature. *Auditing: A Journal of Practice*, 32(Supplement 1), 385-421. https://doi.org/10.2308/ajpt-50350
- Kramer, RJ (1994). "Organizing for Global Competitiveness: The Matrix Design," Conference Board report no. 1088-94-RR (New York: Conference Board).
- Lagace, M. (2013). Are the Big Four audit firms to big to fail? Retrieved from: https://hbswk.hbs.edu/item/are-the-big-four-audit-firms-too-big-to-fail
- Lammintakanen, J., Saranto, K., Kivinen, T., & Kinnunen, J. (2002). The digital portfolio: a tool for human resource management in health care?. *Journal of Nursing Management*, 10(6), 321-328. https://doi.org/10.1046/j.1365–2834.2002.00346.x

- Lawrence, A., Minutti-Meza, M., & Zhang, P. (2011). Can Big 4 versus non-Big 4 differences in audit-quality proxies be attributed to client characteristics?. *The Accounting Review*, 86(1), 259-286. https://doi.org/10.2308/accr.00000009
- Leeds. J. (2002). Anderson auditor details shredding of shredding of Enron papers. Retrieved from: https://www.latimes.com/archives/la-xpm-2002-may-14-fi-andersen14-story.html
- Lennox, C. (2005). Management ownership and audit firm size. *Contemporary Accounting*\*Research, 22(1), 205-227. https://doi.org/10.1506/K2CG-U6VO-NPTC-EQBK
- Lennox, C. S., Wu, X., & Zhang, T. (2014). Does mandatory rotation of audit partners improve audit quality?. *The Accounting Review*, 89(5), 1775-1803. https://doi.org/10.2308/accr-50800
- Lennox, C., & Pittman, J. (2010). Auditing the auditors: Evidence on the recent reforms to the external monitoring of audit firms. *Journal of Accounting and Economics*, 49(1-2), 84-103. https://doi.org/10.1016/j.jacceco.2009.04.002
- Lennox, C., Wang, C., & Wu, X. (2023). Delegated leadership at public accounting firms. *Journal of Accounting and Economics*, 76(1), 101572. https://doi.org/10.1016/j.jacceco.2022.101572
- Lewicki, R. J., & Bunker, B. B. (1996). Developing and maintaining trust in work relationships. *Trust in organizations: Frontiers of Theory and Research*, 114, 139.
- Lincoln, Y. S., & Guba, E. G. (1985). Naturalistic inquiry. sage.
- Litt, B., Sharma, D. S., Simpson, T., & Tanyi, P. N. (2014). Audit partner rotation and financial reporting quality. *Auditing: A Journal of Practice and Theory*, 33(3), 59–86. https://doi.org/10.2308/ajpt-50753

- Lockwood, N. R. (2003). Work/life balance. Challenges and Solutions. *Benefits Quarterly*, 19(4), 94.
- Macknight, J. (2011). Aligning accounting standards: The grail of one global GAAP: retrieved from: https://www.theglobaltreasurer.com/2011/03/11/aligning-accounting-standards-the-grail-of-one-global-gaap/
- Marriage, M. (2018). The big flaw: auditing in crisis. Retrieved from: https://www.ft.com/content/29ccd60a-85c8-11e8-a29d-73e3d454535d
- Martic. K. (2023). Millennials in the workplace: 11 ways to attract and keep them: Retrieved from: https://haiilo.com/blog/millennials-in-the-workplace-11-ways-to-attract-and-keep-them/
- Martin. M. (2022). Exodus worsens labour crunch as NZ reopens borders. Retrieved from: https://www.mpamag.com/nz/news/general/exodus-worsens-labour-crunch-as-nz-reopens-its-borders/412871
- Masters. B. (2011). Enron's fall raised the bar in regulation. Retrieved from: https://www.ft.com/content/9790ea78-1aa9-11e1-ae14-00144feabdc0
- Matheson, J. (2020). Audit reviews: the process, pitfalls and published themes. Retrieved from: https://www.kingsleynapley.co.uk/insights/blogs/regulatory-blog/audit-reviews-the-process-pitfalls-and-published-themes
- Mattocks, R., Huang, T. C., Moroney, R. A., & Prasad, A. (2022). Does the length of the cooling-off period affect audit quality? *A Journal of Practice & Theory*, 41(3), 149-174. https://doi.org/10.2308/AJPT-19-091
- Maurer, M. (2002). Arthur Anderson legacy, 20 years after its demise, its complicated.

  Retrieved from: https://www.wsj.com/articles/arthur-andersens-legacy-20-years-after-its-demise-is-complicated-11661938200

- Mautz, R. K., & Sharaf, H. A. (1961). The philosophy of auditing. Sarasota, FL: *American Accounting Association*.
- Messier, W. F., Owhoso, V., & Rakovski, C. (2008). Can audit partners predict subordinates' ability to detect errors? *Journal of Accounting Research*, 46(5), 1241–1264. https://doi.org/10.1111/j.1475–679X.2008.00307.x
- Michie, S. (2002). Causes and management of stress at work. *Occupational and environmental medicine*, 59(1), 67-72. https://doi.org/10.1136/oem.59.1.67
- Mills, D., & Timmins, J. (2004). Firm dynamics in New Zealand: A comparative analysis with OECD countries (No. 04/11). New Zealand Treasury Working Paper.
- Moison, J. (2021). Reporting a deficit how to address the auditor shortage in New Zealand.

  Retrieved from: https://www.fma.govt.nz/library/opinion/auditor-shortage/
- Morgan, G., & Quack, S. (2005). internationalization of UK and German law firms. *Organization Studies*, 26(12), 1765-1785. https://doi.org/10.1177/0170840605059156
- Moshirian, F. (2011). The global financial crisis and the evolution of markets, institutions and regulation. *Journal of Banking & Finance*, *35*(3), 502-511. https://doi.org/10.1016/j.jbankfin.2010.08.010
- Myers, J. N., Myers, L. A., & Omer, T. C. (2003). Exploring the term of the auditor-client relationship and the quality of earnings: A case for mandatory auditor rotation? *The Accounting Review*, 78(3), 779–799. https://doi.org/10.2308/accr.2003.78.3779
- Myers, J. N., Myers, L. A., Palmrose, Z. V., & Scholz, S. (2005). The length of auditor-client relationships and financial statement restatements. *online*] (cited March 2010).

- Myers, M. D., & Newman, M. (2007). The qualitative interview in IS research: Examining the craft. *Information and Organization*, *17*(1), 2-26. https://doi.org/10.1016/j.infoandorg.2006.11.001
- New Zealand Companies Office. Financial reporting for FMC reporting entities. Retrieved from: https://companies-register.companiesoffice.govt.nz/help-centre/financial-reporting/financial-reporting-for-fmc-reporting-entities/
- O'Dwyer, M., & Pickford, J. (2022). Cost of UK plan to break big four stranglehold rises to £1bn. Retrieved from: https://www.ft.com/content/347e5dbb-c8e1-4cd0-8989-93fba1f935d4
- O'Neil, P. (2015). Regulatory prosecutions the experience so far. Retrieved from: https://www.fma.govt.nz/library/speeches-and-presentations/presentation-by-paul-oneil-to-the-competition-and-law-policy-institute-of-new-zealand/
- Office of the Auditor General. (2004). Annual report 2004–05. Retrieved form: https://oag.parliament.nz/2005/ar-2004-05/about.htm
- Ogharanduk, B., Jubb, D., Lochrie, S., Curran, R., & D O'Gorman, K. (2016). Methods and techniques for qualitative data gathering.
- Osanloo, A., & Grant, C. (2016). Understanding, selecting, and integrating a theoretical framework in dissertation research: Creating the blueprint for your "house". *Administrative Issues Journal: Connecting Education, Practice, and Research*, 4(2). https://doi.org/10.5929/2014.4.2.9
- Palmrose, Z. V. (2006). Maintaining the value and viability of independent auditors as gatekeepers under SOX: An auditing master proposal. *Financial gatekeepers: Can they protect investors*, 103-135. https://doi.org/10.7864/j.ctt6wpg1w.7

- Parker, R. H. (2005). Naming and branding: accountants and accountancy bodies in the British Empire and Commonwealth, 1853-2003. *Accounting History*, *10*(1), 7-46. https://doi.org/10.1177/103237320501000102
- Pells. S. (2022). New Zealand's areas of (economic) strength. Retrieved from:

  https://www.mbie.govt.nz/dmsdocument/18921-new-zealands-areas-of-economicstrength-a-literature-review
- Pentland, B. T. (1993). Getting comfortable with the numbers: Auditing and the micro-production of macro-order. *Accounting, Organizations and Society*, *18*(7-8), 605-620. https://doi.org/10.1016/0361-3682(93)90045-8
- Pickens, J. (2005). Attitudes and perceptions. *Organizational Behavior in Health Care*, 4(7), 43-76.
- Pittman, J., Wang, L., & Wu, D. (2022). Network analysis of audit partner rotation.

  \*Contemporary Accounting Research, 39(2), 1085-1119. https://doi.org/10.1111/1911-3846.12743
- Pope, C., Ziebland, S., & Mays, N. (2000). Qualitative research in health care: Analysing qualitative data. *BMJ: British Medical Journal*, *320*(7227), 114. https://doi.org/10.1136/bmj.320.7227.114
- Porter, M. E. (1980). Industry structure and competitive strategy: Keys to profitability. *Financial Analysts Journal*, *36*(4), 30-41. https://doi.org/10.2469/faj.v36.n4.30
- Power, M. K. (2003). Auditing and the production of legitimacy. *Accounting, Organizations and Society*, 28(4), 379-394. https://doi.org/10.1016/S0361-3682(01)00047-2
- Quinn, J. B. (1992). *Intelligent Enterprise: A knowledge and service based paradigm for Industry*. Simon and Schuster.

- Qureshi, M. I., Iftikhar, M., Abbas, S. G., Hassan, U., Khan, K., & Zaman, K. (2013).

  Relationship between job stress, workload, environment and employees turnover intentions: What we know, what should we know. *World Applied Sciences Journal*, 23(6), 764-770. https://doi.org/10.5829/idosi.wasj.2013.23.06.313
- Rabah, M. (2020). Agile internal auditing vs. traditional waterfall auditing. Retrieved from: https://www.xpertsleague.com/agile-internal-auditing-vs-traditional-waterfall-auditing/
- Rainsbury, E. A. (2019). Audit quality in New Zealand early evidence from the regulator.

  Australia Accounting Review, 29(3), 455–467. https://doi.org/10.1111/auar.12276
- Ramlawati, R., Trisnawati, E., Yasin, N., & Kurniawaty, K. (2021). External alternatives, job stress on job satisfaction and employee turnover intention. *Management Science Letters*, 11(2), 511-518. https://doi.org/10.5267/j.msl.2020.9.016
- Richard, C. (2006). Why an auditor can't be competent and independent: A French case study. *European accounting review*, *15*(2), 153-179. https://doi.org/10.1080/09638180500104832
- Riketta, M. (2005). Organizational identification: A meta-analysis. *Journal of vocational behavior*, 66(2), 358-384. https://doi.org/10.1016/j.jvb.2004.05.005
- Robert, M. B. L. (1999). A unified account of the effects of caricaturing faces. *Visual Cognition*, 6(1), 1-42. https://doi.org/10.1080/713756800
- Rocco, T. S., & Plakhotnik, M. S. (2009). Literature reviews, conceptual frameworks, and theoretical frameworks: Terms, functions, and distinctions. *Human Resource Development Review*, 8(1), 120-130. https://doi.org/10.1177/1534484309332617
- Rogstadius, J., Kostakos, V., Kittur, A., Smus, B., Laredo, J., & Vukovic, M. (2011). An assessment of intrinsic and extrinsic motivation on task performance in

- crowdsourcing markets. In *Proceedings of the international AAAI conference on web and social media* (Vol. 5, No. 1, pp. 321-328).
- Romano, R. (2004). The Sarbanes-Oxley Act and the making of quack corporate governance. *Yale LJ*, *114*, 1521. https://doi.org/10.2139/ssrn.596101
- Schuler, R. S., & Jackson, S. E. (1987). Linking competitive strategies with human resource management practices. *Academy of Management Perspectives*, *1*(3), 207-219. https://doi.org/10.5465/ame/.1987.4275740
- Scott, W. R. (2005). Institutional theory: Contributing to a theoretical research program. *Great minds in management: The Process of Theory Development*, *37*(2), 460-484.
- Securities and Exchange Commission. (2003). Strengthening the Commission's Requirements Regarding Auditor Independence. Retrieved from: https://www.sec.gov/rules/final/33-8183.htm
- Seetharaman, A., Gul, F. A., & Lynn, S. G. (2002). Litigation risk and audit fees: Evidence from UK firms cross-listed on US markets. *Journal of Accounting and Economics*, 33(1), 91-115. https://doi.org/10.1016/S0165-4101(01)00046-5
- Segal, T. (2021). Enron Scandal: The fall of a wall street darling. Retrieved from: https://www.investopedia.com/updates/enron-scandal-summary/
- Sharma, D. S., Tanyi, P. N., & Litt, B. A. (2017). Costs of mandatory periodic audit partner rotation: Evidence from audit fees and audit timeliness. *Auditing: A Journal of Practice & Theory*, *36*(1), 129-149. https://doi.org/10.2308/ajpt-51515
- Sikka, P. (2009). Financial crisis and the silence of the auditors. *Accounting, organizations* and society, 34(6-7), 868-873. https://doi.org/10.1016/j.aos.2009.01.004

- Simpson, J. A. (2007). Foundations of interpersonal trust. *Social psychology: Handbook of Basic Principles*, 2, 587-607.
- Singh, A., Singh, H., Sultana, N., & Evans, J. (2019). Independent and joint effects of audit partner tenure and non-audit fees on audit quality. *Journal of Contemporary*Accounting & Economics, 15, 186-205. https://doi.org/10.1016/j.jcae.2019.04.005
- Sivignon. C. (2022). Numbers don't add up: demand on auditors rise amid global shortage.

  Retrieved from: https://www.stuff.co.nz/national/130756716/numbers-dont-add-up-demands-on-auditors-rise-amid-global-shortage
- Skilling, D. (2020). Frontier firms: An international small advanced economy perspective. *Landfall Strategy Group*.
- Smith, A. (1779). Wealth of nations. Strahan and Cadell, London.
- Smith, C. (2023). Pay rise? Salary increases amid soaring inflation, tight labour market, unlikely. Retrieved from: https://www.nzherald.co.nz/business/cloudy-outlook-for-salary-increases-amid-soaring-inflation-tight-labour-market/QL4T5DZT6JAQZJIPCEQLJNOX34/
- Sokro, E. (2012). Impact of employer branding on employee attraction and retention. *European Journal of Business and Management*, 4(18), 164-173.
- Soliman, F., & Spooner, K. (2000). Strategies for implementing knowledge management: role of human resources management. *Journal of Knowledge Management*, *4*(4), 337–345. https://doi.org/10.1108/13673270010379894
- Sorkin, A., Giang, V., Gandel, S., Hirsh, L., Livni, E., & Gross, J. (2022). The Big 4's tax problem. Retrieved from:

  https://www.nytimes.com/2022/07/07/business/dealbook/big-4-firms.html

- Sturt, D. (2015). Know the 5-year job itch? You can fight it. Retrieved from: https://www.success.com/know-the-5-year-job-itch-you-can-fight-it/
- Sufi, A. (2007). Information asymmetry and financing arrangements: Evidence from syndicated loans. *The Journal of Finance*, *62*(2), 629-668. https://doi.org/10.1111/j.1540–6261.2007.01219.x
- Sy, T., Beach, L., & D'Annunzio, L. S. (2005). Challenges and strategies of matrix organizations. *Human Resource Planning*, 28(1), 39-48.
- The Association of International Certified Professional Accountants. (2019). 2019 trends in the supply of accounting graduates and the demand for public accounting recruits.

  Retrieved from:

  https://us.aicpa.org/content/dam/aicpa/interestareas/accountingeducation/newsandpublications/downloadabledocuments/2019-trends-report.pdf
- Thompson, S. C. (2016). Accounting for a Developing World: A look at International Standards on Developing Countries. University of Tennessee, Knoxville.
- Titman, S., & Trueman, B. (1986). Information quality and the valuation of new issues. *Journal of Accounting and Economics*, 8(2), 159-172. https://doi.org/10.1016/0165-4101(86)90016-9
- Toe, S. (2023). Heavy work load and what to do about it. Retrieved from: https://www.runn.io/blog/how-to-manage-a-heavy-workload-effectively
- Torraco, R. J., & Swanson, R. A. (1995). The strategic roles of human resource development. *Human Resource Planning*, *18*, 10-21.
- Usmani. H. (2022). The future of accounting: demand and evolving technology 2022.

  Retrieved from: https://expertiseaccelerated.com/the-future-of-accounting/

- Vanclay, F. (1999). Social impact assessment. *Handbook of environmental impact assessment*, 1(1999), 301-326.
- Vemparala, T. (2023). Solving the mystery of Millennial and Gen Z job hoppers. Retrieved from: https://www.businessnewsdaily.com/7012-millennial-job-hopping.html
- Vien, C. (2022). Succession issues surge at accounting firms. *Journal of Accountancy*.

  Retrieved from: https://www.journalofaccountancy.com/news/2020/dec/succession-issues-surge-at-accounting-firms.html
- Viscusi, W. K. (Ed.). (2004). Regulation through litigation. Brookings Institution Press.
- Wallison, P. J. (2005). Rein in the Public Company Accounting Oversight Board.
- Werner, M., & Gehrke, N. (2015). Multilevel process mining for financial audits. *IEEE Transactions on Services Computing*, 8(6), 820-832.
- Williams-Alvarez, J. (2022). U.K. Government looks to limit dominance of Big Four audit firms. Retrieved from: https://www.wsj.com/articles/u-k-government-looks-to-limit-dominance-of-big-four-audit-firms-11654029340
- Wilson, A. B., McNellis, C., & Latham, C. K. (2018). Audit firm tenure, auditor familiarity, and trust: Effect on auditee whistleblowing reporting intentions. *International Journal of Auditing*, 22(2), 113-130. https://doi.org/10.1111/ijau.12108
- Winn, A. M. (2021). The joint effects of partner rotation and PCAOB inspections on audit effort. *Auditing: A Journal of Practice & Theory*, 40(3), 155-174. https://doi.org/10.2308/AJPT-18-121
- Wooldridge. A. (2022). The accountant shortage threatens capitalism's future. Retrieved from: https://www.washingtonpost.com/business/the-accountant-shortage-threatens-capitalisms-future/2022/11/01/3e760160-59ab-11ed-bc40-b5a130f95ee7\_story.html

- Wooll, M. (2022). Workplace stress: think it isn't getting to you? Think again: Retrieved from: https://www.betterup.com/blog/effects-of-workplace-stress
- Work, psychological well-being and performance. Occupational Medicine, 50(5), 304-309.
- Yahanpath, N., & Cavanagh, J. (2011). Causes of New Zealand finance company collapses:

  A brief review. *Available at SSRN 1906923*.
- Yerman, E. (2013. How to scratch the 'five year itch' at work. Retrieved from: https://www.bizjournals.com/bizjournals/how-to/human-resources/2013/07/how-to-scratch-the-five-year-itch-at.html
- Young, K., Banerjee, T., & Schwartz, M. (2020). The secret to corporate power. Retrieved from: https://www.yesmagazine.org/democracy/2020/07/13/corporate-power-politics
- Zelizer, E. G. (2002). The Sarbanes-Oxley Act: Accounting for Corporate Corruption. *Loy. Consumer L. Rev.*, 15, 27.
- Zerni, M. (2012). Audit partner specialization and audit fees: Some evidence from Sweden. *Contemporary Accounting Research*, 29(1), 312-340. https://doi.org/10.1111/j.1911-3846.2011.01098.x

## Appendices

## **Appendix 1: Initial contact email to participants – email template**

Subject: Invitation to Participate in Audit Partner Rotation Research

Dear [Recipient],

My name is Adam Fairburn, I am currently completing a Master's thesis at the University of Canterbury in Auditing.

This email is an invitation for you to participate in my research, which is focused on the imposition of 'mandatory audit partner rotation regulation' for the audits of certain entities in New Zealand. The purpose of this research is to give audit partners an opportunity to share their experiences and perspectives on this topic. I am interested to discover how audit firms have responded to the rotation requirements and determine the nature of the regulation's impacts.

As you will be aware, mandatory audit partner rotation regulations have been introduced in many countries, and several studies have identified that their introduction has had significant consequences for audit partners professionally, and in their personal lives. The impacts of these policies are of considerable interest to a range of stakeholders, including regulators, policy makers, shareholders, and others. Ultimately, this research seeks to communicate the opinions of New Zealand audit partners to regulators and other interested parties by way of an academic journal article and to contribute to the international debate surrounding this topic.

Participation in this project will involve an interview discussing your experiences about the aforementioned regulation. The interview will take approximately one hour and can be conducted either face-to-face or via Zoom, at a time and place that is convenient to you. All information provided will be considered completely confidential and your anonymity will be

assured. For further information about this research and procedures for ensuring confidentiality the following documents are attached:

- Consent form
- Information sheet for participants
- Interview protocol

If you agree to participate in this research project, I would be grateful if you could fill in the attached consent form and return a scanned copy to Adam Fairburn at adam.fairburn@pg.canterbury.ac.nz before the [two weeks from the date the email was sent]. I will contact you shortly thereafter to arrange an interview.

Alternatively, if you do not wish to participate, please respond "No" to this email and you will be removed from the mailing list.

Should you have further questions, please feel free to contact me.

Thank you for your time and I look forward to hearing your reply.

Kind Regards

Adam Fairburn

## **Appendix 2: Interview Protocols**

Interview protocols

What I am looking to hear from you in this interview, is your lived experiences and what you observed happening to colleagues and other audit partners throughout this process of change.

- 1. Please describe your experiences of audit partner rotation(s)? What had happened? Please describe this process.
- 2. How did the above rotation(s) affect your relationships with clients?

- 3. From your audit firm perspective what changes have you observed taking place following the introduction of audit partner rotation? What steps have management taken to ensure that audit quality and good client relationships have been maintained?
- 4. From an industry perspective what changes have you observed taking place following the introduction of audit partner rotation? What steps have management taken to ensure that audit quality and good client relationships have been maintained?
- 5. How do you evaluate whether audit partner rotation policies have achieved their objective(s)?
- 6. Are there any implications of mandatory audit partner rotation that you had not anticipated until you have experienced the rotation yourself? Please describe
- 7. The introduction of the FMA in New Zealand can be said to have reduced the number of domestically registered audit companies. How would you evaluate this statement? What do you think might be the reason?
- 8. How do you relate audit partner rotation policies to audit quality you have described?
- 9. Do you feel like audit partner rotation policies are suitable in their current form? What recommendations would you make to regulatory bodies regarding there suitability and the appropriateness of timeframes?
- 10. Before we conclude this interview, is there anything else you would like to add